

Title	Studies on Formation of Carbon-Carbon or Carbon- Oxygen Bonds by Activation of Alkenes and Alkynes Using Heavier Group 13 Metal Salts
Author(s)	喜多, 悠二
Citation	大阪大学, 2019, 博士論文
Version Type	VoR
URL	https://doi.org/10.18910/72357
rights	
Note	

## Osaka University Knowledge Archive : OUKA

https://ir.library.osaka-u.ac.jp/

Osaka University

# **Doctoral Dissertation**

Studies on Formation of Carbon-Carbon or Carbon-Oxygen Bonds by Activation of Alkenes and Alkynes

## **Using Heavier Group 13 Metal Salts**

Yuji Kita

January 2019

**Department of Applied Chemistry** 

**Graduate School of Engineering** 

Osaka University

#### Preface and Acknowledgements

The work of this thesis has been performed from 2013 to 2019 under the guidance of Prof. Dr. Makoto Yasuda at Department of Applied Chemistry, Graduate School of Engineering, Osaka University. The thesis describes transformation of alkenes or alkynes using heavier group 13 metal salts for formation of carbon-carbon or carbon-oxygen bonds.

First and foremost, I would like to express my sincerest gratitude to Prof. Dr. Makoto Yasuda for their precise guidances, helpful suggestions, and hearty encouragements throughout this work.

I would also like to thank Professors Dr. Ikuya Shibata and Dr. Sensuke Ogoshi for their helpful advice and kind assistance.

I really wish to make a grateful acknowledgement to Assistant Prof. Dr. Yoshihiro Nishimoto for his intimate guidance, continuous advice, and kind encouragement.

I gratefully express acknowledgement to Assistant Prof. Dr. Akihito Konishi for his invaluable assistance, helpful suggestion, and stimulating discussion.

I would sincerely like to thank Prof. Dr. Martin Oestreich for giving me an opportunity of my overseas education in Technische Universität Berlin.

I would like to express my heartfelt thanks to Mr. Hiroshi Moriguchi, Dr. Nobuko Kanehisa, Dr. Kyoko Inoue, Dr. Hiroaki Tanaka, Mr. Kunihiko Kamon and Mr. Hiromi Ohi for giving me analytic assistance at the analytical instrumentation facility.

I am deeply thankful to Ms. Yoshimi Shinomiya for giving me her grateful support and heartwarming kindness.

Furthermore, I gratefully wish to thank Mr. Tetsuji Yata for his valuable discussion and active working. I also wish to acknowledge to all the members of Yasuda group for their hearty encouragement, constant support and assistance.

Finally, I would like to express my thanks to respectable parents, Eiji Kita and Sadako Kita for their understanding to my work, constant assistance, and financial support.

Yuji Kita

Department of Applied Chemistry Graduate School of Engineering Osaka University 2-1 Yamadaoka, Suita, Osaka 565-0871, JAPAN January, 2019

#### **List of Publications**

- Coupling Reaction of Enol Derivatives with Silyl Ketene Acetals Catalyzed by Gallium Trihalides Y. Nishimoto, <u>Y. Kita</u>, H. Ueda, H. Imaoka, K. Chiba, M. Yasuda, A. Baba *Chem. Eur. J.* **2016**, *22*, 11837-11845.
- 2) Indium Catalyzed Hydrofunctionalization of Styrene Derivatives Bearing a Hydroxy Group with Organosilicon Nucleophiles
   <u>Y. Kita</u>, T. Yata, Y. Nishimoto, M. Yasuda
   J. Org. Chem. 2018, 83, 740-753.
- Selective Oxymetalation of Terminal Alkynes via 6-Endo Cyclization: Mechanistic Investigation and Application to the Efficient Synthesis of 4-Substituted Isocoumarins <u>Y. Kita</u>, T. Yata, Y. Nishimoto, K. Chiba, M. Yasuda Chem. Sci. 2018, 9, 6041-6052.
- 4) Synthesis of 2,4,6-Triarylpyryliums Bearing a Carbon-Metal Bond and their Application to Bifunctional Photocatalyst
   <u>Y. Kita</u>, Y. Nishimoto, M. Yasuda
   *Manuscript under preparation*

#### **Supplementary List of Publications**

 Kinetic Resolution of α-Hydroxy-Substituted Oxime Ethers by Enantioselective Cu-H-Catalyzed Si-O Coupling
 X. Dong, <u>Y. Kita</u>, M. Oestreich
 Angew. Chem. Int. Ed. 2018, 57, 10728-10731.

General Introduction	
Chapter 1: Coupling Reaction of Enol Derivatives with S	ilyl Ketene Acetals Catalyzed
by Gallium Trihalides	6
1-1. Introduction	6
1-2. Results and Discussion	7
1-3. Conclusion	
1-4. Experimental Section	
1-5. References	
Chapter 2: Indium Catalyzed Hydrofunctionalization of	Styrene Derivatives Bearing a
Hydroxy Group with Organosilicon Nucleophiles	
2-1. Introduction	
2-2. Results and Discussion	
2-3. Conclusion	
2-4. Experimental Section	
2-5. References	
Chapter 3: Selective Oxymetalation of Terminal Alkynes	via 6-Endo Cyclization:
Mechanistic Investigation and Application to the Efficien	t Synthesis of 4-Substituted
Isocoumarins	
3-1. Introduction	
3-2. Results and Discussion	
3-3. Conclusion	
3-4. Experimental Section	113
3-5. References	
Chapter 4: Synthesis of 2,4,6-Triarylpyryliums Bearing a	Carbon-Metal Bond and their
Application to Bifunctional Photocatalyst	
4-1. Introduction	
4-2. Results and Discussion	
4-3. Conclusion	
4-4. Experimental Section	
4-5. References	
Conclusion	

## Contents

#### **General Introduction**

Transformation of alkenes and alkynes is one of the important research topics in organic chemistry because alkenes and alkynes are fundamental materials and their transformation provides a basic and significant strategy to synthesize various organic molecules. Especially, the transformation via formation of carbon-carbon bonds is an essential method to access complex molecular structures (Scheme 1, upper). Chemists have developed a lot of useful reactions to form carbon-carbon bonds such as Mizoroki-Heck reaction, cross-couplings, alkene- or alkyne-metathesis, Sonogashira coupling and carbometalations.<sup>1</sup> The reactions with formation of carbon-oxygen bonds have also been studied (Scheme 1, lower). Addition reactions of alcohols or H<sub>2</sub>O to carbon-carbon multiple bonds and oxidation reactions such as epoxidation and Wacker process are known as representative examples.<sup>2</sup> The construction of heterocyclic compounds has recently attracted much attention in fine chemical synthesis.<sup>3</sup> However, development of new reaction systems, which can introduce functional groups into alkenes and alkynes directly with high selectivety, is highly desired even today in terms of efficient utilization of carbon resources.



Scheme 1. Transformation of alkenes and alkynes with formation of carbon-carbon or carbon-oxygen bonds.

In the synthesis of fine chemicals, the transformation of carbon-carbon multiple bonds mainly depends on transition metal catalysts. On the other hand, main group metals such as aluminum and boron have been utilized as Lewis acid catalysts for activation of carbonyl compounds.<sup>4</sup> However, transformation of unactivated alkenes and alkynes using main group metals has been less explored in

comparison with transition metals because main group metals have no d-orbitals interacting with  $\pi$  and  $\pi^*$  orbitals to show weak interaction with carbon-carbon multiple bonds (Scheme 2). If there is a novel concept that main group metals can activate carbon-carbon multiple bonds, new reaction systems can be developed because main group metals and transition metals show quite different reactivity.



Main group metal is less explored to transform alkenes or alkynes.

Scheme 2. Different character between transition metal and main group metal in the interaction with alkenes and alkynes.

Heavier group 13 metals such as gallium and indium have been utilized to functionalize carboncarbon multiple bonds.<sup>5</sup> As a pioneering example for gallium-mediated transformation, Yamaguchi reported Friedel-Crafts alkenylations via activation of trimethylsilylacetyrene using GaCl<sub>3</sub> (Eq. 1).<sup>6</sup>



Nakamura achived Conia-ene reaction of 1,3-dicarbonyl compounds with simple alkynes in the presence of  $In(OTf)_3$  (Eq. 2).<sup>7</sup> This reaction is one of the representative examples for indium-catalyzed transformation of unactivated alkynes, and the indium moiety of an In-enolate generated *in situ* worked as a Lewis acid to activate the alkynes.



For the formation of carbon-oxygen bonds, Sestelo developed InI<sub>3</sub>-catalyzed cyclization of 2alkynylphenols to synthesize benzofurans (Eq. 3).<sup>8</sup> In this report, theoretical calculation suggested that a dimer of InI<sub>3</sub> efficiently activates the alkyne as well as OH group.



Our group has also reported regio- and stereoselective carbometalation of alkenes or alkynes with silyl ketene acetals using indium or gallium salts (Eq. 4).<sup>9</sup>



These results indicated that gallium and indium show two distinctive properties, moderate Lewis acidity and high  $\pi$ -electron affinity, in contrast to the other group 13 metals (Scheme 3). If a metal salt has high Lewis acidity in the case of the carbometalation, the coordination of a silyl ketene acetal deactivates the Lewis acidic metal salt (1). However, heavier group 13 metal salts can dissociate from silyl ketene acetals due to the moderate Lewis acidity, and their high  $\pi$ -electron affinity can lead to activation of simple carbon-carbon multiple bonds (2) to accomplish the carbometalation. In other words, gallium and indium can work as a Lewis acid to transform unactivated alkenes and alkynes even in the presence of Lewis basic functional groups such as ether and carbonyl groups.



Scheme 3. Characteristic properties of heavier group 13 metal salts.

Based on the strategy of the transformation of carbon-carbon multiple bonds using heavier group 13 metal salts, I developed new synthetic reactions using alkenes and alkynes to form carbon-carbon or carbon-oxygen bonds. This thesis consists of the following four chapters.

Chapter 1 describes the GaBr<sub>3</sub>-catalyzed cross-coupling of enol derivatives with silyl ketene acetals (Eq. 5). This reaction is the first example of cross-coupling between enol derivatives and metal enolates. For the improvement of this coupling, expansion of scope of enol derivatives and mechanistic investigation such as determination of reaction intermediate, kinetic studies and theoretical calculations were carried out.





Chapter 2 deals with the InI<sub>3</sub>-catalyzed formal hydrofunctionalization of styrene derivatives bearing a hydroxy group with organosilicon nucleophiles (Eq. 6). Various types of functional groups were added to the less-polar alkenes in a single synthetic operation. The intramolecular hydroxy group of styrene derivatives plays an important role to accomplish this hydrofunctionalization.

#### Chapter 2: Indium Catalyzed Hydrofunctionalization of Styrene Derivatives Bearing a Hydroxy Group with Organosilicon Nucleophiles



Chapter 3 provides the 6-*endo* selective intramolecular oxyindation of terminal alkynes with an ester moiety to synthesize isocoumarins (Eq. 7). The regioselectivity is striking different with previously reported Lewis acid mediated cyclization of terminal alkynes in which 5-*exo* cyclization occurred due to the Markovnikov rule. Isocoumarins bearing a carbon-metal bond at 4-position were selectively synthesized and the metalated isocoumarins worked as a useful synthetic precursor of isocoumarin derivatives.

#### Chapter 3: Selective Oxymetalation of Terminal Alkynes via 6-Endo Cyclization: Mechanistic Investigation and Application to the Efficient Synthesis of 4-Substituted Isocoumarins



Chapter 4 reports the synthesis of triphenylpyryliums bearing a carbon-metal bond via intramolecular oxymetalation and its application to a photocatalyst possessing a Lewis acid moiety (Eq. 8). This new type of triphenylpyryliums showed a unique reactivity in a photo-induced reaction compared with a well-established triphenylpyrylium photocatalyst.





#### **References and Notes**

- (1) (a) I. P. Beletskaya, A. V. Cheprakov, *Chem. Rev.* 2000, *100*, 3009; (b) T. M. Trnka, R. H. Grubbs, *Acc. Chem. Res.* 2001, *34*, 18; (c) A. G. Fallis, P. Forgione, *Tetrahedron* 2001, *57*, 5899; (d) K. C. Nicolaou, P. G. Bulger, D. Sarlah, *Angew. Chem. Int. Ed.* 2005, *44*, 4442; (e) R. Chinchilla, C. Nájera, *Chem. Soc. Rev.* 2011, *40*, 5084.
- (2) (a) K. Burgess, M. J. Ohlmeyer, *Chem. Rev.* 1991, *91*, 1179; (b) Q.-H. Xia, H.-Q. Ge, C.-P. Ye, Z.-M. Liu, K.-X. Su, *Chem. Rev.* 2005, *105*, 1603; (c) S. E. Mann, L. Benhamou, T. D. Sheppard, *Synthesis* 2015, *47*, 3079; (d) N. Tsuji, J. L. Kennemur, T. Buyck, S. Lee, S. Prévost, P. S. J. Kaib, D. Bykov, C. Farès, B. List, *Science* 2018, *359*, 1501.
- (3) (a) F. Alonso, I. P. Beletskaya, M. Yus, *Chem. Rev.* 2004, 104, 3079. (b) R. Dorel, A. M. Echavarren, *Chem. Rev.* 2015, 115, 9028.
- (4) L. C. Wilkins, R. L. Melen, Coordin. Chem. Rev. 2016, 324, 123.
- (5) (a) C. Bour, V. Gandon, *Synlett* 2015, 26, 1427. (b) S. R. Pathipati, A. van der Werf, N. Selander, *Synthesis* 2017, 49, 4931. (c) J. P. Sestelo, L. A. Sarandeses, M. M. Martínez, L. Alonso-Marañón, *Org. Biomol. Chem.* 2018, 16, 5733.
- (6) M. Yamaguchi, Y. Kido, A. Hayashi, M. Hirama, Angew. Chem. Int. Ed. Engl. 1997, 36, 1313.
- (7) (a) M. Nakamura, K. Endo, E. Nakamura, J. Am. Chem. Soc. 2003, 125, 13002. (b) K. Endo, T. Hatakeyama, M. Nakamura, E. Nakamura, J. Am. Chem. Soc. 2007, 129, 5264.
- (8) L. Alonso-Marañón, M. M. Martínez, L. A. Sarandeses, E. Gómez-Bengoa, J. P. Sestelo, *J. Org. Chem.* 2018, 83, 7970.
- (9) (a) Y. Nishimoto, R. Moritoh, M. Yasuda, A. Baba, *Angew. Chem. Int. Ed.* 2009, 48, 4577. (b) Y. Nishimoto, H. Ueda, Y. Inamoto, M. Yasuda, A. Baba, *Org. Lett.* 2010, 12, 3390. (c) Y. Nishimoto, H. Ueda, M. Yasuda, A. Baba, *Chem. Eur. J.* 2011, 17, 11135. Allylsilanes were used instead of silyl ketene acetals: (d) Y. Nishimoto, J. Yi, T. Takata, A. Baba, M. Yasuda, *Molecules* 2018, 23, 1884.

#### **Chapter 1**

### **Coupling Reaction of Enol Derivatives with Silyl Ketene Acetals Catalyzed by Gallium Trihalides**

#### 1-1. Introduction

The cross-coupling reaction using alkenyl electrophiles is one of the most important methods for the construction of carbon frameworks such as pharmaceuticals, natural products, and organic materials.<sup>1</sup> Generally, alkenyl halides are used as an electrophilic coupling partner because of their high reactivity, while they have some problems such as relatively high cost and potential toxicity, as well as the formation of metal halide salts as by-products. In this context, enol derivatives such as enol ethers and alkenyl carboxylates have attracted much attention because they are inexpensive, readily available, and environmentally benign (Scheme 1A).<sup>2</sup> However, a coupling reaction using enol derivatives remains a challenging issue because the intrinsic feature of inert C-O bonds disturbs the achievement of desirable coupling reactions. Although several types of transition metal-catalyzed couplings using enol ethers,<sup>3</sup> alkenyl carboxylates,<sup>4</sup> alkenyl carbamates,<sup>5</sup> alkenyl phosphates,<sup>6</sup> and silvl enol ethers<sup>7</sup> have been reported, applicable organometallic partners have been limited to alkyl or aryl metals (Scheme 1B). As far as other types of nucleophiles, such as metal enolates, only two catalytic coupling reactions, the Pdcatalyzed dimerization of silvl vinyl ethers<sup>8</sup> and our reported InBr<sub>3</sub>-catalyzed coupling of enol ethers with ketene silyl imines,<sup>9</sup> have previously been achieved.<sup>10</sup> To the best of our knowledge, the cross coupling between enol derivatives and metal enolates had never been achieved prior to our previous report of a GaBr<sub>3</sub>-catalyzed coupling between enol ethers and silvl ketene acetals.<sup>11,12</sup>



Scheme 1. Coupling reaction using enol derivatives as electrophiles.

Herein, we report the full details of the coupling reaction of enol derivatives such as enol ethers and alkenyl carboxylates with silyl ketene acetals catalyzed by GaBr<sub>3</sub> (Scheme 1B). This paper (1) elucidates the scope of enol derivatives and silyl ketene acetals, and (2) describes the mechanistic study via kinetic experiments, the observation of a key intermediate, and theoretical calculations. Typical transition metal-catalyzed coupling reactions using enol derivatives proceed via a mechanism that includes oxidative addition/reductive elimination (Scheme 2A).<sup>3,4,5,6,7</sup> Recently, Kambe et al. reported Rh-catalyzed coupling between enol ethers and Grignard reagents via the *syn* addition of an organorhodium/*anti* elimination of a rhodium alkoxide (Scheme 2B).<sup>3i</sup> We revealed that the present coupling reaction by three components, GaBr<sub>3</sub>, an enol derivative, and a silyl ketene acetal, followed by *syn* elimination of a gallium alkoxide (Scheme 2C).



Scheme 2. Comparison of reaction mechanisms of coupling reaction using enol derivatives.

#### 1-2. Results and Discussion

#### **Coupling Reaction of Enol Ethers with Silyl Ketene Acetals**

# (i) Optimization of Reaction Conditions in the Coupling Reaction between Enol Ether and Silyl Ketene Acetal

We recently developed carbometalations of alkenes and alkynes using silyl ketene acetals and metal halides such as InBr<sub>3</sub><sup>13</sup>, GaBr<sub>3</sub><sup>14</sup>, and BiBr<sub>3</sub><sup>15</sup>. The moderate Lewis acidity of these metal halides promoted the effective activation of carbon-carbon multiple bonds without deactivation by the coordination of an oxygen atom of ketene silyl acetates to achieve carbometallation in an *anti*-addition

fashion (Scheme 3A). During the course of a study on this carbometallation, when ethyl vinyl ether was employed as an alkene the predictable alkylindium compound was not obtained, but the corresponding coupling product was obtained quantitatively (Scheme 3B). The generation of Me<sub>3</sub>SiBr and InBr<sub>2</sub>(OEt) via carbogallation and  $\beta$ -elimination from the alkylindium compound, respectively, should be possible based on these results. Therefore, we investigated the catalytic coupling of enol ethers with silyl ketene acetals in expectation of the regeneration of InBr<sub>3</sub> by the transmetallation between Me<sub>3</sub>SiBr and InBr<sub>2</sub>(OEt).



Scheme 3. Unexpected coupling reaction based on our developed carbometalation.

We examined the effect of catalysts on the coupling of butyl vinyl ether **1a** with dimethylketene trimethylsilyl acetal **2a** in dichloroethane at 80 °C (Table 1). At first, this coupling reaction did not occur without a catalyst (entry 1). A survey of InBr<sub>3</sub>, GaBr<sub>3</sub>, and BiBr<sub>3</sub>, that effectively promoted carbometallation,<sup>13,14,15</sup> was carried out. A catalytic amount of InBr<sub>3</sub> (10 mol%) smoothly accelerated the desired coupling reaction to afford  $\alpha$ -vinyl ester **3aa** in 86% yield (entry 2). The reaction catalyzed by GaBr<sub>3</sub> gave an excellent yield, 97% (entry 3). In contrast, BiBr<sub>3</sub> showed a low catalytic effect (entry 4). Other indium and gallium salts such as InCl<sub>3</sub>, InI<sub>3</sub>, In(OTf)<sub>3</sub>, GaCl<sub>3</sub>, GaI<sub>3</sub>, and Ga(OTf)<sub>3</sub> effectively accelerated the coupling reaction to give the product **3aa** (entries 5-10), while other bismuth salts were ineffective (entries 11-13). When employing indium and gallium salts at a lower temperature (50 °C), GaBr<sub>3</sub> and GaI<sub>3</sub> showed high catalytic ability (entries 14-19). While ZnCl<sub>2</sub> showed modest catalytic ability (entry 20), typical strong Lewis acids such as BF<sub>3</sub>•OEt<sub>2</sub>, AlCl<sub>3</sub>, and TiCl<sub>4</sub> were ineffective (entries 21-23). Sc(OTf)<sub>3</sub> and Y(OTf)<sub>3</sub> also produced no desired product (entries 24 and 25). Transition metal salts did not work as catalysts (entries 26 and 27). These results show that the characteristic features of

gallium and indium halides are important for the completion of a catalytic reaction system. The effect of solvents was also investigated. As far as non-polar solvents, toluene can be employed though the reaction in hexane gave low yield (entries 28 and 29). Polar solvents such as CH<sub>3</sub>CN and THF were unsuitable because they coordinated with GaBr<sub>3</sub> to disturb the reaction (entries 30 and 31). Finally, GaI<sub>3</sub> and GaBr<sub>3</sub> were the most effective catalysts, and, thus, we decided to use entry 3 as the standard condition.

	OSiMe	<sup>e</sup> 3 Catalyst (10 mol	%)
🥟 OBu	ТОВ	u Solvent	ОВи
1a	' 2a	80 °C, 2 h	3aa
Entry	Catalyst	Solvent	Yield of <b>3aa</b> [%] <sup>[b]</sup>
1	none	ClCH <sub>2</sub> CH <sub>2</sub> Cl	0
2	InBr <sub>3</sub>	ClCH <sub>2</sub> CH <sub>2</sub> Cl	86
3	GaBr <sub>3</sub>	ClCH <sub>2</sub> CH <sub>2</sub> Cl	97
4	BiBr <sub>3</sub>	ClCH <sub>2</sub> CH <sub>2</sub> Cl	trace
5	GaCl <sub>3</sub>	ClCH <sub>2</sub> CH <sub>2</sub> Cl	100
6	GaI <sub>3</sub>	ClCH <sub>2</sub> CH <sub>2</sub> Cl	96
7	Ga(OTf) <sub>3</sub>	ClCH <sub>2</sub> CH <sub>2</sub> Cl	64
8	InCl <sub>3</sub>	ClCH <sub>2</sub> CH <sub>2</sub> Cl	100
9	InI <sub>3</sub>	ClCH <sub>2</sub> CH <sub>2</sub> Cl	94
10	In(OTf) <sub>3</sub>	ClCH <sub>2</sub> CH <sub>2</sub> Cl	92
11	BiCl <sub>3</sub>	ClCH <sub>2</sub> CH <sub>2</sub> Cl	trace
12	BiI <sub>3</sub>	ClCH <sub>2</sub> CH <sub>2</sub> Cl	trace
13	Bi(OTf) <sub>3</sub>	ClCH <sub>2</sub> CH <sub>2</sub> Cl	17
14 <sup>[c]</sup>	GaCl <sub>3</sub>	ClCH <sub>2</sub> CH <sub>2</sub> Cl	52
15 <sup>[c]</sup>	GaBr <sub>3</sub>	ClCH <sub>2</sub> CH <sub>2</sub> Cl	95
16 <sup>[c]</sup>	GaI <sub>3</sub>	ClCH <sub>2</sub> CH <sub>2</sub> Cl	94
17 <sup>[c]</sup>	InCl <sub>3</sub>	ClCH <sub>2</sub> CH <sub>2</sub> Cl	trace
18 <sup>[c]</sup>	InBr <sub>3</sub>	ClCH <sub>2</sub> CH <sub>2</sub> Cl	21

Table 1. Investigation of effects of catalyst and solvents.<sup>[a]</sup>

19 <sup>[c]</sup>	InI <sub>3</sub>	ClCH <sub>2</sub> CH <sub>2</sub> Cl	59
20	ZnCl <sub>2</sub>	ClCH <sub>2</sub> CH <sub>2</sub> Cl	59
21	$BF_3 \cdot OEt_2$	ClCH <sub>2</sub> CH <sub>2</sub> Cl	2
22	AlCl <sub>3</sub>	ClCH <sub>2</sub> CH <sub>2</sub> Cl	trace
23	TiCl <sub>4</sub>	ClCH <sub>2</sub> CH <sub>2</sub> Cl	trace
24	Sc(OTf) <sub>3</sub>	ClCH <sub>2</sub> CH <sub>2</sub> Cl	trace
25	Y(OTf) <sub>3</sub>	ClCH <sub>2</sub> CH <sub>2</sub> Cl	0
26	CuBr <sub>2</sub>	ClCH <sub>2</sub> CH <sub>2</sub> Cl	trace
27	PdCl <sub>2</sub>	ClCH <sub>2</sub> CH <sub>2</sub> Cl	trace
28	GaBr <sub>3</sub>	toluene	71
29	GaBr <sub>3</sub>	hexane	24
30	GaBr <sub>3</sub>	CH <sub>3</sub> CN	11
31	GaBr <sub>3</sub>	THF	20

[a] 1a (1 equiv), 2a (1.5 equiv), catalyst (0.1 equiv), solvent (0.5 M),

80 °C, 2 h. [b] Yields were determined by GC. [c] 50 °C, 30 min.

#### (ii) Scope of Ketene Silyl Acetals and Enol Ethers

The scope of ketene silyl acetals **2** for the GaBr<sub>3</sub>-catalyzed coupling was investigated, and the results are summarized in Table 2. Dialkylketene silyl acetals **2b**, **2c**, and **2d** were more bulky than **2a**, and afforded the corresponding  $\alpha$ -vinyl esters in high yields (entries 1-3). Alkyl arylketene silyl acetal **2e** was also a suitable nucleophile (entry 4). As far as a monosubstituted ketene silyl acetal, reactions using various types of alkylketene and arylketene silyl acetals gave the desired products in high yields (entries 5-7). The reaction with unsubstituted ketene silyl acetal **2i** resulted in a low yield because **2i** would decompose under these conditions (entry 8). Heteroaryl-substituted ketene silyl acetals **2j** and **2k** worked as effective nucleophiles (entries 9 and 10). *O*- and *S*-substituted ketene silyl acetals **2l** and **2m** were applicable to give moderate yields of the corresponding functionalized  $\alpha$ -vinyl esters **3al** and **3am**, respectively (entries 11 and 12).

∕∕OBu 1a	$H^{+} \xrightarrow{R^{1} OSiMe_{3}}{R^{2}} \xrightarrow{R^{2}} 2$	GaBr <sub>3</sub> (10 mol%) CICH <sub>2</sub> CH <sub>2</sub> CI 80 °C, 2 h R <sup>1</sup>	CO <sub>2</sub> R <sup>3</sup> R <sup>2</sup> 3
Entry	Silyl Ketene Acetal <b>2</b>	Product 3	Yield [%] <sup>[b]</sup>
1	OSiMe <sub>3</sub> Me <sub>v</sub> OMe Et <b>2b</b>	Me Et <b>3ab</b>	92
2	$\begin{array}{c} \text{OSiMe}_3\\ \text{Et} \\ \text{Et} \\ \text{CEt} \\ CE$	Et Et <b>3ac</b>	93
3	OSiMe <sub>3</sub> OMe 2d	CO <sub>2</sub> Me	90
4	OSiMe <sub>3</sub> Me <sub>y</sub> Ph <b>2e</b>	Me Ph <b>3ae</b>	81
5	OSiMe <sub>3</sub> OMe Bu <b>2f</b>	Bu <b>3af</b>	82
6	OSiMe <sub>3</sub> OEt <i>t</i> Bu <b>2g</b>	CO <sub>2</sub> Et <i>t</i> Bu <b>3ag</b>	43
7	OSiMe <sub>3</sub> OMe Ph <b>2h</b>	Ph <b>3ah</b>	83
8[c]	OSiMe <sub>3</sub>	CO <sub>2</sub> Hex 3ai	12
9	OSiMe <sub>3</sub> OEt	S 3aj	80
10	OSiMe <sub>3</sub> OMe	CO <sub>2</sub> Me N Me <b>3ak</b>	84

Table 2. Scope of silyl ketene acetal 2 in coupling reaction of butyl vinyl ether 1a.<sup>[a]</sup>



[a] 1a (1 equiv), 2 (1.5 equiv), GaBr<sub>3</sub> (0.1 equiv), ClCH<sub>2</sub>CH<sub>2</sub>Cl,
80 °C, 2 h. [b] Isolated yields. [c] 2 (3 equiv), GaBr<sub>3</sub> (0.2 equiv).

Next, we examined the scope of enol ethers, as shown in Table 3. A variety of  $\alpha$ -substituted enol ethers **4** were found to be suitable substrates. The reaction of alkyl-substituted enol ethers **4a** and **4b** smoothly occurred to give the desired products **5an** and **5bn** in 77 and 69% yields, respectively (entries 1 and 2). Functional groups such as internal alkene, chloro, and alkoxy moieties tolerated these reaction conditions, and a loading of 30 mol% of GaCl<sub>3</sub> rather than GaBr<sub>3</sub> effectively accelerated the reactions of substrates **4c**, **4d**, and **4e** to give the corresponding products **5cn**, **5dn**, and **5en**, respectively (entries 3-5).<sup>16</sup> Methoxy enyne **4f** also reacted with ketene silyl acetal to smoothly produce the ester **5fn** with an enyne moiety in 73% yield (entry 6). Aryl-substituted enol ethers **4g**, **4h**, **4i**, and **4j** were applicable to the coupling reaction, although the yields were moderate (entries 7-10). The isopropyl vinyl ether **1b** as well as the butyl vinyl ether **1a** were applicable to this reaction system, though the yield was moderate (entry 11). In contrast, the phenyl vinyl ether **1c** gave a high yield in spite of the steric hindrance of **1c** that was similar to that of isopropyl vinyl ether **1b** (entry 12).<sup>17</sup> Therefore, the leaving ability of an alkoxy group influenced the reaction rate.



Table 3. Scope of enol ethers in coupling reaction with silyl ketene acetal 2.<sup>[a]</sup>



[a] 1 or 4 (1 equiv), 2n (1.5 equiv), GaBr<sub>3</sub> (0.1 equiv), ClCH<sub>2</sub>CH<sub>2</sub>Cl (0.5 M), 2 h, 80 °C. [b] Isolated yields. [c] Reaction time 12 h. [d] GaCl<sub>3</sub> (30 mol%), 2n (3 equiv).
[e] 2a (1.5 equiv). [f] Yields were determined by GC.

#### Coupling of Vinyl Carboxylates with Silyl Ketene Acetals

Vinyl carboxylates as well as enol ethers are very valuable as alkenylation reagents because they are more stable and easier to handle than enol ethers.<sup>4</sup> We investigated the coupling reaction of various vinyl carboxylates **6** with silyl ketene acetal **2a** (Table 4). The reaction of vinyl acetate **6a** with silyl ketene acetal **2a** at 80 °C in the presence of GaBr<sub>3</sub> catalyst gave the coupling product **3aa** in 29% yield (entry 1). Vinyl 2-ethylhexanoate **6b**, vinyl pivalate **6c**, and vinyl benzoate **6d**, which possess the steric hindrance of a carboxylate moiety, afforded high yields (entries 2-4). The low yield shown in entry 1 can likely be ascribed to Claisen condensation between vinyl acetate **6a** and silyl ketene acetal **2a**.<sup>18</sup>

Table 4. GaBr<sub>3</sub>-Catalyzed coupling reaction of vinyl carboxylate 6 with silyl ketene acetal 2a.<sup>[a]</sup>

6	$ \bigcup_{R}^{O} + \underbrace{2a}^{OS} $	iMe <sub>3</sub> <u>G</u> a OBu	aBr <sub>3</sub> (10 mol%) CICH <sub>2</sub> CH <sub>2</sub> CI 80 °C, 2 h <b>3aa</b>
Entry	R		Yield of <b>3aa</b> [%] <sup>[b]</sup>
1	Me	6a	29
2	CH(Et)Bu	6b	84
3	<i>t</i> Bu	6c	80
4	Ph	6d	74

[a] 6 (1 equiv), 2a (1.5 equiv), GaBr<sub>3</sub> (0.1 equiv), ClCH<sub>2</sub>CH<sub>2</sub>Cl (0.5 M), 80 °C, 2 h. [b] Yields were determined by <sup>1</sup>H NMR.

The scope of silyl ketene acetals **2** was surveyed in the reaction of vinyl pivalate **6c** (Table 5). Dialkylketene silyl acetals **2d** and **2c** were applicable to the coupling reaction of vinyl pivalate **6c** while bulky diethylketene silyl acetal **2c** gave only 22% yield (entries 1 and 2). The reaction using phenyl-substituted ketene silyl acetals **2e** and **2h** afforded the corresponding products in moderated yields (entries 3 and 4). Thiophenyl-substituted ketene silyl acetal **2j** also underwent cross coupling to give a high yield (entry 5). On the whole, the yields of the reaction using vinyl carboxylate **6** were slightly lower than those of the reaction using vinyl ether **1**. Therefore, vinyl carboxylate **6** had lower reactivity than vinyl ether **1** in the present coupling reaction, although the yields were satisfactory.

OSiMe<sub>2</sub>  $CO_2R^3$ GaBr<sub>3</sub> (10 mol% CICH<sub>2</sub>CH<sub>2</sub>CI  $R^{1}R^{2}$ 80 °C, 2 h 6c 3 Yield (%)<sup>[b]</sup> Entry Silyl Ketene Product Acetal 3 2 79 1 OSiMe<sub>3</sub> CO<sub>2</sub>Me OMe 3ad 2d 2 22 OSiMe<sub>3</sub> CO<sub>2</sub>Et F `OEt Εť Èt 3ac έt 2c 3 51 OSiMe<sub>3</sub> CO<sub>2</sub>Me `OMe Ър Mé 3ae Ph 2e 4 63 OSiMe<sub>3</sub> CO<sub>2</sub>Me OMe Ρ'n 3ha 2h 83 5 OSiMe<sub>3</sub> CO<sub>2</sub>Et OFt 3ja 2j

Table 5. Scope of silvl ketene acetal 2 in reaction of vinyl pivalate 6c.<sup>[a]</sup>

[a] 6c (1 equiv), 2 (1.5 equiv), GaBr<sub>3</sub> (0.1 equiv), ClCH<sub>2</sub>CH<sub>2</sub>Cl (0.5 M),

80 °C, 2 h. [b] Yields were determined by <sup>1</sup>H NMR.

#### Monitoring of Coupling Reaction between Vinyl Ether and Silyl Ketene Acetal

To reveal the reaction mechanism, we monitored the coupling reaction of vinyl ether **1** with silyl ketene acetal **2** via NMR spectroscopy. When conducting the coupling reaction among vinyl ether **1d**, the silyl ketene acetal **2n**, and GaBr<sub>3</sub> at room temperature, the alkylgallium intermediate **7**, coupling product **3dn**, and Me<sub>3</sub>SiBr were observed via <sup>1</sup>H NMR after 5 min (Figure 1).<sup>19</sup> The methoxy signal of intermediate **7** appeared at 4.03 ppm (0.37 ppm downfield shift in comparison with that of coupling product **3dn**). This result indicated the intramolecular coordination of a carbonyl group to gallium to form a six-membered ring.<sup>20</sup> Vinyl ether **1d** was consumed after 25 min, and then the intermediate **7** was transformed into the coupling product **3dn** after 70 min. To avoid an overlapping of the signals in the spectrum, the reaction of phenyl vinyl ether **1c** with deuterated silyl ketene acetal **2o** was conducted (Scheme 4). Alkylgallium intermediate **8** was successfully observed without the corresponding coupling

product under low-temperature conditions.<sup>21</sup> This result suggested that the elimination from the alkylgallium intermediate **8** to give the coupling product was slower than the carbogallation. Coupling constant values,  ${}^{2}J$  (H<sup>1a</sup>-H<sup>1e</sup>) = 14 Hz,  ${}^{3}J$  (H<sup>1a</sup>-H<sup>2</sup>) = 3 Hz, and  ${}^{3}J$  (H<sup>1e</sup>-H<sup>2</sup>) = 4 Hz, of alkylgallium **8** clearly show that the PhO group occupies the axial position in the chair-like six-membered ring due to anomeric effect.<sup>22,23</sup> In contrast to vinyl ether **1**, the intermediate from vinyl carboxylate **6c** was not observed even at low temperature and only starting materials and the coupling product were confirmed.



**Figure 1.** Monitoring of the coupling reaction of vinyl ether **1d** with silyl ketene acetal **2n** in the presence of GaBr<sub>3</sub> by <sup>1</sup>H NMR. GaBr<sub>3</sub> (0.5 equiv), **1d** (1 equiv), **2n** (1.5 equiv), CD<sub>2</sub>Cl<sub>2</sub>, room temperature.



Scheme 4. Observation of alkylgallium intermediate 8 in the reaction of vinyl ether 1c with deuterated silyl ketene acetal 20 by <sup>1</sup>H NMR.

#### Mechanistic Studies on Coupling Reaction of Enol Derivatives with Silyl Ketene Acetal

Kinetic studies were carried out to gain insight into the turnover-limiting step of the GaBr<sub>3</sub>-catalyzed coupling reaction of vinyl ether **1** with silyl ketene acetal **2**. We investigated the dependence of the initial rate of the GaBr<sub>3</sub>-catalyzed coupling reaction of vinyl ether **1a** with silyl ketene acetal **2a** in dichloroethane at 30 °C on the varied concentrations of each reagent (Figure 2). The plots of the initial rates vs. the concentrations of GaBr<sub>3</sub>, vinyl ether **1a**, and ketene silyl acetal **2a** are shown in Figure 2.

The kinetic analysis shows that this coupling reaction was of the zeroth order for vinyl ether **1a** and silyl ketene acetal **2a**, and was of the first order for GaBr<sub>3</sub> (rate =  $k_{obs}$ [GaBr<sub>3</sub>]). Next, we performed kinetic studies for the GaBr<sub>3</sub>-catalyzed coupling between vinyl pivalate **6c** and silyl ketene acetal **2a**, and the results are shown in Figure 3. In contrast to vinyl ether **1a**, a coupling reaction using vinyl pivalate **6c** was of the first order for vinyl pivalate **6c** and the GaBr<sub>3</sub> catalyst, and was of the zeroth order for silyl ketene acetal (rate =  $k_{obs}$ [GaBr<sub>3</sub>][**6c**]). Therefore, the leaving group had a significant effect on the turnover-limiting step.



**Figure 2.** Initial rate kinetics of GaBr<sub>3</sub>-catalyzed coupling reaction of butyl vinyl ether **1a** with silyl ketene acetal **2a**.



**Figure 3.** Initial rate kinetics of GaBr<sub>3</sub>-catalyzed coupling reaction of vinyl pivalate **6c** with ketene silyl acetal **2a**.

A plausible mechanism for the coupling reaction using vinyl ether **1** based on the kinetic studies and a confirmation of the alkylgallium intermediate is illustrated in Scheme 5.<sup>24</sup> GaBr<sub>3</sub> is coordinated by not

only vinyl ether 1 but also silyl ketene acetal 2 so that the equilibrium between the GaBr<sub>3</sub>-vinyl ether complex (GaBr<sub>3</sub>·1) and the GaBr<sub>3</sub>-ketene silyl acetal complex (GaBr<sub>3</sub>·2) should be considered. The interaction between GaBr<sub>3</sub> and the olefin moiety of vinyl ether 1 increases the positive charge at the  $\alpha$ carbon of oxygen. The positive charge is stabilized by the OR group. The nucleophilic attack of silyl ketene acetal 2 regioselectively occurs in an *anti*-addition fashion (A) to provide  $\beta$ -oxyalkylgallium intermediate 9 and Me<sub>3</sub>SiBr. The gallium atom is coordinated by the intramolecular carbonyl oxygen to form a six-membered ring. The OR group coordinates to gallium to kick out the ester moiety, and then syn-elimination (10) between GaBr<sub>2</sub> and the OR group proceeds to give the coupling product  $3^{25}$  Finally, the transmetalation of  $Br_2Ga(OR)$  11 with Me<sub>3</sub>SiBr regenerates the GaBr<sub>3</sub> catalyst. Recently, we reported that carbogallation, carboindation, and carbobismuthination of alkenes/alkynes using silyl ketene acetals proceeded in an anti-addition manner.<sup>13,14,15</sup> In the D-labeled experiment, the E-isomer of deuterated phenyl vinyl ether 1c-d reacted with 2a to exclusively furnish E-isomer product 3aa-d with the retention of its stereochemical configuration (Eq. 1). This result suggests that the coupling reaction using vinyl ether 1 occurs stereospecifically. Therefore, the proposed anti-addition/syn-elimination is supported by our reported carbometalations and the D-labeled experiment. The elimination process (9)  $\rightarrow$  10  $\rightarrow$  3 and 11) is a turnover-limiting step because of the dependence of the reaction rate on only the concentration of GaBr<sub>3</sub> (rate =  $k_{obs}$ [GaBr<sub>3</sub>]) (Figure 2), which was also shown by observation of the alkylgallium intermediate 9 (Figure 1 and Scheme 4).<sup>26</sup>

The coupling reaction of vinyl carboxylate 6 would also proceed via the anti-carbogallation/synelimination mechanism illustrated in Scheme 6. The mechanism of vinyl carboxylate 6 significantly differs from that of vinyl ether 1 in that the syn-elimination occurs via the six-membered ring intermediate 14 including the coordination to gallium of the carbonyl oxygen of the leaving group. The turnover-limiting step is either the coordination of vinyl carboxylate  $\mathbf{6}$  to GaBr<sub>3</sub> or the carbogallation because the rate equation is presented by  $k_{obs}$  [GaBr<sub>3</sub>][6] (Figure 3) and alkylgallium intermediates 12 and 13 could not be observed in monitoring the coupling reaction via NMR spectroscopy. Bulky silyl ketene acetals such as diethyl-, methylphenyl-, and phenylsubstituted ones (2c, 2e, and 2h) had lower reactivities than dimethylsubstituted one 2a (entry 3 in Table 4, and entries 2, 3, and 4 in Table 5), suggesting that silvl ketene acetal 2 participates in the turnover-limiting step. Therefore, we assumed that silvl ketene acetal 2 has the inhibition effect via the coordination to GaBr<sub>3</sub> (GaBr<sub>3</sub>• $6 + 2 \rightarrow 6 +$ GaBr<sub>3</sub>•2), and that carbogallation among GaBr<sub>3</sub>, vinyl carboxylate 6, and silyl ketene acetal 2 is the turnover limiting step. Actually, the experimental rate equation, rate =  $k_{obs}$ [GaBr<sub>3</sub>][6] is consistent with the theoretical rate equation, rate =  $k_1k_2k_{-1}$  [GaBr<sub>3</sub>][6].<sup>27</sup> In contrast to vinyl ether 1, the conformational change from the six-membered ring of intermediate 12 to the six-membered ring of intermediate 13 and  $\beta$ -elimination via a six-membered cyclic transition state would easily occur. On the other hand, the

diminished stabilization effect of a positive charge by the carboxylate group slowed the carbogallation step. Therefore, the turnover-limiting step in the coupling reaction of vinyl carboxylate **6** was different from that of vinyl ether **1**.



Scheme 5. Plausible mechanism for coupling of vinyl ether 1 with silyl ketene acetal 2.





Scheme 6. Plausible mechanism for coupling of vinyl carboxylate 6 with silyl ketene acetal 2.

#### Investigation of Elimination Process in Coupling Reaction of Vinyl Ether 1

The conformational change  $(9 \rightarrow 10)$  in the turnover-limiting step of the coupling reaction of vinyl ether 1 (Scheme 5) seemed unlikely to occur because the stable six-membered ring structure via the coordination of the carbonyl group to gallium transforms into a strained four-membered ring structure. Therefore, we investigated our proposed elimination process  $(9 \rightarrow 10 \rightarrow 3 + 11)$  via theoretical calculation to verify the possibility of this conformational change and to consider the influence of conformational change and *syn*-elimination  $(10 \rightarrow 3 + 11)$  on the turnover-limiting step. Figure 4 shows the potential energy profile of the elimination process from alkylgallium intermediate **9a** which is afforded by the reaction of butyl vinyl ether **1a** with silyl ketene acetal **2n**. In the conformational change from the stable six-membered ring structure of **9a** to the strained four-membered ring structure of **10a**, the dissociation of the carbonyl group and the association of the alkoxy group occur in concert (**TS9a**). As expected, the four-membered ring structure of **10a** is less stable than the six-membered ring structure of **9a**. The activation energy of the path from **9a** is 14.3 kcal/mol, which supports the possibility

of this conformational change. The elimination pathway from the four-membered ring structure of **10a** to **17a** via transition state **TS10a** is exothermic and requires activation energies of 13.6 kcal/mol. The important bond lengths (Ga-C<sup>1</sup>, Ga-O, C<sup>1</sup>-C<sup>2</sup>, and C<sup>2</sup>-O) of intermediate **10a** and the transition state of **TS10a** are shown in Figure 5. The C<sup>2</sup>-O bond is elongated from 1.48 Å to 2.08 Å, and the degree of its length change is larger than that of the other bonds. The Ga-O and C<sup>1</sup>-C<sup>2</sup> bonds are shortened. On the other hand, the Ga-C<sup>1</sup> bond distance is only minutely changed. Therefore, the cleavage of the C<sup>2</sup>-O bond by the assistance of the gallium is precedent in the elimination step. The overall potential energy surface of the elimination process showed that both the conformational change and the elimination step require a large energy barrier in order to influence the reaction rate. These theoretical results suggested that the moderate Lewis acidity of gallium is important for the conformational change from the stable sixmembered ring structure of **9a** to the strained four-membered ring structure of **10a**, and for the *syn*-elimination with the cleavage of the carbon-oxygen bond. If the Lewis acidity was strong, such a flexible conformational change would not occur, and if the Lewis acidity was weak, the carbon-oxygen bond would not be cleaved.



Figure 4. Potential energy profile of the elimination step from 9a to 17a in the reaction of vinyl ether 1a with silyl ketene acetal 2n.



Figure 5. Optimized structures and some bond lengths in intermediate 10a and transition state TS10a (distances in Å).

Next, we considered the rate equation of the coupling reaction between vinyl ether **1** and silyl ketene acetal **2** on the basis of mechanistic studies. Theoretical calculation analysis revealed that the conformational change from a four-membered ring to a six-membered ring  $(10 \rightarrow 9)$  proceeds via a much smaller energy barrier than that required for the elimination of GaBr<sub>2</sub>(OR) so that the elimination process involves an equilibrium between **9** and **10** that lies toward **9** (Eq. 2). Therefore, assuming a steady state for **10** allows the theoretical rate equation of the production of **3** to be expressed as rate =  $k_1k_2k_{-1}^{-1}$ [GaBr<sub>3</sub>], which is consistent with the experimental rate equation (Eq. 3).<sup>27</sup> And, the constant term  $k_1k_2k_{-1}^{-1}$  suggested that both conformational change and *syn*-elimination steps determine the reaction rate. Using equation 3, we obtained excellent fits with  $k_{obs}$  (<sub>303 K</sub>) = 4.2 x 10<sup>-4</sup> s<sup>-1</sup> for the production of **3** in the catalytic run using different amounts of GaBr<sub>3</sub>, as shown in Figure 3. An Arrhenius plot using  $k_{obs}$ s at some temperatures experimentally gave an activation energy of 15.6 kcal/mol.<sup>28</sup> This activation energy approximated the computed activation energy (Figure 6), which suggested that the elimination process revealed by theoretical calculation is probable.



#### 1-3. Conclusion

We have established a novel coupling reaction of enol derivatives with silvl ketene acetals catalyzed by GaBr<sub>3</sub> catalyst. Various types of enol ethers such as vinyl ether 1 and  $\alpha$ -substituted enol ether 4 were applicable to this reaction system. Functionalized enol ethers possessing chloro, olefin, and ether moieties afforded the corresponding products smoothly. We successfully demonstrated the wide scope of suitable ketene silvl acetals 2 via the use of those bearing phenoxy, thiophenyl, indolyl, and thiophenyl groups. Vinyl carboxylate 6 as well as vinyl ether 1 were also found to be feasible alkenylating reagents. In this paper, we revealed the details of the reaction mechanism based on the observation of intermediates, kinetic studies, and computational analysis. As for the reaction of vinyl ether 1, we observed the alkylgallium intermediate of a reaction of vinyl ether 1 with silvl ketene acetal 2 via <sup>1</sup>H-NMR spectroscopy, which strongly supported the carbogallation/elimination mechanism. Kinetic studies have shown that the coupling reaction of vinyl ether 1 is on the order of zero for enol ether 1 and silvl ketene acetal 2, and is of the first order for GaBr<sub>3</sub> (rate =  $k_{obs}$ [GaBr<sub>3</sub>]) and that the turnover-limiting step is the elimination process. On the other hand, the rate equation of the reaction of vinyl carboxylate **6** can also be represented as rate =  $k_{obs}$ [GaBr<sub>3</sub>][**6**], and the turnover-limiting step involved carbogallation. We investigated the details of the  $\beta$ -elimination process in a coupling reaction using vinyl ether 1 via computational analysis in order to show that the moderate Lewis acidity of GaBr3 contributes to the flexible conformational change of the alkylgallium intermediate from a stable six-membered ring to a strained four-membered ring and syn-elimination with the C-O bond cleavage.

#### 1-4. Experimental Section

#### General

New compounds were characterized by <sup>1</sup>H and <sup>13</sup>C NMR, <sup>13</sup>C off-resonance techniques, <sup>1</sup>H-<sup>1</sup>HCOSY, NOE, NOESY, HMQC, HMBC, IR, MS, HRMS, and elemental analysis. <sup>1</sup>H (400 MHz) and <sup>13</sup>C NMR (100 MHz) spectra were obtained with TMS as internal standard. IR spectra were recorded as thin films or as solids in KBr pellets. Column chromatography was performed on silica gel (MERK C60 or Fuji Silysia FL100DX). Bulb-to-bulb distillation (Kugelrohr) was accomplished at the oven temperature and

pressure indicated. Yields were determined by <sup>1</sup>H NMR using internal standards (1,1,2,2-tetrachloroethane or 1,1,1,2-tetrachloroethane). We applied the HF/DFT hybrid method originally proposed by Becke,9 referenced as B3LYP three parameter hybrid functional. All calculations were performed with Gaussian09 rev.C.01 and 6-31+G (d,p) for H, C, O, Br, DGDZVP for Ga were used for basis sets. All molecular geometries were fully optimized and energies were calculated including zero point energy correction by the normal mode analysis for each structure.

#### Materials

Dehydrated CH<sub>2</sub>Cl<sub>2</sub> and dehydrated ClCH<sub>2</sub>CH<sub>2</sub>Cl were purchased and used as obtained. Enol ethers **1a**, **1b**, **1d**, and **4k** are commercially available. All enol carboxylates are commercially available. Silyl ketene acetal **2n** is commercially available. Synthetic methods and spectral data of enol ethers **1c**, **1c**-*d*, **4a**, **4b**, **4f**, **4g**, **4h**, **4i**, **4j**, and silyl ketene actals **2b**, **2c**, **2d**, **2e**, **2f**, **2g**, **2h**, **2i**, **2j**, **2k**, **2l**, **2o**, and **2p** were reported in our previous communication, and these compounds were re-described. Silyl ketene acetal **2a**, **2m**, and enol ethers **4c**, **4d**, **4e** are new compounds, and synthetic methods and spectral data of these compounds are shown below. All metal salt catalysts are commercially available.

#### (1c) (vinyloxy)benzene<sup>29</sup>

#### Ph0

To a solution of (2-bromoethoxy)benzene (64 mmol, 12.9 g) in DMSO (50 mL) was added the solution of potassium *tert*-butoxide (54 mmol, 6.1 g) in DMSO (30 mL) at 15 °C. After the reaction mixture was stirred at room temperature for overnight, 1 M HCl (30 mL) was added dropwise at 0 °C. The mixture was extracted with diethyl ether (10 x 3 mL) and washed with sat. NaHCO<sub>3</sub> aq. The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by distillation under reduced pressure to give the product (4.6 g, 59%).

#### (1c-d) (E)-2-deuterio-1-phenoxyethylene<sup>30</sup>



To a solution of lithium aluminum hydride (44.5 mmol, 1.7 g) in 60 mL of diglyme, phenoxydeuterioacetylene (39 mmol, 4.6 g) was added dropwise at 0 °C. After the reaction mixture was stirred for 2 h at 0 °C, water was added dropwise. The mixture was extracted with diethyl ether (10 x 3 mL) and washed with sat. NaHCO<sub>3</sub> aq. The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by distillation under reduced pressure to give the product (1.58 g, 33%). bp: 107 °C/52 mmHg; IR: (neat) 1628 (OC=C) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.36 (dd, J = 8.8, 7.2 Hz, 2H, m), 7.11 (t, J = 7.2 Hz, 1H, p), 7.04 (d, J = 8.8 Hz, 2H, o), 6.68 (d, J = 13.8 Hz, 1H, 1-H), 4.79 (d, J = 13.8 Hz, 3H, 2-H); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 156.8 (s, i), 148.1 (d, C-1), 129.6 (d, m), 123.1 (d, p), 117.0 (d, o), 94.8 (t,  $J_{CD} = 24.7$  Hz, C-2); MS: (EI, 70 eV) m/z 121 (M<sup>+</sup>, 100), 92 (69); HRMS: (EI, 70 eV) Calculated (C<sub>8</sub>H<sub>7</sub>DO) 121.0638 (M<sup>+</sup>) Found: 121.0635.

#### (2a) 1-Butoxy-2-methyl-1-trimethylsiloxy-1-propene

To a solution of diisopropylamine (13.0 g, 129 mmol) in THF (100 mL) was added n-BuLi (80 mL, 1.6 M in hexane) at 0 °C. The resulted solution was stirred for 10 min at 0 °C. Then, butyl isobutyrate (17.5 g, 121 mmol) and trimethylsilylchloride (18 mL, 141 mmol) was slowly added to the LDA solution at - 78 °C. After the solution was stirred for 30 min at -78 °C, the reaction mixture was stirred for 2 h at room temperature. The reaction mixture was poured into cold sat. NaHCO<sub>3</sub> aq. and hexane, and the resulted mixture was extracted with hexane. The organic phase was washed with sat. NaHCO<sub>3</sub> aq., dried (MgSO<sub>4</sub>) and concentrated. The obtained crude oil was purified by distillation under the reduced pressure to give the desired product (24.6 g, 94%). bp: 42-47 °C/0.082 mmHg; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 3.68 (t, J = 6.4 Hz, 2H, 1'-H<sub>2</sub>), 1.61-1.53 (m, 2H, 2'-H<sub>2</sub>), 1.57 (s, 3H), 1.51 (s, 3H), 1.44-1.34 (m, 2H, 3'-H<sub>2</sub>), 0.92 (t, J = 7.2 Hz, 4'-H<sub>3</sub>), 0.19 (s, 9H, SiMe<sub>3</sub>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 148.3 (s, C-1), 91.4 (s, C-2), 68.7 (t, C-1'), 31.6 (t, C-2'), 19.3 (t, C-3'), 16.9 (q), 16.3 (q), 13.9 (q, C-4'), 0.03 (q, SiMe<sub>3</sub>); MS: (EI, 70 eV) *m/z* 216 (M+, 24), 144 (30), 129 (43), 75 (35), 73 (59), 70 (100); HRMS: (EI, 70 eV) Calculated (C<sub>11</sub>H<sub>24</sub>O<sub>2</sub>Si) 216.1546 (M<sup>+</sup>) Found: 216.1549.

#### (2b) 1-Methoxy-1-trimethylsiloxy-2-methyl-1-butene<sup>31</sup>

To a solution of diisopropylamine (13.2 g, 120 mmol) in THF (100 mL) was added n-BuLi (75 mL, 1.6 M in hexane) at 0 °C. The resulted solution was stirred for 10 min at room temperature. Then, a solution of methyl 2-methylbutanoate (11.3 g, 100 mmol) in THF (20 mL) was slowly added to the LDA solution at -78 °C. After the solution was stirred for 2 h at -78 °C, trimethylsilylchloride (13.0 g, 120 mmol) was added to the reaction mixture, and the reaction mixture was stirred for 20 h at room temperature. The

reaction mixture was poured into ice water and hexane, which was extracted with hexane. The organic phase was washed with brine, dried (MgSO<sub>4</sub>) and concentrated. The obtained crude oil was purified by distillation (bp. 72-76 °C, 25 mmHg) to give the desired product (8.6 g, 46%).

#### (2c) 2-Ethyl-1-ethoxy-1-trimethylsiloxy-1-butene<sup>32</sup>



To a solution of diisopropylamine (13.2 g, 120 mmol) in THF (100 mL) was added n-BuLi (75 mL, 1.6 M in hexane) at 0 °C. The resulted solution was stirred for 10 min at room temperature. Then, a solution of ethyl 2-ethylbutanoate (14.2 g, 100 mmol) in THF (20 mL) was slowly added to the LDA solution at -78 °C. After the solution was stirred for 2h at -78 °C, trimethylsilylchloride (13.0 g, 120 mmol) was added to the reaction mixture, and the reaction mixture was stirred for 20 h at room temperature. The reaction mixture was poured into ice water and hexane, which was extracted with hexane. The organic phase was washed with brine, dried (MgSO<sub>4</sub>) and concentrated. The obtained crude oil was purified by distillation (bp. 88-94 °C, 20 mmHg) to give the desired product (12.4 g, 62%).

#### (2d) {Cyclohexylidene(methoxy)methoxy}trimethylsilane<sup>33</sup>



To a solution of diisopropylamine (11.5 g, 110 mmol) in THF (80 mL) was added n-BuLi (63 mL, 1.6 M in hexane) at 0 °C. The resulted solution was stirred for 10 min at room temperature. Then, a solution of methyl cyclohexanecarboxyrate (11.4 g, 80 mmol) in THF (30 mL) was slowly added to the LDA solution at -78 °C. After the solution was stirred for 2h at -78 °C, trimethylsilylchloride (10.8 g, 100 mmol) was added to the reaction mixture, and the reaction mixture was stirred for 20 h at room temperature. The reaction mixture was poured into ice water and hexane, which was extracted with hexane. The organic phase was washed with brine, dried (MgSO<sub>4</sub>) and concentrated. The obtained crude oil was purified by distillation (bp. 100-102 °C, 20 mmHg) to give the desired product (14.3 g, 83%).

#### (2e) 1-Methoxy-1-trimethylsiloxy-2-phenylpropene<sup>33</sup>

OSiMe<sub>3</sub> OMe

To a solution of diisopropylamine (9.4 g, 93 mmol) in THF (50 mL) was added n-BuLi (59 mL, 1.6 M in hexane) at 0 °C. The resulted solution was stirred for 10 min at room temperature. Then, a solution of methyl 2-phenylpropanoate (12.7 g, 78 mmol) in THF (30 mL) was slowly added to the LDA solution at -78 °C. After the solution was stirred for 2 h at -78 °C, trimethylsilylchloride (10.8 g, 100 mmol) was added to the reaction mixture, and the reaction mixture was stirred for 20 h at room temperature. The reaction mixture was poured into ice water and hexane, which was extracted with hexane. The organic phase was washed with brine, dried (MgSO<sub>4</sub>) and concentrated. The obtained crude oil was purified by distillation (bp. 74 °C, 1.9 mmHg) to give the desired product (13.2 g, 72%).

#### (2f) (1-Methoxyhex-1-enyloxy)trimethylsilane<sup>40</sup>



To a solution of diisopropylamine (15.2 g, 150 mmol) in THF (95 mL) was added n-BuLi (86 mL, 1.6 M in hexane) at 0 °C. The resulted solution was stirred for 10 min at room temperature. Then, a solution of methyl hexanoate (14.0 g, 100 mmol) in THF (30 mL) was slowly added to the LDA solution at -78 °C. After the solution was stirred for 2 h at -78 °C, trimethylsilylchloride (16.1 g, 150 mmol) was added to the reaction mixture, and the reaction mixture was stirred for 20 h at room temperature. The reaction mixture was poured into ice water and hexane, which was extracted with hexane. The organic phase was washed with brine, dried (MgSO<sub>4</sub>) and concentrated. The obtained crude oil was purified by distillation (bp. 65-70 °C, 5 mmHg) to give the desired product (16.4 g, 75%).

#### (2g) (E)-((1-Ethoxy-3,3-dimethylbut-1-en-1-yl)oxy)trimethylsilane<sup>34</sup>



To a solution of diisopropylamine (9.0 g, 90 mmol) in THF (150 mL) was added n-BuLi (56 mL, 90 mmol, 1.6 M in hexane) at 0 °C. The resulted solution was stirred for 10 min at room temperature. Then, ethyl 3,3-dimethylbutanoate (10.8 g, 80 mmol) was slowly added to the LDA solution at -78 °C. After the solution was stirred for 1 h at -78 °C, trimethylsilylchloride (13.5 g, 120 mmol) was added to the reaction mixture, and the reaction mixture was stirred for 1 h at room temperature. The reaction mixture was poured into ice water and hexane, which was extracted with hexane. The organic phase was washed with brine, dried (MgSO<sub>4</sub>) and concentrated. The obtained crude oil was purified by distillation (1.1 mmHg) to give the desired product (13.4 g, 82%).

#### (2h) Phenylketene methyl trimethylsilyl acetal<sup>35</sup>



To a solution of diisopropylamine (85 mmol, 8.6 g) in THF (60 mL) was added *n*-BuLi (53 mL, 1.6 M in hexane) at 0 °C. The resulted solution was stirred for 10 min at room temperature. Then, a solution of methyl phenylacetate (85 mmol, 12.8 g) in THF (20 mL) was slowly added to the LDA solution at - 78 °C. After the solution was stirred for 2 h at -78 °C, chlorotrimethylsilane (213 mmol, 23.1 g) was added to the reaction mixture, and the reaction mixture was stirred for overnight at room temperature. The solvent was evaporated, and pentane (100 mL) was added to the residual mixture. Pentane was evaporated from the filtrate. The obtained crude oil was purified by distillation (bp. 70 °C, 0.24 mmHg) to give the desired product (15.7 g, 91%).

#### (2i) ((1-(Hexyloxy)vinyl)oxy)trimethylsilane<sup>36</sup>



To a solution of diisopropylamine (120 mmol, 12.5 g) in THF (60 mL) was added *n*-BuLi (70 mL, 1.6 M in hexane) at 0 °C. The resulted solution was stirred for 10 min at 0 °C. Then, a solution of hexyl acetate (100 mmol, 14.4 g), trimethylsilyl chloride (120 mmol, 12.3 g) in THF (20 mL) was slowly added to the LDA solution at -78 °C. After the solution was stirred for 20 min at -78 °C, the reaction mixture was stirred for 2 h at room temperature. The solvent was evaporated, and pentane (100 mL) was added to the residual mixture. Pentane was evaporated from the filtrate. The obtained crude oil was purified by distillation to give the desired product (8.9 g).

#### (2j) ((1-Ethoxy-2-(thiophen-3-yl)vinyl)oxy)trimethylsilane (E/Z = 50:50)



To a solution of diisopropylamine (70 mmol, 7.1 g) in THF (80 mL) was added *n*-BuLi (45 mL, 1.6 M in hexane) at 0 °C. The resulted solution was stirred for 10 min at room temperature. Then, ethyl 2-(thiophen-3-yl)acetate (60 mmol, 10.3 g) was slowly added to the LDA solution at -78 °C. After the solution was stirred for 2 h at -78 °C, chlorotrimethylsilane (80 mmol, 8.7 g) was added to the reaction mixture, and the reaction mixture was stirred for overnight at room temperature. The solvent was evaporated, and pentane (100 mL) was added to the residual mixture. Pentane was evaporated from the filtrate. The obtained crude oil was purified by distillation (bp. 70 °C, 0.2 mmHg) to give the desired

product (10.0 g, 69%) including a small amount of the starting ester and some by-product. b.p. 93 °C/1.6 mmHg; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 6.71-6.65 (m, 2H), 6.60-6.57 (m, 2H), 6.47-6.43 (m, 2H), 4.77 (s, 1H, 2-H), 4.68 (s, 1H, 2-H), 3.75 (q, J = 6.8 Hz, 2H, OCH<sub>2</sub>CH<sub>3</sub>), 3.57 (q, J = 6.8 Hz, 2H, OCH<sub>2</sub>CH<sub>3</sub>), 1.07 (t, J = 6.8 Hz, 3H, OCH<sub>2</sub>CH<sub>3</sub>), 1.03 (t, J = 6.8 Hz, OCH<sub>2</sub>CH<sub>3</sub>), 0.03 (s, 9H, SiMe<sub>3</sub>), 0.00 (s, 9H, SiMe<sub>3</sub>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 156.1 (s, C-1), 153.3 (s, C-1), 140.7 (s, C-3), 140. 0 (s, C-3), 126.11 (d), 126.06 (d), 121.1 (d), 121.0 (d), 120.5 (d), 120.3 (d), 82.4 (d, C-2), 75.3 (d, C-2), 63.7 (t, OCH<sub>2</sub>CH<sub>3</sub>), 62.9 (t, OCH<sub>2</sub>CH<sub>3</sub>), 14.9 (q, OCH<sub>2</sub>CH<sub>3</sub>), 14.3 (t, OCH<sub>2</sub>CH<sub>3</sub>), 0.5 (q, SiMe<sub>3</sub>), -0.3 (q, SiMe<sub>3</sub>); MS: (EI, 70 eV) *m/z* one isomer 242 (M<sup>+</sup>, 30), 124 (100), 73 (51), another isomer 242 (M<sup>+</sup>, 31), 124 (100), 73 (49); HRMS: (CI, 200 eV) Calculated (C<sub>11</sub>H<sub>18</sub>O<sub>2</sub>SSi) 242.0797 (M + 1) Found: one isomer 242.0805.

#### (2k) ((1-Methoxy-2-(thiophen-3-yl)vinyl)oxy)trimethylsilane<sup>37</sup>



To a solution of diisopropylamine (15 mmol, 1.6 g) in THF (20 mL) was added *n*-BuLi (10 mL, 1.6 M in hexane) at 0 °C. The resulted solution was stirred for 10 min at room temperature. Then, methyl 2-(1-methyl-1*H*-indol-3-yl)acetate (12 mmol, 2.5 g) was slowly added to the LDA solution at -78 °C. After the solution was stirred for 2 h at -78 °C, chlorotrimethylsilane (20 mmol, 2.2 g) was added to the reaction mixture, and the reaction mixture was stirred for overnight at room temperature. The solvent was evaporated, and pentane (100 mL) was added to the residual mixture. Pentane was evaporated from the filtrate. The obtained crude oil was purified by distillation (bp. 165-180 °C, 0.23 mmHg) to give the desired product (3.0 g, 86%) including a small amount of the starting ester and some by-product.

#### (2l) ((1-Methoxy-2-phenoxyvinyl)oxy)trimethylsilane<sup>38</sup>

OSiMe<sub>3</sub> PhO

To a solution of diisopropylamine (60 mmol, 6.1 g) in THF (80 mL) was added *n*-BuLi (38 mL, 1.6 M in hexane) at 0 °C. The resulted solution was stirred for 10 min at room temperature. Then, methyl 2-phenoxyacetate (50 mmol, 8.1 g) was slowly added to the LDA solution at -78 °C. After the solution was stirred for 2 h at -78 °C, chlorotrimethylsilane (75 mmol, 8.2 g) was added to the reaction mixture, and the reaction mixture was stirred for overnight at room temperature. The solvent was evaporated, and pentane (100 mL) was added to the residual mixture. Pentane was evaporated from the filtrate. The obtained crude oil was purified by distillation (bp. 70 °C, 0.2 mmHg) to give the desired product (5.7 g,

48%) including a small amount of the starting ester and some by-product.

#### (2m) ((1-ethoxy-2-(phenylthio)vinyl)oxy)trimethylsilane

To a solution of <sup>*i*</sup>Pr<sub>2</sub>NH (3.57 g, 35.3 mmol) in THF (60 mL) was added n-BuLi (1.6 M in hexane, 22.5 mL, 36 mmol) at 0 °C. After stirring at 0 °C for 5 min, ethyl 2-(phenylthio)acetate (5.80 g, 29.6 mmol) was dropwised to the reaction mixture at -78 °C. The mixture was stirred at -78 °C for 1 h and then Me<sub>3</sub>SiCl (5.16 g, 47.5 mmol) was dropwised to the mixture at -78 °C. The solution was stirred at -78 °C for 2 h and then warmed to room temperature. The solvent of the reaction mixture was evaporated and diluted with dry hexane (30 mL). The hexane layer was removed and concentrated by rotary evaporator. The residue was purified by distillation under reduced pressure to give the crude product (5.29g (73 wt%), 49%) including α-silyl ester and starting material, thioester. IR: (neat) 1612 (C=C) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.34-7.02 (m, 5H, Ar), 4.27 (s, 1H, 1-H), 3.95 (q, *J* = 7.3 Hz, 2H, 3-H<sub>2</sub>), 1.36 (t, *J* = 7.3 Hz, 3H, 4-H<sub>3</sub>), 0.206 (s, 9H, SiMe<sub>3</sub>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 162.9 (s, C-2), 140.4 (s, *i*), 128.4 (d), 125.3 (d), 124.2 (d, *p*), 64.4 (d, C-1), 64.0 (t, C-3), 14.4 (q, C-4), 0.279 (q, SiMe<sub>3</sub>); MS: (CI, 200 eV) *m/z* 269 (M + 1, 100), 223 (M - OEt, 59), 195 (M - SiMe<sub>3</sub>, 41); HRMS: (CI, 200 eV) Calculated (C<sub>13</sub>H<sub>21</sub>O<sub>2</sub>SSi) 269.1032 (M + 1) Found: 269.1026.

#### (20) Trimethyl((2-(methyl-d<sub>3</sub>)-1-phenethoxyprop-1-en-1-yl-3,3,3-d<sub>3</sub>)oxy)silane

$$\begin{array}{c} OSiMe_{3}\\ D_{3}C_{1} \\ \hline \\ CD_{3} \\ \end{array} Ph$$

To a solution of ethyl cyanoacetate (11.3 g, 100 mmol) and  $K_2CO_3$  (55 g, 400 mmol) in anhydrous acetone (150 mL) was added CD<sub>3</sub>I (34.2 g, 240 mmol). After the reaction mixture was refluxed for 4 h, the reaction was quenched by water. The aqueous layer was extracted with ether, and the combined organic layer was washed with brine, dried (MgSO<sub>4</sub>) and concentrated. This crude product, ethyl 2-cyano-2-(methyl- $d_3$ )propanoate-3,3,3- $d_3$  was not purified further and was carried on directly to the next step.

To a solution of KOH (6.8 g, 120 mmol) in EtOH (250 mL) and  $H_2O$  (25 mL) was added ethyl 2-cyano-2-(methyl- $d_3$ )propanoate-3,3,3- $d_3$ . After the reaction mixture was refluxed for 3 h, the reaction was quenched by water. The aqueous layer was acidified by 1 N HCl aq., and was extracted with ether. the combined organic layer was washed with brine, dried (MgSO<sub>4</sub>) and concentrated to yield 9.0 g, 80 mmol of the crude product. This crude product was not purified further and was carried on directly to the next step.

To 12 N HCl aq. (90 mL) was added 2-cyano-2-(methyl- $d_3$ )propanoic-3,3,3- $d_3$  acid (4.7 g), and then the reaction mixture was heated 120 °C for 2 h. After the mixture was cooled at 0 °C, conc. H<sub>2</sub>SO<sub>4</sub> (45 mL) was added slowly. The reaction heated at 160 °C for 20 h. The reaction was poured into cold water (0 °C), and the aqueous layer was extracted with ether. The combined organic layer was dried over Na<sub>2</sub>SO<sub>4</sub> and concentrated to give the crude product, 2-(methyl- $d_3$ )propanoic-3,3,3- $d_3$  acid. The same reaction using 2.3 g of 2-cyano-2-(methyl- $d_3$ )propanoic-3,3,3- $d_3$  acid was carried out. The combined crude product (5.6 g) was not purified further and was carried on directly to the next step.

To a solution of 2-(methyl- $d_3$ )propanoic-3,3,3- $d_3$  acid (5.6 g) and K<sub>2</sub>CO<sub>3</sub> (15.3 g, 110 mmol) in DMF (150 mL) was added 2-bromoethylbenzene (16.6 g, 90 mmol). The reaction mixture was stirred at room temperature for 38 h. The mixture was poured into water, and the aqueous layer was extracted with ether. The combined organic layer was washed with brine, dried over MgSO4, concentrated, and purified by column chromatography to give the product, phenethyl 2-(methyl- $d_3$ )propanoate-3,3,3- $d_3$  (8.1 g, 41 mmol). Total yield from ethyl cyanoacetate was 41%.

<sup>1</sup>H NMR (400 Mz, CDCl<sub>3</sub>) 7.33-7.21 (m, 5H), 4.28 (t, J = 6.5 Hz, 2H), 2.94 (t, J = 6.5 Hz, 2H), 2.49 (s, 1H), <sup>13</sup>C NMR (100 Mz, CDCl<sub>3</sub>) 176.9, 137.8, 128.8, 128.3, 126.4, 64.6, 35.0, 33.4, 17.9, HRMS: (EI, 70 eV) Calculated (C<sub>12</sub>H<sub>11</sub>D<sub>6</sub>O<sub>2</sub>) 199.1605 (M<sup>+</sup>) Found: 199.1605.

To a solution of <sup>1</sup>Pr<sub>2</sub>NH (1.35 g, 13.3 mmol) in THF (10 mL) was added n-BuLi (1.6 M in hexane, 7.5 mL, 12.0 mmol) at 0 °C. After stirring at 0 °C for 5 min, phenethyl 2-(methyl-*d<sub>3</sub>*)propanoate-3,3,3-*d<sub>3</sub>* (2.07 g, 10.4 mmol) was dropwised to the reaction mixture at -78 °C. The mixture was stirred at -78 °C for 55 min and then Me<sub>3</sub>SiCl (1.18 g, 10.9 mmol) was dropwised to the mixture at -78 °C. The solution was stirred at -78 °C for 35 min and then warmed to rt. The solvent of the reaction mixture was evaporated and diluted with diethyl ether (20 mL). This solution was washed by saturated NaHCO<sub>3</sub> aq. (30 mL x 3) and dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by distillation under reduced pressure to give the product (1.69 g, 60%). bp: 120 °C /0.15 mmHg; IR: (neat) 1697 (C=C) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.26-7.13 (m, 5H, Ar), 3.89 (t, *J* = 7.2, 2H, 3-H<sub>2</sub>), 2.88 (t, *J* = 7.2 Hz, 2H, 4-H<sub>2</sub>), 0.15 (s, 9H, SiMe<sub>3</sub>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 148.0 (s, C-2), 138.4 (s, *i*), 128.9 (d), 128.3 (d), 126.1 (d, *p*), 91.3 (s, C-1), 69.3 (t, C-3), 35.9 (t, C-4), 16.0 (septet, J<sup>1</sup><sub>C-D</sub> = 18.0 Hz, 1-CD<sub>3</sub>), 15.4 (septet, J<sup>1</sup><sub>C-D</sub> = 19.7 Hz, 1-CD<sub>3</sub>), -0.05 (q, SiMe<sub>3</sub>) MS: (CI, 200 eV) *m/z* 271 (M + 1, 60), 105 (100); HRMS: (CI, 200 eV) Calculated (C<sub>15</sub>H<sub>19</sub>D<sub>6</sub>O<sub>2</sub>Si) 271.2000 (M + 1) Found: 271.1998; Analysis: C<sub>15</sub>D<sub>6</sub>H<sub>18</sub>O<sub>2</sub>Si (270.19) Calculated: C, 66.61; D, 4.47; H, 6.71, Found: C, 66.69; D, 4.62; H, 6.94.
#### (2p) Trimethyl{(2-methyl-1-phenoxyprop-1-en-1-yl)oxy}silane<sup>39</sup>



To a solution of diisopropylamine (133 mmol, 13.4 g) in THF (140 mL) was added *n*-BuLi (122 mmol, 76 mL, 1.6 M in hexane) at 0 °C. The resulted solution was stirred for 10 min at room temperature. Then, a solution of isobutyric acid phenyl ester (111 mmol, 18.3 g), chlorotrimethylsilane (140 mmol, 15.9 g) and THF (50 mL) was slowly added to the LDA solution at -78 °C. The reaction mixture was stirred for 1 h at -78 °C, and for further 8 h at room temperature. The solvent was evaporated, and hexane (100 mL) was added to the residual mixture. Hexane was evaporated from the filtrate. The obtained crude oil was purified by distillation (bp. 70 °C, 0.2 mmHg) to give the desired product (25.5 g, 97%).

#### (4a) 2-Methoxyoct-1-ene<sup>40</sup>



To a solution of 2,2-dimethoxyoctane (6.7 mmol, 1.17 g) in  $CH_2Cl_2$  (40 mL) was added triisobutylaluminum (17 mL, 1 M in toluene) at room temperature. The reaction mixture was stirred at room temperature for 6 h. The reaction was quenched with an excess of 20% aq. trisodium citrate (3 h stirring). The mixture was extracted with diethyl ether (10 x 3 mL) and washed with sat. NaHCO<sub>3</sub> aq.. The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography to give the product (0.80 g, 84%).

# (4b) (3-Methoxybut-3-en-1-yl)benzene<sup>41</sup>



According to the synthetic method for 1d, this compound was synthesized from (3,3-dimethoxybutyl)benzene (30 mmol). The yield was 77%.

#### (4c) (Z)-2-methoxynona-1,7-diene



To a 100 mL three necked flask filled with THF (20 mL) was added n-BuLi (1.6 M in hexane, 32.5 mL, 50.0 mmol) at 0 °C. The mixture was cooled at -90 °C and then 2-methoxypropene (14.6 g, 202 mmol)

was dropwised over 5 min. The solution of 'BuOK (5.69 g, 50.7 mmol) in THF (35 mL) was dropwised into the reaction mixture over 10 min and the mixture was stirred at -75 °C for 30 min. The mixture was stirred at -50 °C for 30 min and then (*Z*)-6-bromohex-2-ene (8.49 g, 52.0 mmol) was dropwised over 5 min. After stirring at -50 °C for 10 min, the reaction mixture was warmed to 0 °C and quenched by water (75 mL). The solution was extracted by diethyl ether (50 mL x 3) and the collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and concentrated under 100 Torr, 50 °C to give the product (6.62 g, 86%).

IR: (neat) 1655 (C=C) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 5.48-5.34 (m, 2H, 8-H and 9-H), 3.85 (s, 2H, 1-H<sub>2</sub>), 3.52 (s, 3H, 3-H<sub>3</sub>), 2.10 (t, J = 7.3 Hz, 2H, 4-H<sub>2</sub>), 2.04 (dt, J = 7.2, 7.3 Hz, 2H, 7-H<sub>2</sub>), 1.60 (d, J = 6.3 Hz, 3H, 10-H<sub>3</sub>), 1.54-1.46 (m, 2H, 5-H<sub>2</sub>), 1.40-1.30 (m, 2H, 6-H<sub>2</sub>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 164.3 (s, C-2), 130.6 (d), 123.8 (d), 80.1 (t, C-1), 54.7 (q, C-3), 34.8 (t, C-4), 29.0 (t, C-6), 27.0 (C-5, t), 26.6 (C-7, t), 12.7 (q, C-10); MS: (CI, 200 eV) m/z 155 (M + 1, 100); HRMS: (CI, 200 eV) Calculated (C<sub>10</sub>H<sub>19</sub>O) 155.1436 (M + 1) Found: 155.1438

#### (4d) 9-chloro-2-methoxynon-1-ene



To a 100 mL three necked flask filled with THF (20 mL) was added n-BuLi (1.6 M in hexane, 32.5 mL, 50.0 mmol) at 0 °C. The mixture was cooled at -90 °C and then 2-methoxypropene (15.1 g, 209 mmol) was dropwised over 5 min. The solution of 'BuOK (5.65 g, 50.3 mmol) in THF (35 mL) was dropwised into the reaction mixture over 10 min and the mixture was stirred at -75 °C for 30 min. The mixture was stirred at -50 °C for 30 min and then 1-bromo-6-chlorohexane (10.4 g, 52.2 mmol) was dropwised over 5 min. After stirring at -50 °C for 25 min, the reaction mixture was warmed into 0 °C and the solvent of the mixture was evaporated. The residue was washed by dry hexane (50 mL x 2) and the hexane layer was concentrated. The residue was purified by distillation under reduced pressure to give the product (2.42 g, purify: 80 wt%, 25% yield).

bp: 45 °C /0.09 mmHg; IR: (neat) 1655 (C=C) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>)

3.84 (s, 2H, 1-H<sub>2</sub>), 3.56-3.50 (2H, 10-H<sub>2</sub>), 3.52 (s, 3H, 1-H<sub>3</sub>), 2.08 (t, J = 7.25 Hz, 2H, 4-H<sub>2</sub>), 1.83-1.72 (m, 2H, 9-H<sub>2</sub>), 1.54-1.38 (m, 4H, 5-H<sub>2</sub> and 8-H<sub>2</sub>), 1.38-1.24 (m, 4H, 6-H<sub>2</sub> and 7-H<sub>2</sub>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 164.3 (s, C-2), 80.0 (t, C-1), 54.6 (q, C-3), 45.1 (t, C-10), 34.8 (t, C-4), 32.6 (t, C-9), 28.9 (t), 28.6 (t), 27.2 (t), 26.7 (t); MS: (EI, 90 eV) m/z 192 (M<sup>+</sup> + 2, 6) 190 (M<sup>+</sup>, 17), 155 (M<sup>+</sup> - Cl, 19), 85 (100), 72 (27), 55 (63); HRMS: (EI, 90 eV) Calculated (C<sub>10</sub>H<sub>19</sub>ClO) 190.1124 (M<sup>+</sup>) Found: 190.1123

#### (4e) 2,11-dimethoxyundec-1-ene



To a 100 mL three necked flask filled with THF (12 mL) was added n-BuLi (1.6 M in hexane, 19 mL, 30 mmol) at 0 °C. The mixture was cooled at -90 °C and then 2-methoxypropene (6.06 g, 84 mmol) was dropwised over 5 min. The solution of 'BuOK (3.37 g, 30.0 mmol) in THF (20 mL) was dropwised into the reaction mixture over 10 min and the mixture was stirred at -75 °C for 30 min. The mixture was stirred at -50 °C for 30 min and then 1-bromo-8-methoxyoctane (4.44 g, 19.9 mmol) was dropwised over 5 min. After stirring at -50 °C for 25 min, the reaction mixture was warmed into 0 °C and the solvent of the mixture was evaporated. The residue was washed by dry hexane (20 mL x 2) and the hexane layer was concentrated. The residue was purified by distillation under reduced pressure to give the product (1.21 g, 61%). bp: 60 °C /0.14 mmHg; IR: (neat) 1655 (C=C) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 3.84 (s, 2H, 1-H<sub>2</sub>), 3.52 (s, 3H, 3-H<sub>3</sub>), 3.36 (t, *J* = 6.8 Hz, 2H, 12-H<sub>2</sub>), 3.33 (s, 3H, 13-H<sub>3</sub>), 2.07 (t, *J* = 7.7 Hz, 2H, 4-H<sub>2</sub>), 1.60-1.53 (m, 2H, 11-H<sub>2</sub>), 1.51-1.43 (m, 2H, 5-H<sub>2</sub>), 1.37-1.20 (m, 10H, 6-H<sub>2</sub>, 7-H<sub>2</sub>, 8-H<sub>2</sub>, 9-H<sub>2</sub> and 10-H<sub>2</sub>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 164.4 (s, C-2), 79.9 (t, C-1), 72.9 (t, C-12), 58.5 (q, C-13), 54.6 (q, C-3), 34.9 (t, C-4), 29.6 (t, C-11), 29.5 (t), 29.4 (t), 29.3 (t), 29.1 (t), 27.3 (t, C-5), 26.1 (t); MS: (EI, 90 eV) *m*/*z* 214 (M<sup>+</sup>, 9), 85 (97), 72 (100); HRMS: (EI, 90 eV) Calculated (C<sub>13</sub>H<sub>26</sub>O<sub>2</sub>) 214.1933 (M<sup>+</sup>) Found: 214.1933.

#### (4f) 3-Methoxy-1-phenyl-3-buten-1-yne



According to the synthetic method for **1f**, benzoic acid (2.5 mmol, 0.31 g), followed by trimethylsilyl chloride (75 mmol, 9.5 mL) was added to 3,3-dimethoxy-1-phenyl-1-butyne (25 mmol, 4.7 g) in pyridine (25 mL). The mixture had been stirred at 70 °C for 3 h, and then, was cooled to 0 °C. Aqueous NaOH (15 %, 60 mL) was added slowly, and the aqueous layer was extracted with ether. Then, the wash with an equal volume of dilute aqueous NaOH removed pyridine. The organic layer was washed with brine and dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 100:0, column length 11 cm) and distillation under reduced pressure. (2.7 g, 68%). bp: 120 °C/1.2 mmHg; IR: (neat) 1604 (OC=C) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.52-7.47 (m, 2H, *o*), 7.36-7.30 (m, 3H, *m* and *p*), 4.68 (d, *J* = 2.4 Hz, 1H, 4-*H*H), 4.58 (d, *J* =

2.4 Hz, 1H, 4-H*H*), 3.65 (s, 3H, OMe); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 145.2 (s, C-3), 131.7 (d, *o*), 128.7 (d, *p*), 128.3 (d, *m*), 122.0 (s, *i*), 93.0 (t, C-4), 87.4 (s, C-1), 84.6 (s, C-2), 55.4 (q, OMe); MS: (EI, 70 eV) *m*/*z* 158 (M<sup>+</sup>, 83), 128 (M<sup>+</sup> - OMe, 100), 127 (30), 115 (35); HRMS: (EI, 70 eV) Calculated (C<sub>11</sub>H<sub>10</sub>O) 158.0732 (M<sup>+</sup>) Found: 158.0734.

# (4g) (1-Methoxyvinyl)benzene<sup>42</sup>



Benzoic acid (3.0 mmol, 0.37 g), followed by trimethylsilyl chloride (90 mmol, 9.7 g) was added to the dimethyl acetal of acetophenone (30 mmol, 4.8 g) in pyridine (30 mL). The mixture had been stirred at 70 °C for 3 h, and then, was cooled to 0 °C. Aqueous NaOH (15%, 60 mL) was added slowly, and the aqueous layer was extracted with ether. Then, the wash with an equal volume of dilute aqueous NaOH removed pyridine. The organic layer was washed with brine and dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 100:0, column length 11 cm) (2.5 g, 64%).

# (4h) 1-(1-Methoxyvinyl)-4-methylbenzene<sup>43</sup>



According to the synthetic method for **1f**, benzoic acid (5 mmol, 0.61 g), followed by trimethylsilyl chloride (150 mmol, 15.9 g) was added to the dimethyl acetal of *p*-methyacetophenone (50 mmol, 8.1 g) in pyridine (25 mL). The mixture had been stirred at 70 °C for 3 h, and then, was cooled to 0 °C. Aqueous NaOH (15%, 60 mL) was added slowly, and the aqueous layer was extracted with ether. Then, the wash with an equal volume of dilute aqueous NaOH removed pyridine. The organic layer was washed with brine and dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 100:0, column length 11 cm) to give the pure product (3.0 g, 41%).

## (4i) 1-Chloro-4-(1-methoxyvinyl)benzene<sup>43</sup>

OMe

According to the synthetic method for **1f**, benzoic acid (3.5 mmol, 0.45 g), followed by trimethylsilyl chloride (105 mmol, 11.3 g) was added to the dimethyl acetal of *p*-chloroacetophenone (35 mmol, 7.0 g) in pyridine (30 mL). The mixture had been stirred at 70 °C for 3 h, and then, was cooled to 0 °C. Aqueous NaOH (15%, 60 mL) was added slowly, and the aqueous layer was extracted with ether. Then, the wash with an equal volume of dilute aqueous NaOH removed pyridine. The organic layer was washed with brine and dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 100:0, column length 11 cm) (6.7 g, 71%).

#### (4j) 2-(1-Methoxyethenyl)naphthalene



According to the synthetic method for **1f**, benzoic acid (5.4 mmol, 0.66 g), followed by TMSCI (150 mmol, 16.2 g) was added to the dimethyl acetal of 2-acetonaphthalene (44 mmol, 9.5 g) in pyridine (25 mL). The mixture had been stirred at 70 °C for 3 h, and then, was cooled to 0 °C. Aqueous NaOH (15%, 60 mL) was added slowly, and the aqueous layer was extracted with ether. Then, the wash with an equal volume of dilute aqueous NaOH removed pyridine. The organic layer was washed with brine and dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 100:0, column length 11 cm) (4.9 g, 60%). IR: (neat) 1643 (OC=C) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 8.20 (d, J = 1.2 Hz, 1H, 1'-H), 7.94-7.81 (m, 2H, 5'-H and 8'-H), 7.84 (d, J = 8.8 Hz, 1H, 4'-H), 7.76 (dd, J = 8.8, 1.2 Hz, 1H, 3'-H), 7.55-7.47 (m, 2H, 6'-H and 7'-H), 4.86 (d, J = 2.8 Hz, 1H, 2-*H*H), 4.37 (d, J = 2.8 Hz, 1H, 2-H*H*), 3.85 (s, 3H, OMe); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 160.6 (s, C-1), 133.5 (s), 133.3 (s), 133.1 (s), 128.5 (d, C-8'), 127.6 (d), 127.5 (d), 126.1 (d), 124.3 (d, C-1'), 123.4 (d, C-3'), 82.4 (t, C-2), 55.3 (q, OMe); MS: (EI, 70 eV) *m/z* 184 (M<sup>+</sup>, 100), 183 (71), 154 (33), 153 (51), 152 (28), 141 (63); HRMS: (EI, 70 eV) Calculated (C<sub>13</sub>H<sub>12</sub>O) 184.0888 (M<sup>+</sup>) Found: 184.0885.

#### Typical Procedure for coupling reaction of enol ether with ketene silyl acetal (Table 1, entry 2)

To a solution of GaBr<sub>3</sub> (0.05 M in dichloroethane, 6 mL) and ethylmethylketene methyl trimethylsilyl acetal (4.5 mmol, 0.849 g) was added butyl vinyl ether (3.0 mmol, 0.300 g). The mixture was stirred at 80 °C for 2 h and then was quenched by 1 M HCl aq. The mixture was extracted with diethyl ether (10 x 3 mL). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 98:2, column length 11 cm) (0.391 g,

92%).

# Experimental Procedure in the observation of alkylgallium intermediate 7 (Figure 1)

To a solution of dimethylketene trimethylsilyl acetal 2n (1.5 mmol), GaBr<sub>3</sub> (0.5 mmol), and CD<sub>2</sub>Cl<sub>2</sub> (1.5 mmol) was added a solution of ethyl vinyl ether 1d (1 mmol) and CD<sub>2</sub>Cl<sub>2</sub> (1 mmol) at -30 °C. After the reaction mixture was stirred for a few minute, a portion of the reaction mixture was added to NMR tube. And then, the coupling reaction was monitored by 1H NMR at room temperature.

#### Experimental Procedure in the observation of alkylgallium intermediate 8 (Scheme 4)

A solution of GaBr<sub>3</sub> (0.3 mmol), deuterated ketene silyl acetal **20** (2 mmol) and CD<sub>2</sub>Cl<sub>2</sub> (1 mL) was prepared (solution A). Solution A (0.4 mL) was added to NMR tube, and then the tube was cooled to - 78 °C. A solution of vinyl ether **1c** (1 mmol) and CD<sub>2</sub>Cl<sub>2</sub> (1 mL) was prepared (solution B). To cooled NMR tube of solution A was added 0.4 mL of solution B. This sample was monitored by 1H NMR at - 30 °C.



# Experimental Procedure in the initial rate kinetics of the coupling reaction of butyl vinyl ether 1a with ketene silyl acetal 2a (Figure 2)

To a solution of GaBr<sub>3</sub> (x mmol) and ClCH<sub>2</sub>CH<sub>2</sub>Cl (1 mL) was added 1 M ClCH<sub>2</sub>CH<sub>2</sub>Cl solution of silyl ketene acetal **2a** (y mmol, y mL) and dodecane as an internal standard, and then, the resulting solution was stirred at 30 °C. 1 M ClCH<sub>2</sub>CH<sub>2</sub>Cl solution of butyl vinyl ether **1a** (z mmol, z mL) was added to the resulting solution. After stirring for indicated period, small amount of reaction mixture was sampled via syringe and poured into ethyl acetate to quench the reaction. The yield was observed by GC.

#### The effect of injection of Me<sub>3</sub>SiBr on the coupling reaction

To a solution of GaBr<sub>3</sub> (0.1 mmol) and ClCH<sub>2</sub>CH<sub>2</sub>Cl (4 mL) was added silyl ketene acetal **2a** (1.5 mmol) and dodecane as an internal standard, and then, the resulting solution was stirred at 50 °C. Me<sub>3</sub>SiBr followed by butyl vinyl ether 1a (1 mmol) was added to the resulting solution. After stirring for indicated period, small amount of reaction mixture was sampled via syringe and poured into ethyl acetate to quench the reaction. The yield was observed by GC.

[TMSBr]/ M	[TMSBr]/ mol%	initial rate
0	0	0.000188
0.04083	15	0.000206
0.0822	30	0.000130



#### Derivation of rate equation of reaction between vinyl pivalate 6c and silyl ketene acetal 2

Calculations revealed the details of the elimination process (Eq. 2). We derived the rate equation from the results of kinetic studies and calculations. The facts that the carboindation is fast (Eq. 1) and that the elimination process is the turnover-limiting step (Eq. 2), established the inequality,  $k_2$ ,  $k_3 < k_1$  (Eq. 3). Therefore, the concentration of [GaBr<sub>3</sub>] is nearly zero under catalytic conditions, [GaBr<sub>3</sub>] << [1], [2] (Eq. 4), and the total amount of gallium species is expressed by equation 5. Assuming a steady-state situation for 7 (Eq. 2 and 6) allows its concentration to be expressed according to equation 7. Substitution of equation 5 into equation 7 furnishes equation 8 relating [7] to [GaBr<sub>3</sub>]<sub>0</sub>. Therefore, the reaction rate depends on only the concentration of loaded GaBr<sub>3</sub>, [GaBr<sub>3</sub>]<sub>0</sub> (Eq. 9). Considering the relationship,  $k_2$ ,  $k_3 << k_{-2}$ , led to by calculations for the elimination process (Eq. 10), the rate equation is represented by equation 11. The first order rate-dependence on the concentrations of loaded GaBr<sub>3</sub> is consistent with kinetic studies.

$$GaBr_{3} + \bigcap_{\mathbf{1}} OR + R^{1} + \bigcap_{\mathbf{R}^{2}} OR^{3} + \prod_{\mathbf{M}^{2} \neq \mathbf{0}R^{3}} \prod_{\mathbf{M}^{2} = \mathbf{3}} Br_{2}Ga^{,,0} + \bigcap_{\mathbf{R}^{1}} R^{2} + \prod_{\mathbf{R}^{2} \neq \mathbf{1}} OR + \prod_{\mathbf{M}^{2} = \mathbf{1}} P^{2}Ga^{,0} + \prod_{\mathbf{R}^{2} = \mathbf{1}} P^{2}G$$

rate = 
$$\frac{d[\mathbf{3}]}{dt} = k_3[\mathbf{10}] = \frac{k_2 k_3}{k_2 + k_2 + k_3} [GaBr_3]$$
 (9)

$$k_2, k_3 << k_{-2}$$
 (10)

$$rate = \frac{k_2 k_3}{k_{-2}} [\text{GaBr}_3] = k_{\text{obs}} [\text{GaBr}_3]$$
(11)

#### Derivation of rate equation of reaction between vinyl pivalate 6c and silyl ketene acetal 2

We considered that silyl ketene acetal **2** has the inhibition effect via the coordination to GaBr<sub>3</sub> (Eq. 14) and carbogallation among GaBr<sub>3</sub>, **6c**, and **2a** is the turnover limiting step (Eq. 15). Assuming the equilibrium among GaBr<sub>3</sub>, **2a**, and **6c** (Eq. 17) leads to the rate equation, Eq. 18, involving the concentrations of **6c** and **20**. The relation among amount of loaded GaBr<sub>3</sub>, **20**, and **21** is expressed by Eq. 19. Using Eqs 17 and 19, equation 20 for the concentration of **20** is led. Substitution of equation 20 into rate low (equation 18) furnishes equation 21. The equilibrium among GaBr<sub>3</sub>, **2a**, and **6c** (Eq. 14) lies to the left ( $K_1 < 1$ ), which is supported by theoretical calculation (Figure A), so that the approximation 22 is possible. Finally, the rate low depending on [GaBr<sub>3</sub>]<sub>0</sub> and [**6c**] is obtained.



Arrhenius plot using  $k_{obs}$ s of the coupling reaction of vinyl ether 1a with ketene silyl acetal 2a catalyzed GaBr<sub>3</sub>





apparent  $E_{act} = 7324.2 \text{ x } 8.3145/4.183/1000 = 14.6 \text{ kcal/mol}$ 

#### **Product Data**

Spectral data of the products **3ab**, **3ac**, **3ad**, **3ae**, **3af**, **3ah**, **3ai**, **5an**, **5gn**, **5hn**, and **5in** were in an excellent agreement with reported data. Spectral data for the products **3aa**, (*E*)-**3aa**-*d*, **3aj**, **3ak**, **5fn**, and **5jn** were described in our previous paper, and were reshown below. Spectral data for the products, **3am**, **5cn**, **5dn**, **5en**, **5kn**, and Phenethyl 2,2-bis(methyl-d<sub>3</sub>)but-3-enoate which are new compounds, were also shown below.

# (3aa) Butyl 2,2-dimethyl-3-butenoate



To a solution of GaBr<sub>3</sub> (0.05 M in dichloroethane, 6 mL) and dimethylketene butyl trimethylsilyl acetal (4.5 mmol, 0.976 g) was added butyl vinyl ether (3.0 mmol, 0.304 g). The mixture was stirred at 80 °C for 2 h and then was quenched by 1 M HClaq.. The mixture was extracted with diethyl ether (10 x 3 mL). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 98:2, column length 11 cm) (0.457 g, 93%). IR: (neat) 1732 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 6.00 (dd, J = 17.6, 10.8 Hz, 1H, 3-H), 5.06 (dd, J = 17.6, 1.2 Hz, 1H, 4-H<sup>B</sup>), 5.03 (dd, J = 10.8, 1.2 Hz, 1H, 4-H<sup>A</sup>), 4.04 (t, J = 6.6 Hz, 2H, 1'-H<sub>2</sub>), 1.63-1.54 (m, 2H, 2'-H<sub>2</sub>), 1.41-1.31 (m, 2H, 3'-H<sub>2</sub>), 1.27 (s, 6H, 2-Me<sub>2</sub>), 0.90 (t, J = 7.4 Hz, 3H, 4'-H<sub>3</sub>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 176.3 (s, C-1), 142.6 (d, C-3), 112.6 (t, C-4), 64.5 (t, C-1'), 44.8 (s, C-2), 30.6 (t, C-2'), 24.5 (q, 2-Me<sub>2</sub>), 19.7 (t, C-3'), 13.6 (q, C-4'); MS: (EI, 70 eV) *m/z* 170 (M<sup>+</sup>, 0.72), 69 (M<sup>+</sup> - CO<sub>2</sub>C<sub>4</sub>H<sub>9</sub>, 100), 57 (48), 41 (35); HRMS: (EI, 70 eV) Calculated (C<sub>10</sub>H<sub>18</sub>O<sub>2</sub>) 170.1307 (M<sup>+</sup>) Found: 170.1299; Analysis: C<sub>10</sub>H<sub>18</sub>O<sub>2</sub> (170.25) Calcd: C, 70.55; H, 10.66 Found: C, 70.67; H, 10.65.

#### {(E)-3aa-d} Butyl (E)-4-deuterio-2,2-dimethyl-3-butenoate



To a solution of GaBr<sub>3</sub> (0.05 M in dichloroethane, 2 mL) and dimethylketene butyl trimethylsilyl acetal (1.5 mmol, 0.325 g) was added *trans*-1-deuterio-2-phenoxy ethene (1.0 mmol, 0.116 g). The mixture was stirred at 80 °C for 2 h and then was quenched by 1 M HClaq.. The mixture was extracted with diethyl ether (10 x 3 mL). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 98:2, column length 11 cm) to give the product (0.127 g, 77%). IR: (neat) 1732 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 6.03 (dt,  $J_{\text{HH}} = 17.6$  Hz,  $J_{\text{HD}} = 1.4$  Hz, 1H, 3-H), 5.07 (d, J = 17.6 Hz, 1H, 4-H), 4.07 (t, J = 6.6 Hz, 2H, 1'-H<sub>2</sub>), 1.65-1.56 (m, 2H, 2'-H<sub>2</sub>), 1.43-1.34 (m, 2H, 3'-H<sub>2</sub>), 1.30 (s, 6H, 2-Me<sub>2</sub>), 0.93 (t, J = 7.4 Hz, 3H, 4'-H<sub>3</sub>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 176.4 (C-1), 142.6 (C-3), 112.4 ( $J_{\text{CD}} = 24.7$  Hz, C-4), 64.6 (C-1'), 44.9 (C-2), 30.6 (C-2'), 24.6 (2-Me<sub>2</sub>), 19.1 (C-3'), 13.7 (C-4'); MS: (EI, 70 eV) *m/z* 171 (M<sup>+</sup>, 1.62), 70 (M<sup>+</sup> - CO<sub>2</sub>C<sub>4</sub>H<sub>9</sub>, 100), 57 (54), 41 (32); HRMS: (EI, 70 eV) Calculated (C<sub>10</sub>H<sub>17</sub>DO<sub>2</sub>) 171.1370 (M<sup>+</sup>) Found: 171.1368.

# (3ab) Methyl 2-ethyl-2-methyl-3-butenoate<sup>44</sup>



To a solution of GaBr<sub>3</sub> (0.05 M in dichloroethane, 6 mL) and ethylmethylketene methyl trimethylsilyl acetal (4.5 mmol, 0.849 g) was added butyl vinyl ether (3.0 mmol, 0.300 g). The mixture was stirred at 80 °C for 2 h and then was quenched by 1 M HClaq.. The mixture was extracted with diethyl ether (10 x 3 mL). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 98:2, column length 11 cm) (0.391 g, 92%). This compound is known, and NMR spectra are shown below.

# (3ac) Ethyl 2,2-diethyl -3-butenoate<sup>45</sup>



To a solution of GaBr<sub>3</sub> (0.05 M in dichloroethane, 6 mL) and diethylketene ethyl trimethylsilyl acetal (4.5 mmol, 0.957 g) was added butyl vinyl ether (3.0 mmol, 0.307 g). The mixture was stirred at 80 °C

for 2 h and then was quenched by 1 M HClaq.. The mixture was extracted with diethyl ether (10 x 3 mL). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 98:2, column length 11 cm) (0.391 g, 92%). This compound is known, and NMR spectra are shown below.

# (3ad) Methyl 1-vinylcyclohexanecarboxylate<sup>46</sup>



To a solution of GaBr<sub>3</sub> (0.05 M in dichloroethane, 6 mL) and ketene methyl trimethylsilyl acetal (4.5 mmol, 0.964 g) was added butyl vinyl ether (3.0 mmol, 0.305 g). The mixture was stirred at 80 °C for 2 h and then was quenched by 1 M HClaq.. The mixture was extracted with diethyl ether (10 x 3 mL). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography to give the product (0.461 g, 90%). This compound is known, and NMR spectra are shown below.

# (3ae) methyl 2-methyl-2-phenylbut-3-enoate<sup>47</sup>



To a solution of GaBr<sub>3</sub> (0.05 M in dichloroethane, 2 mL) and ketene methyl trimethylsilyl acetal (1.5 mmol, 0.380 g) was added butyl vinyl ether (1.0 mmol, 0.101 g). The mixture was stirred at 80 °C for 2 h and then was quenched by 1 M HClaq.. The mixture was extracted with diethyl ether (10 x 3 mL). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography to give the product (0.156 g, 81%).

# (3af) Methyl 2-vinylhexanoate<sup>48</sup>



To a solution of GaBr<sub>3</sub> (0.05 M in dichloroethane, 6 mL) and butylketene methyl trimethylsilyl acetal (4.5 mmol, 0.900 g) was added butyl vinyl ether (3.0 mmol, 0.302 g). The mixture was stirred at 80 °C for 2 h and then was quenched by 1 M HClaq.. The mixture was extracted with diethyl ether (10 x 3

mL). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography to give the product (0.379 g, 82%).

# (3ag) ethyl 2-(tert-butyl)but-3-enoate

$$1 \xrightarrow{2}{3} \xrightarrow{5}{5} \xrightarrow{6} 7$$

To a solution of GaBr<sub>3</sub> (0.0324 g, 0.105 mmol) and *tert*-butylketene ethyl trimethylsilyl acetal (0.321 g, 1.48 mmol) in dichloroethane (2 mL) was added butyl vinyl ether (0.123 g, 1.23 mmol). The mixture was stirred at 80 °C for 2 h and then quenched by diethyl ether (2 mL) and sat.Na<sub>2</sub>CO<sub>3</sub> solution (2 mL). The mixture was extracted with diethyl ether (2 mL x 3). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 98:2, column length 11 cm, diameter 26 mm, Spherical silica gel) to give the product (0.090 g, 43%). IR: (neat) 1734 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 5.94 (dt, J = 16.9, 10.1 Hz, 1H, 2-H), 5.16 (dd, J = 10.1, 1.9 Hz, 1H, 1-H<sup>A</sup>), 5.08 (dd, J = 16.9, 1.9 Hz, 1H, 1-H<sup>B</sup>), 4.15 (dq, J = 10.6, 7.0 Hz, 1H, 6-H<sup>A</sup>), 4.12 (dq, J = 10.6, 7.0 Hz, 1H, 6-H<sup>B</sup>), 2.77 (d, J = 10.1 Hz, 1H, 3-H), 1.26 (t, J = 7.0 Hz, 3H, 7-H<sub>3</sub>), 0.969 (s, 9H, 4-Me<sub>3</sub>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 173.1 (s, C-5), 134.2 (d, C-2), 118.3 (t, C-1), 61.0 (d, C-3), 60.0 (t, C-6), 33.6 (s, C-4), 27.6 (q, 4-Me), 14.2 (q, C-7); MS: (CI, 200 eV) *m/z* 171 (M + 1, 100); HRMS: (CI, 200 eV) Calculated (C<sub>10</sub>H<sub>19</sub>O<sub>2</sub>) 171.1385 (M + 1) Found: 171.1383.

# (3ah) Methyl 2-phenylbut-3-enoate<sup>49</sup>



To a solution of GaBr<sub>3</sub> (0.05 M in dichloroethane, 2 mL) and phenylketene methyl trimethylsilyl acetal (1.5 mmol, 0.338 g) was added butyl vinyl ether (1.0 mmol, 0.102 g). The mixture was stirred at 80 °C for 2 h and then was quenched by 1 M HClaq.. The mixture was extracted with diethyl ether (10 x 3 mL). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography to give the product (0.148 g, 83%).

(3ai) Pentyl but-3-enoate<sup>50</sup>

To a solution of GaBr<sub>3</sub> (0.05 M in dichloroethane, 6 mL) and phenylketene methyl trimethylsilyl acetal (9.0 mmol, 1.96 g) was added butyl vinyl ether (3.0 mmol, 0.305 g). The mixture was stirred at 80 °C for 2 h and then was quenched by 1 M HClaq.. The mixture was extracted with diethyl ether (10 x 3 mL). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography to give the product (0.0621 g, 12%).

#### (3aj) Ethyl 2-(3-thiophenyl)-3-butenoate



To a solution of GaBr<sub>3</sub> (0.05 M in dichloroethane, 2 mL) and 1-ethoxy-2-(3-thiophenyl)-1-(trimethylsiloxy)ethene (1.5 mmol, 0.364 g) was added butyl vinyl ether (1.0 mmol, 0.104 g). The mixture was stirred at 80 °C for 2 h and then was quenched by 1 M HClaq.. The mixture was extracted with diethyl ether (10 x 3 mL). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 98:2, column length 11 cm) (0.151 g, 80%). IR: (neat) 1736 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.25-7.22 (m, 1H, 5'-H), 6.99-6.94 (m, 2H, 2'-H and 4'-H), 6.19 (ddd, J = 17.6, 10.0, 8.8 Hz, 1H, 3-H), 5.25 (d, J = 10.0 Hz, 1H, 4-H<sup>A</sup>), 5.24 (d, J = 17.6, 1H, 4-H<sup>B</sup>), 4.57 (d, J = 8.8 Hz, 1H, 2-H), 4.21 (q, J = 7.2Hz, 2H,  $CH_2CH_3$ ), 1.28 (t, J = 7.2 Hz, 3H,  $CH_2CH_3$ ); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 171.2 (s, C-1), 140.0 (s, C-3'), 135.1 (d, C-3), 126.7 (d), 125.3 (d), 124.9 (d, C-5'), 117.8 (t, C-4), 61.3 (t,  $CH_2CH_3$ ), 51.0 (d, C-2), 14.0 (q,  $CH_2CH_3$ ); MS: (EI, 70 eV) m/z 196 (M<sup>+</sup>, 26), 123 (M<sup>+</sup> - CO<sub>2</sub>CH<sub>2</sub>CH<sub>3</sub>, 100); HRMS: (EI, 70 eV) Calculated (C<sub>10</sub>H<sub>12</sub>O<sub>2</sub>S) 196.0558 (M<sup>+</sup>) Found: 196.0562.

#### (3ak) Methyl 2-(N-methylindol-3-yl)-3-butenoate



To a solution of GaBr<sub>3</sub> (0.05 M in dichloroethane, 2 mL) and *N*-methyl-3-indolylketene methyl trimethylsilyl acetal (1.5 mmol, 0.407 g) was added butyl vinyl ether (1.0 mmol, 0.098 g). The mixture was stirred at 80 °C for 2 h and then was quenched by 1 M HClaq.. The mixture was extracted with diethyl ether (10 x 3 mL). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated

and the residue was purified by column chromatography (hexane/ethyl acetate = 98:2, column length 11 cm) (0.189 g, 84%). IR: (neat) 1736 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.71 (d, J = 8.2 Hz, 1H, 4'-H), 7.35 (t, J = 8.2 Hz, 1H, 7'-H), 7.30 (d, J = 8.2 Hz, 1H, 6'-H), 7.20 (t, J = 8.2 Hz, 1H, 5'-H), 7.10 (s, 1H, 2'-H), 6.28 (ddd, J = 17.2, 10.0, 7.8 Hz, 1H, 3-H), 5.23 (dt, J = 17.2, 1.2 Hz, 1H, 4-H<sup>B</sup>), 5.21 (dt, J = 10.0, 1.2 Hz, 1H, 4-H<sup>A</sup>), 4.61 (dd, J = 7.8, 1.2 Hz, 1H, 2-H), 3.77 (s, 3H), 3.72 (s, 3H); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 173.2 (s, C-1), 136.9 (s, C-8'), 135.3 (d, C-3), 126.9 (d, C-2'), 126.5 (s, C-9'), 121.7 (d, C-6), 119.13 (d), 119.10 (d), 116.8 (t, C-4), 110.8 (s, C-3'), 109.3 (d, C-7'), 52.0 (q, OMe), 47.1 (d, C-2), 32.6 (q, NMe); MS: (EI, 70 eV) m/z 229 (M<sup>+</sup>, 35), 170 (M<sup>+</sup> - CO<sub>2</sub>Me, 100); HRMS: (EI, 70 eV) Calculated (C<sub>14</sub>H<sub>15</sub>NO<sub>2</sub>) 229.1103 (M<sup>+</sup>) Found: 229.1103.

# (3al) Methyl 2-phenoxy-3-butenoate



To a solution of GaBr<sub>3</sub> (0.05 M in dichloroethane, 2 mL) and phenoxyketene methyl trimethylsilyl acetal (1.5 mmol, 0.349 g) was added butyl vinyl ether (1.0 mmol, 0.102 g). The mixture was stirred at 80 °C for 2 h and then was quenched by 1 M HClaq.. The mixture was extracted with diethyl ether (10 x 3 mL). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 98:2, column length 11 cm) (0.085 g, 44%). IR: (neat) 1759 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.29 (t, J = 7.6 Hz, 2H, m), 7.00 (t, J = 7.6 Hz, 1H, p), 6.94 (d, J = 7.6 Hz, 2H, o), 6.07 (ddd, J = 16.8, 10.8, 5.6 Hz, 1H, 3-H), 5.62 (dt, J = 16.8, 1.4 Hz, 1H, 4-H<sup>B</sup>), 5.42 (dt, J = 10.8, 1.4 Hz, 1H, 4-H<sup>A</sup>), 5.18 (dd, J = 5.6, 1.4 Hz, 1H, 2-H), 3.77 (s, 3H, OMe); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 170.0 (s, C-1), 157.1 (s, i), 131.5 (d, C-3), 129.4 (d, m), 121.7 (d, p), 119.2 (t, C-4), 115.2 (d, o), 77.3 (d, C-2), 52.4 (q, OMe); MS: (EI, 70 eV) m/z 192 (M<sup>+</sup>, 33), 160 (31), 133 (M<sup>+</sup> - CO<sub>2</sub>Me, 100), 105 (38), 99 (41), 77 (30); HRMS: (EI, 70 eV) Calculated (C<sub>11</sub>H<sub>12</sub>O<sub>3</sub>) 192.0786 (M<sup>+</sup>) Found: 197.0787.

#### (3am) ethyl 2-(phenylthio)but-3-enoate



To a solution of GaI<sub>3</sub> (0.1032 g, 0.229 mmol) and phenylthioketene ethyl trimethylsilyl acetal (0.8823 g, 2.40 mmol) in dichloroethane (2 mL) was added butyl vinyl ether (0.1974 g, 1.97 mmol). The mixture was stirred at 80 °C for 4 h and then quenched by diethyl ether (2 mL) and sat.NaHCO<sub>3</sub> solution (2 mL). The mixture was extracted with diethyl ether (2 mL x 3). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetat = 98:2, column length 11 cm, diameter 26 mm, Spherical silica gel) to give the product (0.258 g, 59%). IR: (neat) 1736 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.48-7.23 (m, 5H, Ar), 5.96 (ddd, J = 17.4, 10.5, 9.2 Hz, 1H, 2-H), 5.192 (d, J = 17.4 Hz, 1H, 1-H<sup>A</sup>), 5.190 (d, J = 10.5 Hz, 1H, 1-H<sup>B</sup>), 4.28 (d, J = 9.2 Hz, 1H, 3-H), 4.13 (q, J = 7.2 Hz, 2H, 5-H<sub>2</sub>), 1.20 (t, J = 7.2 Hz, 3H, 6-H<sub>3</sub>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 170.0 (s, C-4), 133.6 (d), 132.6 (s, *i*), 132.1 (d, C-2), 128.9 (d), 128.2 (d, *p*), 119.0 (t, C-1), 61.5 (t, C-5), 54.5 (d, C-3), 13.9 (q, C-6); MS: (CI, 200 eV) *m/z* 223 (M+1, 100), 177 (M - OEt, 11), 149 (M - CO<sub>2</sub>Et, 10); HRMS: (EI, 70 eV) Calculated (C<sub>12</sub>H<sub>14</sub>O<sub>2</sub>S) 222.0715 (M<sup>+</sup>) Found: 222.0716.

# (5an) Methyl 2,2-dimethyl-3-methylenenonanoate<sup>13a</sup>



To a solution of GaBr<sub>3</sub> (0.05 M in dichloroethane, 2 mL) and dimethylketene methyl trimethylsilyl acetal (1.5 mmol, 0.270 g) was added 2-methoxyoct-1-ene (1.0 mmol, 0.137 g). The mixture was stirred at 80 °C for 12 h and then was quenched by 1 M HClaq.. The mixture was extracted with diethyl ether (10 x 3 mL). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography to give the product (0.145 g, 71%).

# (5bn) Methyl 2,2-dimethyl-3-methylidene-5-phenylpentanoate



To a solution of GaBr<sub>3</sub> (0.05 M in dichloroethane, 2 mL) and dimethylketene methyl trimethylsilyl acetal (1.5 mmol, 0.270 g) was added 2-methoxy-4-phenyl-1-butene (0.9 mmol, 0.152 g). The mixture was stirred at 80 °C for 12 h and then was quenched by 1 M HClaq.. The mixture was extracted with diethyl ether (10 x 3 mL). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 98:2, column length 11 cm) (0.114 g, 69%). IR: (neat) 1736 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.33-7.27 (m, 2H, *m*), 7.24-7.17 (m, 3H, *o* and *p*), 5.09 (s, 1H, 3-CHH), 5.03 (s, 1H, 3-CHH), 3.67 (s, 3H, OMe), 2.79 (t, J =

8.6 Hz, 2H, 5-H<sub>2</sub>), 2.28 (t, J = 8.6 Hz, 2H, 4-H<sub>2</sub>), 1.36 (s, 6H, 2-Me<sub>2</sub>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 177.0 (s, C-1), 151.4 (s, C-3), 142.0 (s, *i*), 128.32 (d), 128.30 (d), 125.9 (d, *p*), 109.0 (t, 3-CH<sub>2</sub>), 52.0 (q, OMe), 48.2 (s, C-2), 34.9 (t, C-5), 34.0 (t, C-4), 24.8 (q, 2-Me<sub>2</sub>); MS: (CI, 200 eV) *m/z* 233 (M<sup>+</sup> + 1, 100), 173 (25), 131 (27); HRMS: (CI, 200 eV) Calculated (C<sub>15</sub>H<sub>21</sub>O<sub>2</sub>) 233.1542 (M + 1) Found: 233.1540.

#### (5cn) methyl (Z)-2,2-dimethyl-3-methylenedec-8-enoate



To a solution of GaCl<sub>3</sub> (0.0535 g, 0.304 mmol) and dimethylketene methyl trimethylsilyl acetal (0.511 g, 2.93 mmol) in dichloroethane (2 mL) was added (*Z*)-7-bromohept-2-ene (0.204 g, 1.07 mmol). The mixture was stirred at 80 °C for 2 h and then quenched by diethyl ether (2 mL) and sat.NaHCO<sub>3</sub> solution (2 mL). The mixture was extracted with diethyl ether (2 mL x 3). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 98:2, column length 11 cm, diameter 26 mm, Spherical silica gel) to give the product (0.07 g, 30%). IR: (neat) 1736 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 5.48-5.32 (m, 2H, 2-H and 3-H), 4.98 (s, 1H, 9-H<sup>A</sup>), 4.89 (s, 1H, 9-H<sup>B</sup>), 3.65 (s, 3H, 12-H<sub>3</sub>), 2.03 (dt, *J* = 7.2, 7.3 Hz, 2H, 4-H<sub>2</sub>), 1.95 (t, *J* = 7.7 Hz, 2H, 7-H<sub>2</sub>), 1.59 (d, *J* = 6.3 Hz, 3H, 1-H<sub>3</sub>), 1.51-1.41 (m, 2H, 6-H<sub>2</sub>), 1.41-133 (m, 2H, 5-H<sub>2</sub>), 1.31 (s, 6H, 10-Me<sub>2</sub>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 177.2 (s, C-11), 152.0 (s, C-8), 130.5 (d, C-2), 123.8 (d, C-3), 108.4 (t, C-9), 51.9 (q, C-12), 48.1 (s, C-10), 32.4 (t, C-7), 29.3 (t, C-5), 28.0 (t, C-6), 26.7 (t, C-4), 24.8 (q, 10-Me), 12.7 (q, C-1); HRMS: (EI, 90 eV) Calculated (C<sub>14</sub>H<sub>24</sub>O<sub>2</sub>) 224.1776 (M<sup>+</sup>) Found: 224.1774.

#### (5dn) methyl 10-chloro-2,2-dimethyl-3-methylenedecanoate



To a solution of GaCl<sub>3</sub> (0.0535 g, 0.304 mmol) and dimethyllketene methyl trimethylsilyl acetal (0.511 g, 2.93 mmol) in dichloroethane (2 mL) was added 9-chloro-2-methoxynon-1-ene (0.262 g, 1.10 mmol). The mixture was stirred at 80 °C for 2 h and then quenched by diethyl ether (2 mL) and sat. Na<sub>2</sub>CO<sub>3</sub> solution (2 mL). The mixture was extracted with diethyl ether (2 mL x 3). The collected organic layer

was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 98:2, column length 11 cm, diameter 26 mm, Spherical silica gel) to give the product (0.20 g, 72%). IR: (neat) 1736 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 5.00 (s, 1H, 9-H<sup>A</sup>), 4.90 (s, 1H, 9-H<sup>B</sup>), 3.66 (s, 3H, 12-H<sub>3</sub>), 3.52 (t, J = 6.4 Hz, 2H, 1-H<sub>2</sub>), 1.93 (t, J = 7.7 Hz, 2H, 7-H<sub>2</sub>), 1.76 (tt, J = 7.3, 7.2 Hz, 2H, 2-H<sub>2</sub>), 1.52-1.39 (m, 4H, 3-H<sub>2</sub> and 6-H<sub>2</sub>), 1.37-1.27 (m, 10H, 4-H<sub>2</sub>, 5-H<sub>2</sub> and 9-Me<sub>2</sub>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 177.2 (s, C-11), 152.0 (s, C-8), 108.4 (t, C-9), 51.9 (q, C-12), 48.2 (s, C-10), 45.1 (t, C-1), 32.6 (t, C-2), 32.0 (t, C-7), 29.3 (t), 28.8 (t), 28.3 (t), 26.8 (t, C-3), 24.8 (q, 10-Me<sub>2</sub>); HRMS: (EI, 90 eV) Calculated (C<sub>14</sub>H<sub>25</sub>ClO<sub>2</sub>) 260.1543 (M<sup>+</sup>) Found: 260.1537

# (5en) methyl 12-methoxy-2,2-dimethyl-3-methylenedodecanoate



To a solution of GaCl<sub>3</sub> (0.0525 g, 0.298 mmol) and dimethyllketene methyl trimethylsilyl acetal (0.519 g, 2.98 mmol) in dichloroethane (2 mL) was added 2,10-dimethoxydec-1-ene (0.166 g, 0.774 mmol). The mixture was stirred at 80 °C for 2 h and then quenched by diethyl ether (2 mL) and sat. Na<sub>2</sub>CO<sub>3</sub> solution (2 mL). The mixture was extracted with diethyl ether (2 mL x 3). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 98:2, column length 11 cm, diameter 26 mm, Spherical silica gel) to give the product (0.123 g, 56%). IR: (neat) 1736 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 4.91 (s, 1H, 12-H<sup>A</sup>), 4.82 (s, 1H, 12-H<sup>B</sup>), 3.66 (s, 3H, 15-H<sub>3</sub>), 3.36 (t, *J* = 6.2 Hz, 2H, 2-H<sub>2</sub>), 3.33 (s, 3H, 1-H<sub>3</sub>), 1.94 (t, *J* = 7.7 Hz, 2H, 10-H<sub>2</sub>), 1.59-1.53 (m, 2H, 3-H<sub>2</sub>), 1.48-1.41 (m, 2H, 9-H<sub>2</sub>), 1.32 (s, 6H, 13-Me<sub>2</sub>), 1.39-1.18 (m, 10H, 4-H<sub>2</sub>, 5-H<sub>2</sub>, 6-H<sub>2</sub>, 7-H<sub>2</sub> and 8-H<sub>2</sub>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 177.2 (s, C-14), 152.1 (s, C-11), 108.3 (t, C-12), 72.9 (t, C-2), 58.4 (q, C-1), 51.9 (q, C-15), 48.1 (s, C-13), 32.1 (t, C-10), 29.6 (t, C-3), 29.5 (t), 29.44 (t), 29.41 (t), 28.4 (t, C-9), 26.1 (t), 24.8 (q, 13-Me); HRMS: (EI, 90 eV) Calculated (C<sub>17</sub>H<sub>32</sub>O<sub>3</sub>) 284.2351 (M<sup>+</sup>) Found: 284.2354.

#### (5fn) Methyl 2,2-dimethyl-3-methylidene-5-phenylpentynoate



To a solution of GaBr<sub>3</sub> (0.05 M in dichloroethane, 2 mL) and dimethylketene methyl trimethylsilyl

acetal (1.5 mmol, 0.262 g) was added 2-methoxy-4-phenyl-1-buten-3-yne (1.0 mmol, 0.154 g). The mixture was stirred at 80 °C for 12 h and then was quenched by 1 M HClaq.. The mixture was extracted with diethyl ether (10 x 3 mL). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 98:2, column length 11 cm) (0.162 g, 73%). IR: (neat) 1736 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.47-7.39 (m, 2H, o), 7.33-7.26 (m, 3H, m and p), 5.58 (s, 1H, 3-CHH), 5.45 (s, 1H, 3-CHH), 3.72 (s, 3H, OMe), 1.47 (s, 6H, 2-Me<sub>2</sub>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 175.8 (s, C-1), 135.4 (s, C-3), 131.5 (d, o), 128.24 (d, p), 128.20 (d, m), 122.9 (s, i), 119.8 (t, 3-CH<sub>2</sub>), 90.7 (s, C-5), 87.8 (s, C-4), 52.2 (q, OMe), 47.8 (s, C-2), 24.6 (q, 2-Me<sub>2</sub>); MS: (EI, 70 eV) m/z 228 (M<sup>+</sup>, 75), 213 (65), 200 (20), 169 (49), 168 (M<sup>+</sup> - CO<sub>2</sub>Me, 100), 167 (21), 154 (25), 153 (27), 127 (36), 115 (39); HRMS: (EI, 70 eV) Calculated (C<sub>15</sub>H<sub>16</sub>O<sub>2</sub>) 228.1150 (M<sup>+</sup>) Found: 228.1154.

# (5gn) Methyl 2,2-dimethyl-3-phenylbut-3-enoate<sup>14</sup>



To a solution of GaBr<sub>3</sub> (0.1 M in dichloroethane, 2 mL) and dimethylketene methyl trimethylsilyl acetal (3.0 mmol, 0.526 g) was added (1-methoxyvinyl)benzene (1.0 mmol, 0.124 g). The mixture was stirred at 80 °C for 12 h and then was quenched by 1 M HClaq.. The mixture was extracted with diethyl ether (10 x 3 mL). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography to give the product (0.086 g, 43%).

# (5hn) Methyl 2,2-dimethyl-3-(p-tolyl)but-3-enoate<sup>13a</sup>



To a solution of GaBr<sub>3</sub> (0.1 M in dichloroethane, 2 mL) and dimethylketene methyl trimethylsilyl acetal (3.0 mmol, 0.521 g) was added 1-(1-methoxyvinyl)-4-methylbenzene (1.0 mmol, 0.141 g). The mixture was stirred at 80 °C for 12 h and then was quenched by 1 M HCl aq. The mixture was extracted with diethyl ether (10 x 3 mL). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography to give the product (0.089 g, 43%).

# (5in) Methyl 3-(4-chlorophenyl)-2,2-dimethylbut-3-enoate<sup>13a</sup>



To a solution of GaBr<sub>3</sub> (0.1 M in dichloroethane, 2 mL) and dimethylketene methyl trimethylsilyl acetal (3.0 mmol, 0.517 g) was added 1-chloro-4-(1-methoxyvinyl)benzene (1.0 mmol, 0.173 g). The mixture was stirred at 80 °C for 12 h and then was quenched by 1 M HCl aq. The mixture was extracted with diethyl ether (10 x 3 mL). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography to give the product (0.148 g, 61%).

# (5jn) Methyl 2,2-dimethyl-3-(2-naphthyl)-3-butenoate



To a solution of GaBr<sub>3</sub> (0.1 M in dichloroethane, 2 mL) and dimethylketene methyl trimethylsilyl acetal (2.9 mmol, 0.512 g) was added 1-(2-naphthyl)-1-methoxyethene (1.0 mmol, 0.187 g). The mixture was stirred at 80 °C for 12 h and then was quenched by 1 M HCl aq. The mixture was extracted with diethyl ether (10 x 3 mL). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 98:2, column length 11 cm) (0.149 g, 59%). IR: (neat) 1732 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.87-7.79 (m, 2H, 5'-H and 8'-H), 7.77 (d, J = 8.8 Hz, 1H, 4'-H), 7.62 (d, J = 1.6 Hz, 1H, 1'-H), 7.52-7.43 (m, 2H, 6'-H and 7'-H), 7.30 (dd, J = 8.8, 1.6 Hz, 1H, 3'-H), 5.43 (s, 1H, 4-CHH), 5.26 (s, 1H, 4-CHH), 3.71 (s, 3H, OMe), 1.46 (s, 6H, 2-Me<sub>2</sub>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 177.0 (s, C-1), 153.0 (s, C-3), 139.1 (s, C-2'), 133.0 (s, C-9'), 132.4 (s, C-10'), 128.0 (d), 127.5 (d), 127.3 (d), 126.6 (d), 126.4 (d), 126.0 (d), 125.8 (d), 115.0 (t, C-4), 52.1 (q, OMe), 47.7 (s, C-2), 25.8 (q, 2-Me<sub>2</sub>); MS: (EI, 70 eV) *m*/*z* 254 (M<sup>+</sup>, 60), 239 (M<sup>+</sup> - CH<sub>3</sub>, 100), 195 (25), 179 (38), 165 (22), 153 (73), 152 (31), 141 (33); HRMS: (EI, 70 eV) Calculated (C<sub>17</sub>H<sub>18</sub>O<sub>2</sub>) 254.1307 (M<sup>+</sup>) Found: 254.1304; Analysis: C<sub>17</sub>H<sub>18</sub>O<sub>2</sub> (254.32) Calcd: C, 80.28; H, 7.13 Found: C, 79.99; H, 7.27.

#### Phenethyl 2,2-bis(methyl-d<sub>3</sub>)but-3-enoate (Figure2)



To a solution of GaBr<sub>3</sub> (0.0169 g, 0.0546 mmol) and trimethyl ((2-(methyl-d<sub>3</sub>)-1-phenethoxyprop-1-en-1-yl-3,3,3-d<sub>3</sub>)oxy)silane (0.135 g, 0.499 mmol) in dichloroethane (2 mL) was added butyl vinyl ether (0.1237 g, 1.24 mmol). The mixture was stirred at 80 °C for 2 h and then quenched by diethyl ether (2 mL) and sat.Na<sub>2</sub>CO<sub>3</sub> solution (2 mL). The mixture was extracted with diethyl ether (2 mL x 3). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 98:2, column length 11 cm, diameter 26 mm, Spherical silica gel) to give the product (0.0938 g, 84%). IR: (neat) 1732 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.31-7.20 (m, 5H, Ar), 5.98 (dd, *J* = 17.4, 10.6 Hz, 1H, 2-H), 5.05 (d, *J* = 17.4 Hz, 1H, 1-H<sup>A</sup>), 5.04 (d, *J* = 10.6 Hz, 1H, 1-H<sup>B</sup>), 4.28 (t, *J* = 6.8 Hz, 2H, 5-H<sub>2</sub>), 2.93 (t, *J* = 6.8 Hz, 2H, 6-H<sub>2</sub>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 176.2 (s, C-4), 142.4 (d, C-2), 137.8 (s, *i*), 128.9 (d), 128.4 (d), 126.4 (d, *p*), 112.8 (t, C-1), 65.1 (t, C-5), 35.0 (t, C-6), 23.5 (m, 3-CD<sub>3</sub>); MS: (CI, 200 eV) *m/z* 225 (M+1, 100), 105 (CH<sub>2</sub>CH<sub>2</sub>Ph, 83); HRMS: (CI, 200 eV) Calculated (C<sub>14</sub>H<sub>13</sub>D<sub>6</sub>O<sub>2</sub>) 225.1762 (M + 1) Found: 225.1758.

#### 1-5. References

- a) Metal-catalyzed Cross-coupling Reactions; Diederich, F., Stang, P. J., Eds.; Wiley-VCH: Weinheim, Germany, 1998; b) Cross-Coupling Reactions. A Practical Guide; Miyaura, N., Ed.; Springer-Verlag: Berlin, 2002; c) Metal-Catalyzed Cross-Coupling Reactions; de Meijere, A., Diederich, F., Eds.; Wiley-VCH: Weinheim, Germany, 2004; d) J. Tsuji, Palladium Reagents and Catalysts: New Perspectives for the 21st Century, Wiley, New York, 2003; e) Corber, J. P.; Mignani, G. *Chem. Rev.* 2006, *106*, 2651; f) Nicolaou, K. C.; Bulger, P. G.; Sarlah, D. *Angew. Chem. Int. Ed.* 2005, *44*, 4442.
- 2 For selected reviews: a) D.-G. Yu, B.-J. Li, Z.-J. Shi, Acc. Chem. Res. 2010, 43, 1486; b) B.-J. Li, D.-G. Yu, C.-L. Sun, Z.-J. Shi, Chem.–Eur. J. 2011, 17, 1728; c) B. M. Rosen, K. W. Quasdorf, D. A. Wilson, N. Zhang,; A.-M. Resmerita, N. K. Garg, V. Percec, Chem. Rev. 2011, 111, 1346; d) W.-N. Li, Z.-L. Wang, RSC Adv. 2013, 3, 25565.
- 3 a) E. Wenkert, E. L. Michelotti, C. S. Swindell, J. Am. Chem. Soc. 1979, 101, 2246; b) E. Wenkert, E. L. Michelotti, C. S. Swindell, M. Tingoli, J. Org. Chem. 1984, 49, 4894; c) Y. Matsuura, M. Tamura, T. Kochi, M. Sato, N. Chatani, F. Kakiuchi, J. Am. Chem. Soc. 2007, 129, 9858; d) S, Zhu, G. Jin, H. Jiang, Can. J. Chem. 2005, 83, 2127; e) B. T. Guan, S. K. Xiang, T. Wu, Z. P. Sun, B. Q. Wang, K. Q. Zhaob, Z. J. Shi, Chem. Commun. 2008, 1437; f) T. Shimasaki, Y. Konno, M. Tobisu, N. Chatani, Org. Lett. 2009, 11, 4890; g) A. J. Brockway, M. González-López, J. C. Fettinger, J. T. Shaw, J. Org. Chem. 2011, 76, 3515; h) L. G. Xie, Z. X. Wang, Chem. Eur. J. 2011, 17, 4972; i) T. Iwasaki, Y. Miyata, R. Akimoto, Y. Fujii, H. Kuniyasu, N. Kambe, J. Am. Chem. Soc. 2014, 136, 9260.
- 4 a) K. W. Quasdorf, X. Tian, N. K. Garg, J. Am. Chem. Soc. 2008, 130, 14422; b) B. J. Li, Y. Z. Li, X. Y. Lu, J. Liu, B. T. Guan, Z. J. Shi, Angew. Chem. Int. Ed. 2008, 47, 10124; c) J. Y. Yu, R. Kuwano, Angew. Chem., Int. Ed. 2009, 48, 7217; d) H. W. Lee, F. Y. Kwong, Synlett 2009, 3151; e) B. J. Li, L. Xu, Z. H. Wu, B. T.

Guan, C. L. Sun, B. Q. Wang, Z. J. Shi, *J. Am. Chem. Soc.* **2009**, *131*, 14656; f) D. Gärtner, A. L. Stein, S. Grupe, J. Arp, A. J. von Wangelin, *Angew. Chem. Int. Ed.* **2015**, *54*, 10545.

- 5 a) F. H. Porée, A. Clavel, J. F. Betzer, A. Pancrazi, J. Ardisson, *Tetrahedron Lett.* 2003, 44, 7553; b) L. Xu,
  B. J. Li, Z. H. Wu, X. U. Lu, B. T. Guan, B. Q. Wang, K. Q. Zhao, Z. J. Shi, *Org. Lett.* 2010, *12*, 884.
- 6 a) A. S. E. Karlström, K. Itami, J. E. Bäckvall, J. Org. Chem. 1999, 64, 1745; b) Y. Nan, Z. Yang, Tetrahedron Lett. 1999, 40, 3321.
- 7 T. Hayashi, Y. Katsuro, M. Kumada, Tetrahedron Lett. 1980, 21, 3915.
- 8 C. Chen, R. F. Jordan, J. Am. Chem. Soc. 2010, 132, 10254.
- 9 Y. Nishimoto, T. Nishimura, M. Yasuda, Chem. Eur. J. 2015, 21, 18301.
- 10 As far as a coupling with arenol derivatives, Heck reactions using alkenes and direct C-O/C-H cross-couplings using heteroarenes and carbonyl compounds have been recently established. Heck reaction; a) A. R. Ehle, Q. Zhou, M. P. Watson, *Org. Lett.* 2012, *14*, 1202. Direct C-O/C-H cross-couplings using heteroarenes and carbonyl compounds; b) K. Muto, J. Yamaguchi, A. Lei, K. Itami, *J. Am. Chem. Soc.* 2013, *135*, 16384; c) R. Takise, K. Muto, J. Yamaguchi, K. Itami, *Angew. Chem. Int. Ed.* 2014, *53*, 6791; d) M. Moselage, N. Sauermann, S. C. Richter, L. Ackermann, *Angew. Chem. Int. Ed.* 2015, *54*, 6352; e) E. Koch, R. Takise, A. Studer, J. Yamaguchi, K. Itami, *Chem. Commun.* 2015, *51*, 855; f) T. Chen, L.-B. Han, *Angew. Chem. Int. Ed.* 2015, *54*, 8600.
- 11 Y. Nishimoto, H. Ueda, M. Yasuda, A. Baba, Angew. Chem. Int. Ed. 2012, 51, 8073.
- 12 The coupling reaction of carbonyl group-substituted enol ethers via 1,4-addition/elimination in the stoichiometric amount of a base: a) N. Zanatta, R. Barichello, H. G. Bonacorso, M. A. P. Martins, *Synthesis* 1999, 765; b) S. Zhu, G. Jin, H. Jiang, *Can. J. Chem.* 2005, *83*, 2127.
- 13 a) Y. Nishimoto, R. Moritoh, M. Yasuda, A. Baba, *Angew. Chem. Int. Ed.* 2009, *48*, 4577; b) Y. Nishimoto,
  H. Ueda, Y. Inamoto, M. Yasuda, A. Baba, *Org. Lett.* 2010, *12*, 3390.
- 14 Y. Nishimoto, H. Ueda, M. Yasuda, A. Baba, Chem. Eur. J. 2011, 17, 11135.
- 15 a) Y. Nishimoto, M. Takeuchi, M. Yasuda, A. Baba, A. *Chem. Eur. J.* 2013, *19*, 14411; b) Y. Nishimoto,
   M. Takeuchi, M. Yasuda, A. Baba, *Angew. Chem. Int. Ed.* 2012, *51*, 1051.
- 16 Probably, temporarily-generated Me<sub>3</sub>SiBr does ill to functional groups such as alkene, chloro, and methoxy groups.
- 17 A Values: isopropyl group (2.15 kcalmol<sup>-1</sup>), phenyl group (2.7 kcalmol<sup>-1</sup>), *tert*-butyl group (4.9 kcalmol<sup>-1</sup>); M. B. Smith, J. March, March's Advanced Organic Chemistry: Reactions, Mechanisms, and Structure 5<sup>th</sup> ed., John Wiley & Sons, Inc. New York, **2001**.
- 18 We reported the sequential addition of two different carbon nucleophiles to esters via Claisen condensation catalyzed by gallium trihalide. Y. Inamoto, Y. Kaga, Y. Nishimoto, M. Yasuda, A. Baba, *Chem. Eur. J.* 2014, *37*, 11664.

- NMR spectra of alkylgallium dihalides; a) O. T. Beachley, Jr., M. J. Noble, M. R. Churchill, C. H. Lake, *Organometallics*, **1992**, *11*, 1051; b) J. Ohshita, H. Schmidbaur, *J. Organomet. Chem.* **1993**, *453*, 7; c) O. T. Beachley, Jr., J. C. Pazik, M. J. Noble, *Organometallics*, **1994**, *13*, 2885; d) O. T. Beachley, Jr., D. B. Rosenblum, M. R. Churchill, D. G. Churchill, L. M. Krajkowski, *Organometallics* **1999**, *18*, 2543.
- 20 This structural characteristic is consistent with our reported alkylindium compound synthesized by carboindation among InBr<sub>3</sub>, octene, and dimethylketene trimethylsilyl methyl acetal (Ref. 13b).
- 21 See Experimental Section for NMR spectra of Scheme 4.
- 22 Spectroscopic Methods in Organic Chemistry; M. Hesse, H. Meier, B. Zeeh, Ed., Georg Thieme Verlag KG, Stuttgart, Germany.
- 23 R. U. Lemieux, B. Fraser-Reid, Can. J. Chem. 1965, 43, 1460.
- 24 For the addition/elimination mechanism in GaCl<sub>3</sub>-catalyzed reactions, see; a) R. Amemiya, A. Fujii, M. Arisawa, M. Yamaguchi, *J. Organomet. Chem.* 2003, 686, 94; b) R. Amemiya, K. Suwa, J. Toriyama, Y. Nishimura, M. Yamaguchi, *J. Am. Chem. Soc.* 2005, *127*, 8252; c) R. Amemiya, Y. Miyake, M. Yamaguchi, *Tetrahedron Lett.* 2006, 47, 1797; d) Y. Nishimura, R. Amemiya, M. Yamaguchi, *Tetrahedron Lett.* 2006, 47, 1839.
- 25 β-Eliminations from β-oxyalkylmetals. Reviews for Peterson olefinations; a) D. J. Ager, *Synthesis* 1984, 384; b) A. G. M. Barrett, J. M. Hill, E. M. Wallace, J. A. Flygare, *Synlett* 1991, 764; c) L. F. van Staden, D. Gravestock, D. J. Ager, *Chem. Soc. Rev.* 2002, *31*, 195. β-Eliminations from β-oxyalkylmetals. Peterson olefinations using β-hydroxymetals other than β-hydroxysilanes: D. J. Ager, *Org. React.* 1990, *38*, 1.
- 26 The injection of Me<sub>3</sub>SiBr had no influence on the rate (Experimental Section), which suggested that Me<sub>3</sub>SiBr does not take part in the *syn* elimination step unlike the mechanism we proposed in the primary report.
- 27 See Experimental Section for the derivation of rate equations.
- 28 See Experimental Section for the details of Arrhenius plot.
- 29 Preparation: Y. Kimura, S. L. Regen, J. Org. Chem. 1982, 47, 2493. Characterization Data: S.-R. Sheng,
   X.-L. Liu, X.-C. Wang, Q. Xin, C.-S. Song, Synthesis 2004, 2833.
- 30 Preparation: H. Keul, R. L. Kuczkowski, H.-S. Choi, J. Org. Chem. 1985, 50, 3365.
- 31 Characterization Data: A. Iida, J. Osada, R. Nagase, T. Misaki, Y. Tanabe, Org. Lett. 2007, 9, 1859.
- 32 Characterization Data: M. T. Reetz, K. Schwellnus, F. Huebner, W. Massa, R. E. Schmidt, *Chem. Ber*. **1983**, *116*, 3708.
- 33 Characterization Data: S. Stanković, J. H. Espenson, J. Org. Chem. 2000, 65, 5528.
- 34 J. E. Bubois, G. Axiotis, E, Bertounesque Tetrahedron Lett. 1984, 25, 4655.
- 35 Characterization Data: S. Stankovic, J. H. Espenson, J. Org. Chem. 2000, 65, 5528.
- 36 Characterization Data: N. Slougui, G. Rousseau, Synthetic Communications, 1987, 17, 1.

- 37 Characterization Data: W. Adam, K. Takayama, J. Org. Chem. 1980, 45, 447.
- 38 Characterization Data: I. Ojima, S. Inaba, M. Nagai, Synthesis 1981, 545.
- 39 D. Katayev, V. Matouek, R. Koller, A. Togni, Org. Lett. 2015, 17, 5898.
- 40 Preparation: G. Cabrera, R. Fiaschi, E. Napolitano, *Tetrahedron Lett.* **2001**, *42*, 5867. Characterization Data: B. José, A. Fernando, B. Miguel, *Synthesis* **1988**, 144.
- 41 Characterization Data: F. Taherirastgar, L. Brandsma, Chemische Berichte 1997, 130, 45.
- 42 Preparation: D. W. Nikolaos, A. H. Diego, J. K. Thomas, J. Org. Chem. 1994, 59, 1889. Characterization Data: J. P. Richard, K. B. Williams, J. Am. Chem. Soc. 2007, 129, 6952.
- 43 Characterization Data: S. H. Pine, R. J. Pettit, G. D. Geib, S. G. Cruz, C. H. Gallego, *J. Org. Chem.* **1985**, 50, 1212.
- 44 P. M. Savu, J. A. Katzenellenbogen, J. Org. Chem. 1981, 46, 239.
- 45 I. Viera, E. Manta, L. Gonzalez, G. Mahler, Tetrahedron Asymmetry 2010, 21, 631.
- 46 A. S. Kende, P. Fludzinski, J. H. Hill, W. Swenson, J. Clardy, J. Am. Chem. Soc. 1984, 106, 3551.
- 47 K. E. Murphy, A. H. Hoveyda, Org. Lett. 2005, 7, 1255.
- 48 V. Calo, L. Lopez, G. Pesce, Gazzetta Chimica Italiana 1990, 120, 203.
- 49 H. Iida, Y. Watanabe, C. Kibayashi, Chem. Pharm. Bull. 1985, 33, 351.
- 50 J. M. Larsson, T. S. N. Zhao, K. J. Szabo, Org. Lett. 2011, 13, 1888.

# Chapter 2

# Indium Catalyzed Hydrofunctionalization of Styrene Derivatives Bearing a Hydroxy Group with Organosilicon Nucleophiles

#### 2-1. Introduction

Alkenes are important fundamental materials in organic synthesis and industrial chemistry, and many useful transformations of alkenes have been established.<sup>1</sup> Therefore, the direct addition reaction of alkenes with various reagents such as electrophilic, nucleophilic, and radical ones to give functionalized alkanes with high selectivity and high step economy is highly desired even now. In this context, many examples for hydrofunctionalization of alkenes through the addition of E-H (E = C, O, N, P, and S) to alkenes have already been reported.<sup>2</sup> One example involves the hydrofunctionalization of Michael acceptors such as  $\alpha,\beta$ -unsaturated carbonyl compounds or nitriles with a variety of organosilicon nucleophiles, which is a well-developed method<sup>3</sup> (Scheme 1-A). However, the application of less-polar alkenes to the process of hydrofunctionalization has been insufficiently developed, and available organosilicon nucleophiles are limited to silvl enolates, silvl cyanides and silvl azides<sup>4</sup> (Scheme 1-B). Our group also reported the regioselective hydrofunctionalization of simple alkenes with silvl ketene acetals and a stoichiometric amount of indium<sup>4g</sup> or bismuth salts<sup>4f</sup> through carboindation or carbobismuthination, respectively. Herein, the regioselective hydrofunctionalization of less-polar alkenes bearing a hydroxy group with organosilicon nucleophiles was accomplished in the presence of catalytic amount of indium trihalide (Scheme 1-C). The present reaction proceeded through a unique reaction mechanism including Brønsted acid-catalyzed hydroalkoxylaiton of alkenes followed by InI<sub>3</sub>catalyzed substitution reaction of cyclic ether intermediates, which is quite different from our previously reported method through carbometalation. In the present hydrofunctionalization, various organosilicon nucleophiles as well as silvl ketene acetals were allowed to use. Moreover, not only mono-substituted alkenes but also di- and tri-substituted alkenes, which often showed low reactivity for previously developed hydrofunctionalization,<sup>4c,4f,4g,4j</sup> were applicable. It is noted that a wide range of functional groups could be introduced into alkenes via a single synthetic operation.<sup>5</sup>

(A) Michael Addition<sup>[3]</sup>  

$$EWG \checkmark + Si-E + H^+ \text{ source} \longrightarrow H^+$$
  
 $EWG = C(O)R, NO_2$  EWG is necessary.

(B) Reported works using less-polar alkenes<sup>[4]</sup>



E = alkyl,  $\alpha$ -carbonylalkyl, alkenyl, alkynyl, allenyl, cyano, azide, sulfide

**Scheme 1.** (A) Michael addition. (B) Hydrofunctionalization by organosilicon nucleophiles with less-polar alkenes. (C) This work allows the introduction of versatile functional groups to alkenes.

#### 2-2. Results and Discussion

#### **Optimization of Reaction Conditions**

The reaction of styrene with silvl ketene acetal 2a was attempted in the presence of a catalytic amount of indium tribromide, but instead of obtaining the hydrofunctionalization product, the polymerization of styrene occurred (Scheme 2-A). Interestingly, we found that o-(2-hydroxyethyl)styrene (1a) afforded the target product 3aa in a high yield (Scheme 2-B).



Scheme 2. Investigation of indium bromide catalyzed reaction of silyl ketene acetal with styrene (A) or styrene derivative 1a (B).

That result prompted us to investigate the effect of acid catalysts on the hydrofunctionalization of **1a** with **2a** (Table 1). As shown in Scheme 2-(B), InBr<sub>3</sub> gave **3aa** in a 76% yield (entry 1). GaBr<sub>3</sub> also afforded **3aa** in approximately the same yield (entry 2). On the other hand, the desired product **3aa** was not obtained, and starting alkene **1a** and TMS-**1a** were recovered in the cases of the reaction using InCl<sub>3</sub>, In(OTf)<sub>3</sub> or BiI<sub>3</sub> (entries 3-5). The reaction catalyzed by InI<sub>3</sub> raised the yield of **3aa** to 92% (entry 6). Without the addition of **2a**, only cyclic ether **4** was produced in a high yield (entry 7). Thus, the

hydrofunctionalization product could be obtained via **4**. Typical Lewis acids such as  $ZnCl_2$ , AlCl<sub>3</sub>, and TiCl<sub>4</sub> were ineffective for the desired hydrofunctionalization and either gave the cyclic ether **4** or recovered the starting material (entries 8-10). The reaction using BF<sub>3</sub>·OEt<sub>2</sub> resulted in complicated mixture (entry 11). A cationic gold catalyst, which is often used to functionalize alkenes,<sup>6</sup> was ineffective (entry 12). The reaction catalyzed by HI aq. would not afford the product **3aa**, but did produce cyclic ether **4** in a high yield (entry 13).

 Table 2. The effect of acid catalysts on the hydrofunctionalization of styrene derivative 1a with organosilicon nucleophile 2a.<sup>[a]</sup>

	OH Cata (10 r CICH <sub>2</sub> <b>1a</b> (1.0 equiv) 80 °C	alyst nol %) <b>2a</b> (1.5 c 2CH <sub>2</sub> Cl RT, 2 2, time	SiMe <sub>3</sub> OMe equiv) 2 h 3aa	OH H + CO <sub>2</sub> Me		OSiMe <sub>3</sub> IS-1a
Entry	Catalyst	Time (h)	Yield (%) <sup>[b]</sup>		Recovery of 1a (%)	
			3aa	4	TMS-1a	
1	InBr <sub>3</sub>	24	76	0	0	0
2	GaBr <sub>3</sub>	24	72	0	0	0
3	InCl <sub>3</sub>	96	0	16	33	15
4	In(OTf) <sub>3</sub>	24	0	40	3	5
5	BiI <sub>3</sub>	72	0	0	91	0
6	InI <sub>3</sub>	24	92 (71) <sup>[c]</sup>	0	0	0
7 <sup>[d]</sup>	InI <sub>3</sub>	24	0	90	0	0
8	ZnCl <sub>2</sub>	96	0	0	45	8
9	AlCl <sub>3</sub>	24	1	56	0	0
10	TiCl <sub>4</sub>	48	0	57	0	0
11	BF <sub>3</sub> ·OEt <sub>2</sub>	24	0	0	0	0
12	AuCl <sub>3</sub> /AgOTf	24	0	52	0	0
13	HI aq.	24	0	77	0	0

[a] 1st step: **1a** (0.5 mmol), catalyst (0.05 mmol), ClCH<sub>2</sub>CH<sub>2</sub>Cl (1 mL), 80 °C, 24 h. 2nd step: **2a** (0.75 mmol), RT, 2 h. The reaction was quenched with Bu<sub>4</sub>NF (1 M in THF). [b] Yields were determined by <sup>1</sup>H NMR using internal standards (bromoform). [c] Isolated yield. [d] The 2nd step was not carried out.

#### Substrate Scope of Organosilicon Nucleophiles

Reactions of **1a** with various organosilicon nucleophiles **2** catalyzed by InI<sub>3</sub> were carried out (Table 2). The bulky silyl ketene acetal **2b** was suitable for this reaction system to provide the cyclohexane derivative **3ab** in high yield (entry 1). The silyl enol ether **2c** afforded the target product **3ac** (entry 2). Hydrofunctionalization with the silyl cyanide **2d** occurred (entry 3), but the reported method<sup>4b</sup> that uses a Brønsted acid for hydrocyanation of styrene gave only cyclic ether **4** (see Scheme S1 in Supporting Information for detail). Methallylsilane **2e** also provided the hydrofunctionalization product in moderate yield (entry 4).<sup>7</sup> Reactions using allylsilane **2f**, alkenylsilane **2g**, alkynylsilane **2h**, and propargylsilane **2i** proceeded efficiently in the presence of a catalytic amount of InI<sub>3</sub>/Me<sub>3</sub>SiBr (entries 5-8). The addition of Me<sub>3</sub>SiBr was necessary for the reactions using nucleophiles **2f-2i**, which have a relatively low level of nucleophilicity. In these cases, the combination of InX<sub>3</sub> with Me<sub>3</sub>SiX showed high Lewis acidity.<sup>8</sup> To the best of our knowledge, the reaction using **2i** is the first example of an intermolecular Markovnikov hydroallenylation toward less-polar alkenes.<sup>9</sup> Heteroatom nucleophiles such as silyl sulfide **2j** and silyl azide **2k** were applicable to this reaction system (entries 9 and 10).<sup>10</sup>

		<i>Nu-</i> SiMe <sub>3</sub> (10 mol %) Me <sub>3</sub> SiBr CH <sub>2</sub> CH <sub>2</sub> CI RT,	(y mol %) 2-14 h
	8( <b>1a</b> (1.0 equiv)	) °C, 24 h	3ab-3ak H
Entry	<i>Nu</i> -SiMe <sub>3</sub> (x equiv)	Me <sub>3</sub> SiBr (y mol%)	Yield of <b>3</b> (%) <sup>[b]</sup>
1[c]		none	OH O OMe 3ab 74
2	OSiMe <sub>3</sub>	none	OH
	<b>2c</b> (3.0 equiv)		<b>`</b> н <sup>°</sup> <b>Зас</b> 69
3 <sup>[d]</sup>	Me <sub>3</sub> SiCN	none	OH
	<b>2d</b> (5.0 equiv)		с С <sub>н</sub> <b>3ad</b> 31
4	SiMe <sub>3</sub> 2e (1.5 equiv	none	он н <b>Зае</b> 55

Table 2. Scope of hydrofunctionalization using various organosilicon nucleophiles.<sup>[a]</sup>



[a] 1st step: **1a** (0.5 mmol),  $InI_3$  (0.05 mmol),  $CICH_2CH_2Cl$  (1 mL), 80 °C, 24 h. 2nd step: **2** (0.75-2.5 mmol), Me<sub>3</sub>SiBr (0-0.1 mmol), RT, 2-14 h (See Experimental Section for detail). The reaction was quenched with Bu<sub>4</sub>NF (1 M in THF). [b] Isolated yield. [c]  $InI_3$  (20 mol%) [d]  $InI_3$  (20 mol%), 70 °C at the second step. [e] 35 °C at the second step. [f] 50 °C at the second step.

#### Substrate Scope and Limitation of Styrene Derivatives

The scope of styrene derivatives with a hydroxy moiety is shown in Table 3. In all cases, the nucleophilic attack of **2a** took place at a benzylic position exclusively and other regioisomers were not formed. The  $\beta$ -methylstyrene derivative **1b** gave the product **3ba**, and the reaction of the  $\alpha$ -methylstyrene derivative **1c** formed vicinal quaternary carbon atoms efficiently (entries 1 and 2). The reaction using the alkene **1d**, which is not a styrene derivative, provided no hydrofunctionalized product **3da** and complicated mixture was obtained (entry 3). The result suggested that the present reaction system was limited to styrene derivatives. The styrene derivatives **1e-1l**, which have different carbon frameworks, were also suitable to this reaction system. The reaction of  $\beta$ -(hydroxyalkyl)-substituted

styrenes **1e** and **1f** afforded the target products in modest yields (entries 4 and 5) while  $\alpha$ -(hydroxyalkyl)substituted substrates **1g** and **1h** gave higher yields (entries 6 and 7). A variety of  $\alpha$ -(hydroxyalkyl)substituted styrenes were applicable to this reaction system. The reaction of styrene derivatives bearing electron donating groups **1i** and **1j**, as well as electron withdrawing groups **1k** and **1l**, proceeded smoothly to give target products **3ia-3la** in moderate to high yields (entries 8-11). The trisubstituted alkene **1m** was suitable for this reaction system (entry 12). The intermolecular addition of silyl enolates to trisubstituted alkenes without electron-withdrawing group has not been reported and the use of disubstituted alkenes usually resulted in low yields due to their steric hindrance.<sup>4f,4g</sup> Therefore, it is advantage of this work that trisubstituted substrate was applicable and disubstituted alkenes gave moderate to high yield.

	OSiMe <sub>3</sub>				
	$R^1 R^3$			3	
		10 mol %) 2a (1.5 eo	quiv)	∼н	
	R <sup>2</sup> temper	ature, 1-120 h	" ×	OMe	
	<b>1b-1m</b> (1.0 equiv)		3ba-3ma <sub>C</sub>	)	
Entry	1	3	Temp (°C)	Yield (%) <sup>[c]</sup>	
1 <sup>[d]</sup>	ОН	ОН	80	52	
	1b	H 3ba			
2	OH Ic	OH H 3ca	80	53	
3	ОН	OH O OH O OMe	80	0	
4	ОН	3da CO <sub>2</sub> Me OH H 3ea	80	28	
5	OH If	CO <sub>2</sub> Me H 3fa	80	30	
6 <sup>[d]</sup>	HO	H 3ga	RT	59	

Table 3. Scope of hydrofunctionalization using various styrene derivatives.<sup>[a],[b]</sup>



[a] 1st step: **1** (0.5 mmol),  $InI_3$  (0.05 mmol),  $ClCH_2CH_2Cl$  (1 mL), RT-80 °C, 1-120 h (See Experimental Section for detail). 2nd step: **2a** (0.75 mmol), RT, 2 h. The reaction was quenched with  $Bu_4NF$  (1 M in THF). [b] Corresponding cyclic ethers were obtained without addition of **2a** (see Table S1 in Experimental Section for detail). [c] Isolated yield. [d] **2a** (3 equiv). [e] **2a** (4 equiv).

#### **Mechanistic Investigation**

Hydrofunctionalization with styrene derivative 1n, including a methoxyalkyl group, was attempted, but gave neither the target product 5 nor the cyclic ether 4 (Scheme 3-A). Moreover, the hydroalkoxylation of styrene with methanol as an external alcohol in the presence of a catalytic amount of InI<sub>3</sub> afforded no benzyl ether 6 (Scheme 3-B). These control experiments suggested that the intramolecular hydroxy group is necessary for this type of hydrofunctionalization. Taking into account the result of entry 13 in Table 1, the hydroalkoxylation of 1a can be catalyzed by a Brønsted acid, which can be generated via the alcoholysis of InI<sub>3</sub> with 1a *in situ*.<sup>11</sup> Thus, control experiments were carried out as shown in Scheme 4 to identify the active catalyst species for hydroalkoxylation. InI<sub>3</sub> or HI aq. efficiently catalyzed the hydroalkoxylation while the addition of K<sub>2</sub>CO<sub>3</sub>, or 2,6-di(*tert*-butyl)-4methylpyridine<sup>12</sup> as a proton scavenger, entirely prevented the  $InI_3$ -catalyzed hydroalkoxylation of **1a**. Therefore, these results indicated that *in situ*-generated Brønsted acid promoted the formation of the cyclic ether **4**.

(A) Hydrofunctionalization with styrene derivative with a methoxyalkyl group



(B) Hydroalkoxylation of styrene with methanol as an external alcohol



Scheme 3. Investigation of styrene derivatives without OH group.



Scheme 4. Hydroalkoxylation in the presence of a proton scavenger.

#### **Proposed Mechanism**

A proposed mechanism for the InI<sub>3</sub>-catalyzed hydrofunctionalization of **1a** with an organosilicon nucleophile (Me<sub>3</sub>Si*Nu*) is shown in Scheme 5. A trace amount of InI<sub>3</sub> undergoes alcoholysis with **1a** to give HI. The alkene moiety of **1a** is protonated and then alkoxylation occurs to afford cyclic ether **4**. The interaction between InI<sub>3</sub> and the ethereal oxygen of **4** cleaves the oxygen-carbon bond to generate carbocation **9**. A nucleophilic attack of the organosilicon compound to **9** occurs to produce the hydrofunctionalization product **10**. Transmetalation of the alkoxyindium moiety of **10** with Me<sub>3</sub>SiI regenerates the InI<sub>3</sub> catalyst. The indium trihalide catalyzed substitution reaction of ethers by an organosilicon nucleophile (Me<sub>3</sub>Si*Nu*) has been previously reported by our group.<sup>13</sup> Target alcohol product **3** is obtained after quenching by Bu<sub>4</sub>NF.<sup>14</sup> In our previous reports<sup>4f,4g</sup>, an indium trihalide coordinated to a carbon-carbon double bond and acted as a  $\pi$  Lewis acid to activate alkenes. Contrastively, in the present reaction, InI<sub>3</sub> mainly contributes to the generation of the intermediate

with nucleophiles.

A: Hydroalkoxylation of alkene moiety Inl<sub>3</sub> HOR OH Inl<sub>2</sub>OR OH HI нÌ 1a B: Substitution reaction of cyclic ether Me<sub>3</sub>SiO но l<sub>2</sub>InO Me<sub>3</sub>Sil Inl<sub>3</sub> quench Nu  $\ln l_3$ Me<sub>3</sub>Si Nu 11 10 3

Scheme 5. Proposed mechanism.

#### **Selective Functionalization of Alkenes**

We demonstrated the selective functionalization of substrate **10** with two types of alkene moieties (Scheme 6).<sup>15</sup> When **10** underwent our developed hydrofunctionalization using silyl ketene acetal **2a**, hydroalkylation took place selectively at the alkene moiety with a hydroxyalkyl group to afford product **120a** (Scheme 6-A). By contrast, when utilizing the BiBr<sub>3</sub>-mediated hydrofunctionalization conditions,<sup>4f</sup> polymerization of **10** just occurred. Moreover, the selective hydroazidation proceeded in the InI<sub>3</sub>-mediated reaction of **10** with silyl azide **2k** to give **12-N**<sub>3</sub> (Scheme 6-B). On the other hand, the reported iron-mediated hydroazidation<sup>16</sup> using NaN<sub>3</sub> occurred at both alkene moieties of **10** to give product **13**-**N**<sub>3</sub>. The other reported system using Brønsted acid catalyst and Me<sub>3</sub>SiN<sub>3</sub> as an azide source<sup>17</sup> was employed to afford no hydroazidation product but a hydroalkoxylation intermediate and starting material **10**. The selective hydrocyanation of **10** with silyl cyanide **2d** was also observed under InI<sub>3</sub>-catalyzed conditions to give **12od** exclusively, while the reported method<sup>4b</sup> resulted in polymerization of **10** (Scheme 6-C). These results showed the advantage of the InI<sub>3</sub>-catalyzed system for selective functionalization of the complex molecules.



Scheme 6. Selective hydrofunctionalization of the substrate 10 with two types of alkene moieties. Reaction conditions: [a] 1) 10 (0.5 mmol), InI<sub>3</sub> (10 mol %), ClCH<sub>2</sub>CH<sub>2</sub>Cl (10 mL), 5 °C, 17 h. 2) 2a (1.5 mmol), RT, 12 h. [b] 10 (0.2 mmol), BiBr<sub>3</sub> (0.3 mmol), 2a (0.3 mmol) and CH<sub>2</sub>Cl<sub>2</sub> (0.2 mL), RT, 8 h. [c] 1) 10 (0.5 mmol), InI<sub>3</sub> (10 mol %), ClCH<sub>2</sub>CH<sub>2</sub>Cl (10 mL), 5 °C, 5 h. 2) Me<sub>3</sub>SiN<sub>3</sub> (2k, 1.5 mmol), RT, 17 h. [d] 10 (0.5 mmol), NaN<sub>3</sub> (4.0 mmol), Fe<sub>2</sub>(ox)<sub>3</sub>·6H<sub>2</sub>O (2.5 mmol), NaBH<sub>4</sub> (3.2 mmol), THF (20 mL) and H<sub>2</sub>O (20 mL), 0 °C, 0.5 h. [e] 10 (0.5 mmol), TfOH (25 mol %), Me<sub>3</sub>SiN<sub>3</sub> (2k, 1.5 mmol), SiO<sub>2</sub> (1.25 g), CH<sub>2</sub>Cl<sub>2</sub> (2.5 mL), RT, 1 h. [f] 1) 10 (0.5 mmol), InI<sub>3</sub> (10 mol %), ClCH<sub>2</sub>CH<sub>2</sub>Cl (10 mL), 5 °C, 19 h. 2) Me<sub>3</sub>SiCN (2d, 2.5 mmol), RT, 35 h. [g] 10 (0.5 mmol), TfOH (2.2 mmol), Me<sub>3</sub>SiCN (2d, 2.1 mmol), PhCF<sub>3</sub> (4.5 mL), -5 °C, 1 h.

# 2-3. Conclusion

We have developed an InI<sub>3</sub> catalyzed hydrofunctionalization of alkenes with a hydroxyalkyl group using organosilicon nucleophiles such as silyl enolate, allylsilane, alkenylsilane, alkynylsilane, propargylsilane, silyl cyanide, silyl sulfide and silyl azide. Many functional groups that would be used for further elaboration were added to the alkene moiety in a single step. Various styrene derivatives were suitable for this reaction. It is noted that addition reaction of silyl enolate proceeded efficiently for not only monosubstituted alkenes but also multi-substituted ones compared with previously reported hydrofunctionalization. The selective functionalization of styrene derivatives with two alkene moieties was accomplished to show the preference of an alkene moiety with a hydroxyalkyl group. Mechanistic investigation showed that this hydrofunctionalization is composed of two steps: the hydroalkoxylation of an alkene catalyzed by a Brønsted acid and a sequential substitution reaction of cyclic ether catalyzed by InI<sub>3</sub>. This transformation of alkene to ether allowed us to synthesize alkenes with many types of functional groups, which cannot be introduced into alkenes in our previous works. Therefore, this work provides a novel method for functionalization of alkenes.

# 2-4. Experimental Section

#### General

New compounds were characterized by <sup>1</sup>H and <sup>13</sup>C NMR, <sup>13</sup>C off-resonance techniques, COSY, HMQC, HMBC, IR, MS, HRMS. <sup>1</sup>H (400 MHz) and <sup>13</sup>C NMR (100 MHz) spectra were obtained with TMS as internal standard. IR spectra were recorded as thin films. Bulb-to-bulb distillation (Kugelrohr) was accomplished at the oven temperature and pressure indicated. High-resolution mass spectra were obtained by magnetic sector type mass spectrometer. All reactions were carried out under nitrogen. All products were obtained as racemic mixtures.

#### Materials

Dehydrated ClCH<sub>2</sub>CH<sub>2</sub>Cl was purchased and used without further purification. Organosilicon nucleophiles **2a**, **2d**, **2e**, **2f**, **2h** and **2k** are commercially available. Styrene derivatives **1a**, **1d**, **1e**, **1f**, **1g**, **1h**, **1l** and organosilicon nucleophiles **2b**, **2c**, **2g**, **2i** and **2j** were synthesized by literature procedures and spectral data of these compounds were shown below. Styrene derivatives **1b**, **1c**, **1i**, **1j**, **1k**, **1m**, **1n** and **1o** are new compounds, and synthetic method and spectral data of these compounds were shown below. All metal salt and Me<sub>3</sub>SiBr are commercially available.

(1a) 2-(2-vinylphenyl)ethan-1-ol<sup>18</sup>



To a three necked flask, THF (56 mL) and 1-bromo-2-vinylbenzene (4.15 g, 20 mmol) were added. This solution was cooled at -78 °C and *n*-BuLi (1.6 M hexane solution, 14 mL, 22 mmol) was dropped over 15 minutes. After stirring for 1.5 h at -78 °C, to the reaction mixture was dropped ethylene oxide (1.1 M in THF, 25 mL, 30 mmol) and the reaction mixture was stirred for 0.5 h at -78 °C. The mixture was warm to room temperature and quenched by sat. NH<sub>4</sub>Cl aq. (25 mL). This solution was extracted with diethyl ether (20 mL x 3) and the collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 80:20, column length 11 cm, diameter 26 mm, spherical silica gel) to give the product (1.36 g, 45%). This is known compound and spectroscopic data were identical with those from literature.

(1b) 2-(2-(prop-1-en-1-yl)phenyl)ethan-1-ol



To a three necked flask, THF (56 mL) and 1-bromo-2-(prop-1-en-1-yl)benzene (4.15 g, 20 mmol) were added. This solution was cooled at -78 °C and n-BuLi (1.6 M hexane solution, 14 mL, 22 mmol) was dropped over 15 minutes. After stirring for 1.5 h at -78 °C, to the reaction mixture was dropped ethylene oxide (1.1 M in THF, 25 mL, 30 mmol) and the reaction mixture was stirred for 0.5 h at -78 °C. The mixture was warm to room temperature and quenched by sat. NH<sub>4</sub>Cl aq. (25 mL). This solution was extracted with diethyl ether (20 mL x 3) and the collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 80:20, column length 11 cm, diameter 26 mm, spherical silica gel) to give the product (E/Z = 99: 1, 1.67 g, 51%). IR: (neat) 3359 (OH) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.43 (d, J = 5.5 Hz, 1H, 5-H), 7.23-7.11 (m, 3H, 6-H, 7-H and 8-H), 6.65 (dq, J = 15.6, 1.4 Hz, 1H, 9-H), 6.11 (dq, J = 15.6, 6.9 Hz, 1H, 10-H), 3.80 (t, J = 6.9 Hz, 2H, 1-H<sub>2</sub>), 2.94 (t, J = 6.9 Hz, 2H, 2-H<sub>2</sub>), 1.90 (dd, J = 6.9, 1.4 Hz, 3H, 11-H<sub>3</sub>), 1.41 (s, 1H, OH); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 137.3 (s, C-3), 134.9 (s, C-4), 130.2 (d, C-8), 128.3 (d, C-9), 127.9 (d, C-10), 126.9 (d), 126.8 (d), 126.2 (d, C-5), 63.1 (t, C-1), 36.4 (t, C-2), 18.9 (q, C-11); MS: (CI, 70 eV) m/z 163 ([M + 1]<sup>+</sup>, 43); HRMS: (EI, 70 eV); Calculated (C<sub>11</sub>H<sub>14</sub>O) 162.1045 (M); Found: 162.1046.
(1c) 2-(2-(prop-1-en-2-yl)phenyl)ethan-1-ol



To a three necked flask, THF (56 mL) and 1-bromo-2-(prop-1-en-2-yl)benzene (3,49 g, 20 mmol) were added. This solution was cooled at -78 °C and n-BuLi (1.6 M hexane solution, 14 mL, 22 mmol) was dropped over 15 minutes. After stirring for 1.5 h at -78 °C, to the reaction mixture was dropped ethylene oxide (1.1 M in THF, 25 mL, 30 mmol) and the mixture was stirred for 0.5 h at -78 °C. The reaction mixture was warm to room temperature and quenched by sat. NH<sub>4</sub>Cl aq. (25 mL). This solution was extracted with diethyl ether (20 mL x 3) and the collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 80:20, column length 11 cm, diameter 26 mm, spherical silica gel) to give the product (1.53 g, 47%). IR: (neat) 3359 (OH) cm<sup>-1</sup>, 1639 (C=C) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.29-7.15 (m, 3H), 7.15-7.09 (m, 1H, 7-H), 5.23-5.17 (m, 1H, 10-H<sup>A</sup>), 4.89-4.82 (m, 1H, 10-H<sup>B</sup>), 3.81 (t, *J* = 7.2 Hz, 2H, 1-H<sub>2</sub>), 2.92 (t, *J* = 7.2 Hz, 2H, 2-H<sub>2</sub>), 2.05 (s, 3H, 11-H<sub>3</sub>), 1.39 (s, 1H, OH); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 145.4 (s, C-9), 144.3 (s, C-8), 134.6 (s, C-3), 129.6 (d, C-4), 128.4 (d, C-7), 126.9 (d), 126.3 (d), 115.2 (t, C-10), 63.7 (t, C-1), 36.1 (t, C-2), 25.3 (q, C-11); MS: (CI, 70 eV) *m/z* 163 ([M + 1]<sup>+</sup>, 100); HRMS: (EI, 70 eV) Calculated (C<sub>11</sub>H<sub>14</sub>O) 162.1045 (M) Found: 162.1043.

(1d) (2-allylphenyl)methanol<sup>19</sup>



To a three necked flask, (2-bromophenyl)methanol (9.45 g, 50 mmol), hexane (50 mL), Et2O (5 mL), 3,4-dihydro-2*H*-pyran (5.12 g, 60 mmol) and Amberlyst15 (6.3 g) were added. After stirring for 45 minutes, the reaction mixture was purified by column chromatography (hexane, column length 11 cm, diameter 26 mm, silica 75 g and Amberlyst15 6.2 g) to give 2-((2-bromobenzyl)oxy)tetrahydro-2*H*-pyran (7.2 g, 53%).

To a three necked flask, magnesium turning (0.73 g, 30 mmol) and THF (10 mL) were added. Three drops of a solution of 2-((2-bromobenzyl)oxy)tetrahydro-2H-pyran (7.2 g, 27 mmol) in THF (12 mL) was dropped into the vessel, and then ten drops of 1,2-dibromoethane was added to the reaction mixture. The reminding THF solution of 2-((2-bromobenzyl)oxy)tetrahydro-2H-pyran was dropped into the reaction mixture.

then cooled at 30 °C. A solution of allyl bromide (4.4 g, 36 mmol) in THF (10 mL) was dropped over 5 minutes. After stirring for 17 hours, the reaction mixture was quenched by sat. NH<sub>4</sub>Cl aq. (30 mL) under ice bath. The solution was extracted by ethyl acetate (30 mL x 4). The collected organic layer was dried (MgSO<sub>4</sub>) and the solvent was evaporated to give 2-((2-allylbenzyl)oxy)tetrahydro-2H-pyran (6.35 g, 100%).

To a three necked flask, 2-((2-allylbenzyl)oxy)tetrahydro-2H-pyran (6.27 g, 27 mmol), MeOH (38 mL) and 1N HCl aq. (38 mL) were added. After stirring for 20 hours at 50 °C, the reaction mixture was quenched by  $Et_2O$  (100 mL). The mixture was extracted by  $Et_2O$  (50 mL x 3) and the collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 60:40, column length 11 cm, diameter 26 mm, spherical silica gel) to give (2-allylphenyl)methanol as the target product (4.0 g, 100%). This is known compound and spectroscopic data were identical with those from literature.

(1e)(E)-5-phenylpent-4-en-1-ol<sup>20</sup>



To a three necked flask, THF (250 mL) and LiAlH<sub>4</sub> (5.1 g, 130 mmol) were added. This solution was cooled at 0 °C and 5-phenylpent-4-yn-1-ol (4.28 g, 26 mmol) was added to the mixture and the reaction mixture was stirred for 1 h at 0 °C. After refluxing for 2 days, to the reaction mixture was dropped sat. potassium sodium tartrate aq. (100 mL) at 0 °C. This solution was extracted with diethyl ether (100 mL x 3) and the collected organic layer was washed by sat. NaCl aq. (100 mL). The organic layer was dried (MgSO<sub>4</sub>) and the solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 80:20, column length 11 cm, diameter 26 mm, spherical silica gel) to give the product (3,24 g, 77%). This is known compound and spectroscopic data were identical with those from literature.

(1f) (*E*)-4-phenylbut-3-en-1-ol $^{21}$ 



To a three necked flask, THF (21 mL) and LiAlH<sub>4</sub> (1.78 g, 47 mmol) were added. This solution was cooled at 0 °C and 4-phenylbut-3-yn-1-ol (2.34 g, 16 mmol) in THF (21 mL) was dropped to the mixture

over 10 minutes. After refluxing for 15 h, the reaction mixture was quenched by  $Na_2SO_4$ ·10H<sub>2</sub>O at 0 °C. This mixture was filtered and the filtrate was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 80:20, column length 11 cm, diameter 26 mm, spherical silica gel) to give the product (2.21 g, 93%). This is known compound and spectroscopic data were identical with those from literature.

(1g) 5-phenylhex-5-en-1-ol<sup>22</sup>



To a two necked flask,  $Pd(PPh_3)_4$  (0.695 g, 0.6 mmol), 1,4-dioxane (40 mL), phenyl boronic acid (3.05 g, 24 mmol), 5-hexyne-1-ol (2.10 g, 20 mmol) and acetic acid (0.13 g, 2.0 mmol) were added. After stirring for 10 minutes at rt, the reaction mixture was stirred for 16 h at 80 °C. The reaction mixture was filtered and the solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 80:20, column length 11 cm, diameter 26 mm, spherical silica gel) to give the product (3.17 g, 90%). This is known compound and spectroscopic data were identical with those from literature.

(1h) 4-phenylpent-4-en-1-ol<sup>22</sup>



To a two necked flask,  $Pd(PPh_3)_4$  (0.692 g, 0.6 mmol), 1,4-dioxane (60 mL), phenyl boronic acid (3.04 g, 24 mmol), 4-pentyne-1-ol (1.65 g, 20 mmol) and acetic acid (0.22 mL, 4.0 mmol) were added. The reaction mixture was stirred for 16 h at 80 °C. The mixture was filtered and the solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 80:20, column length 11 cm, diameter 26 mm, spherical silica gel) to give the product (2.60 g, 80%). This is known compound and spectroscopic data were identical with those from literature.

(1i) 5-(p-tolyl)hex-5-en-1-ol



To a two necked flask, Pd(PPh<sub>3</sub>)<sub>4</sub> (0.343 g, 0.297 mmol), 1,4-dioxane (20 mL), *p*-tolyl boronic acid (1.70 g, 12.5 mmol), 5-hexyn-1-ol (0.981 g, 0.999 mmol) and acetic acid (0.101 g, 1.68 mmol) were added. After stirring for 24 h at 80 °C, the reaction mixture was filtered through a short pad of celite and the filtrate was evaporated. The residue was purified by column chromatography (hexane/ethyl acetate = 80:20, column length 11 cm, diameter 26 mm, spherical silica gel) to give the product (1.17 g, 61%). IR: (neat) 3444 (OH) cm<sup>-1</sup>, 1624 (C=C) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.29 (d, J = 8.2 Hz, 2H, o), 7.13 (d, J = 8.2 Hz, 2H, m), 5.24 (d, J = 1.4 Hz, 1H, 6-H<sup>A</sup>), 5.01 (d, J = 1.4 Hz, 1H, 6-H<sup>B</sup>), 3.62 (t, J = 6.3 Hz, 2H, 1-H<sub>2</sub>), 2.51 (t, J = 7.7 Hz, 2H, 4-H<sub>2</sub>), 2.34 (s, 3H, 7-H<sub>3</sub>), 1.67-1.54 (m, 2H, 2-H<sub>2</sub>), 1.54-1.41 (m, 2H, 3-H<sub>2</sub>), 1.67-1.41 (br, 1H, OH); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 148.0 (s, C-5), 138.1 (s, p), 137.0 (s, i), 128.9 (d, m), 125.9 (d, o), 111.6 (t, C-6), 62.7 (t, C-1), 35.0 (t, C-4), 32.3 (t, C-2), 24.3 (t, C-3), 21.0 (q, C-7); MS: (EI, 70 eV) m/z 190 (M, 28); HRMS: (EI, 70 eV) Calculated (C<sub>13</sub>H<sub>18</sub>O) 190.1358 (M) Found: 190.1355.

(1j) 5-(4-methoxyphenyl)hex-5-en-1-ol



To a two necked flask, Pd(PPh<sub>3</sub>)<sub>4</sub> (0.352 g, 0.305 mmol), 1,4-dioxane (20 mL), *p*-methoxy phenyl boronic acid (1.85 g, 12.2 mmol), 5-hexyn-1-ol (0.977 g, 9.95 mmol) and acetic acid (0.117 g, 1.95 mmol) were added. After stirring for 24 h at 80 °C, the reaction mixture was filtered through a short pad of celite and the filtrate was evaporated. The residue was purified by column chromatography (hexane/ethyl acetate = 80:20, column length 11 cm, diameter 26 mm, spherical silica gel) to give the product (0.993 g, 48%). IR: (neat) 3400 (OH) cm<sup>-1</sup>, 1608 (C=C) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.35 (d, J = 8.9 Hz, 2H, o), 6.86 (d, J = 8.9 Hz, 2H, m), 5.21 (s, 1H, 6-H<sup>A</sup>), 4.98 (s, 1H, 6-H<sup>B</sup>), 3.81 (s, 3H, OMe), 3.65 (t, J = 6.0 Hz, 2H, 1-H<sub>2</sub>), 2.51 (t, J = 7.5 Hz, 2H, 4-H<sub>2</sub>), 1.66-1.46 (m, 4H, 2-H<sub>2</sub> and 3-H<sub>2</sub>), 1.39 (s, 1H, OH); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 158.9 (s, p), 147.5 (s, C-5), 133.5 (s, i), 127.1 (d, o), 113.6 (d, m), 110.9 (t, C-6), 62.8 (t, C-1), 55.2 (q, OMe), 35.0 (t, C-4), 32.3 (t, C-3), 24.3 (t, C-2); MS: (EI, 70 eV) m/z 206 (M, 24); HRMS: (EI, 70 eV) Calculated (C<sub>13</sub>H<sub>18</sub>O<sub>2</sub>) 206.1307 (M) Found: 206.1308.

(1k) 4-(4-chlorophenyl)pent-4-en-1-ol



To a two necked flask, Pd(PPh<sub>3</sub>)<sub>4</sub> (0.352 g, 0.305 mmol), 1,4-dioxane (20 mL), *p*-chloro phenyl boronic acid (1.81 g, 11.6 mmol), 4-pentyn-1-ol (0.811 g, 9.64 mmol) and acetic acid (0.125 g, 2.08 mmol) were added. After stirring for 24 h at 80 °C, the reaction mixture was filtered through a short pad of celite and the filtrate was evaporated. The residue was purified by column chromatography (hexane/ethyl acetate = 80:20, column length 11 cm, diameter 26 mm, spherical silica gel) to give the product (0.715 g, 36%). IR: (neat) 3433 (OH) cm<sup>-1</sup>, 1716 (C=C) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.34 (d, J = 8.9 Hz, 2H), 7.29 (d, J = 8.9 Hz, 2H), 5.28 (d, J = 0.92 Hz, 1H, 5-H<sup>A</sup>), 5.10 (d, J = 0.92 Hz, 1H, 5-H<sup>B</sup>), 3.65 (t, J = 6.4 Hz, 2H, 1-H<sub>2</sub>), 2.57 (t, J = 7.8 Hz, 2H, 3-H<sub>2</sub>), 1.70 (tt, J = 7.8, 6.4 Hz, 2H, 2-H<sub>2</sub>), 1.29 (s, 1H, OH); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 146.8 (s, C-4), 139.4 (s, *i*), 133.2 (s, *p*), 128.4 (d), 127.4 (d), 113.0 (t, C-5), 62.2 (t, C-1), 31.4 (t, C-3), 30.9 (t, C-2); MS: (EI, 70 eV) *m/z* 196 (M, 2); HRMS: (EI, 70 eV) Calculated (C<sub>11</sub>H<sub>13</sub>ClO) 196.0655 (M) Found: 196.0654.

(11) 4-(4-fluorophenyl)pent-4-en-1-ol<sup>22</sup>



To a two necked flask,  $Pd(PPh_3)_4$  (0.328 g, 0.3 mmol), 1,4-dioxane (20 mL), *p*-fluorophenyl boronic acid (1.69 g, 12 mmol), 4-pentyne-1-ol (0.839 g, 10 mmol) and acetic acid (0.13 g, 2.0 mmol) were added. The reaction mixture was stirred for 24 h at 80 °C. The mixture was filtered and the solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 80:20, column length 11 cm, diameter 26 mm, spherical silica gel) to give the product (0.341 g, 19%). This is known compound and spectroscopic data were identical with those from literature.

(1m) 2-(2-(but-2-en-2-yl)phenyl)ethan-1-ol



To a three necked flask, THF (35 mL) and 1-bromo-2-(but-2-en-2-yl)benzene (2.55 g, 12 mmol) were added. This solution was cooled at -78 °C and n-BuLi (1.6 M hexane solution, 9 mL, 13 mmol) was dropped over 15 minutes. After stirring for 1.5 h at -78 °C, to the reaction mixture was dropped ethylene oxide (1.1 M in THF, 16 mL,18 mmol) and the mixture was stirred for 0.5 h at -78 °C. The reaction mixture was warm to room temperature and quenched by sat. NH<sub>4</sub>Cl aq. (15 mL). This solution was extracted with diethyl ether (20 mL x 3) and the collected organic layer was dried (MgSO<sub>4</sub>). The solvent

was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 80:20, column length 11 cm, diameter 26 mm, spherical silica gel) to give the product (0.368 g, 17%, ratio of stereoisomer = 85:15). IR: (neat) 3352 (OH) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) major stereoisomer 7.30-7.19 (m, 3H, 4-H, 5-H and 6-H), 7.03 (dd, J = 5.6, 3.6 Hz, 1H, 7-H), 5.58 (qq, J = 6.5, 1.7 Hz, 1H, 10-H), 3.80 (dd, J = 7.0, 12.8 Hz, 2H, 1-H<sub>2</sub>), 2.91-2.76 (m, 2H, 2-H<sub>2</sub>), 1.96 (m, 3H, 11-H<sub>3</sub>), 1.41-1.33 (br, 1H, OH), 1.37 (d, J = 6.5 Hz, 3H, 12-H<sub>3</sub>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) major stereoisomer 142.3 (s, C-8), 136.7 (s, C-9), 135.3 (s, C-3), 129.5 (d, C-4), 128.8 (d, C-7), 126.8 (d), 126.6 (d), 122.4 (d, C-10), 63.4 (t, C-1), 36.1 (t, C-2), 26.0 (q, C-11), 14.8 (q, C-12); MS: (CI, 70 eV) *m/z* 176 (M, 5); HRMS: (EI, 70 eV) Calculated (C<sub>12</sub>H<sub>16</sub>O) 176.1201 (M) Found: 176.1205.

(1n) 1-(2-methoxyethyl)-2-vinylbenzene



To a branched reactor vessel, NaOH aq. (50 wt%, 1g), benzene (2 mL), 2-(2-vinylphenyl)ethan-1-ol (0.737 g, 5 mmol), tributylamine (0.0492 g, 0.27 mmol) and dimethyl sulfate (0.962 g, 8 mmol) were added. After stirring for 3 h at 40 °C, the reaction mixture was quenched by diethyl ether (4 mL) and H<sub>2</sub>O (4 mL). This solution was extracted with diethyl ether (5 mL x 3), and washed by H<sub>2</sub>O (4 mL) and sat. NaCl aq. (5 mL). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane) column length 11 cm, diameter 26 mm, spherical silica gel) to give the product (0.649 g, 81%). IR: (neat) 1115 (OMe) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.49 (dd, J = 3.6, 5.3 Hz, 1H), 7.24-7.15 (m, 3H), 7.01 (dd, J = 17.4, 11.1 Hz, 1H, 9-H), 5.64 (dd, J = 17.4, 1.4 Hz, 1H, 10-H<sup>A</sup>), 5.30 (dd, J = 11.1, 1.4 Hz, 1H, 10-H<sup>B</sup>), 3.53 (t, J = 7.7 Hz, 2H, 1-H<sub>2</sub>), 3.34 (s, 3H, 11-H<sub>3</sub>), 2.97 (t, J = 7.7 Hz, 2H, 2-H<sub>2</sub>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 136.9 (s, C-8), 135.9 (s, C-3), 134.4 (d, C-9), 130.0 (d, C-4), 127.7 (d), 126.7 (d), 125.8 (d), 115.7 (t, C-10), 73.0 (t, C-1), 58.6 (q, C-11), 33.5 (t, C-2); MS: (CI, 70 eV) *m/z* 163 ([M + 1]<sup>+</sup>, 100), 131 (52); HRMS: (EI, 70 eV) Calculated (C<sub>11</sub>H<sub>14</sub>O) 162.1045 (M<sup>+</sup>) Found: 162.1046; Analysis: C<sub>11</sub>H<sub>14</sub>O (162.10) Calculated: C, 81.44; H, 8.70 Found: C, 81.16; H, 8.52.

(10) 2-(2-(prop-1-en-2-yl)phenyl)ethan-1-ol



To a two necked flask, Pd(PPh<sub>3</sub>)<sub>4</sub> (0.420 g, 0.363 mmol), 1,4-dioxane (20 mL), *p*-vinyl boronic acid (2.20 g, 14.9 mmol), 4-pentyn-1-ol (1.05 g, 12.5 mmol) and acetic acid (0.13 mL, 2.40 mmol) were added. After stirring for 29 h at 80 °C, the reaction mixture was filtered through a short pad of celite and the filtrate was evaporated. The residue was purified by column chromatography (hexane/ethyl acetate = 80:20, column length 11 cm, diameter 26 mm, spherical silica gel) to give the product (1.77 g, 75%). IR: (neat) 3359 (OH) cm<sup>-1</sup>, 1628 (C=C) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.42-7.36 (m, 4H), 6.71 (dd, J = 18.4, 10.6 Hz, 1H, 6-H), 5.76 (d, J = 18.4 Hz, 1H, 7-H<sup>A</sup>), 5.33 (s, 1H, 5-H<sup>A</sup>), 5.25 (d, J = 10.6 Hz, 1H, 7-H<sup>B</sup>), 5.10 (s, 1H, 5-H<sup>B</sup>), 3.66 (t, J = 6.3 Hz, 2H, 1-H<sub>2</sub>), 2.61 (t, J = 7.7 Hz, 2H, 3-H<sub>2</sub>), 1.73 (tt, J = 6.3, 7.7 Hz, 2H, 2-H<sub>2</sub>), 1.51 (br, 1H, OH); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 147.4 (s, C-4), 140.2 (s, *i*), 136.7 (s, *p*), 136.3 (d, C-6), 126.2 (d), 126.1 (d), 113.7 (t, C-7), 112.5 (t, C-5), 62.4 (t, C-1), 31.4 (t, C-3), 31.1 (t, C-2); MS: (CI, 70 eV) *m/z* 189 ([M + 1]<sup>+</sup>, 100); HRMS: (EI, 70 eV) Calculated (C<sub>13</sub>H<sub>16</sub>O) 188.1201 (M) Found: 188.1199.

(2b) (cyclohexylidene(methoxy)methoxy)trimethylsilane<sup>4g</sup>



To a solution of diisopropylamine (11.5 g, 110 mmol) in THF (80 mL) was added *n*-BuLi (63 mL, 1.6 M in hexane) at 0 °C. The resulted solution was stirred for 10 min at room temperature. Then, a solution of methyl cyclohexanecarboxyrate (11.4 g, 80 mmol) in THF (30 mL) was slowly added to the LDA solution at -78 °C. After the solution was stirred for 2 h at -78 °C, trimethylsilylchloride (10.8 g, 100 mmol) was added to the reaction mixture, and the reaction mixture was stirred for 20 h at room temperature. The reaction mixture was poured into ice water and hexane, which was extracted with hexane. The organic phase was washed with brine, dried (MgSO<sub>4</sub>) and concentrated. The obtained crude oil was purified by distillation (bp. 100-102 °C, 20 mmHg) to give the desired product (14.3 g, 83%). This is known compound and spectroscopic data were identical with those from literature.

(2c) 3-trimetylsilyloxy-2-pentene<sup>23</sup>

To a solution of diisopropylamine (10.1 g, 100 mmol) in THF (100 mL) was added *n*-BuLi (1.6 M in hexane, 63 mL, 100 mmol) at 0 °C. After stirring at 0 °C for 15 min, then to reaction mixture was added 3-pentanone (6.89 g, 80 mmol) dropwise at -78 °C. After stirred for 30 min at -78 °C, Me<sub>3</sub>SiCl (10.9 g, 100 mmol) was added dropwise at -78 °C. The reaction mixture was allowed to reach room temperature and stirred for 30 min. Then, sat. NaHCO<sub>3</sub> aq. (50 mL) was added and the solution was extracted with pentane (50 mL x 3). The organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by distillation under reduced pressure (76 °C, 85 mmHg) to give the product (12.0 g, 76% yield, E/Z = 5:1). This is known compound and spectroscopic data were identical with those from literature.

#### (2g) (E)-trimethyl(styryl)silane<sup>24,25</sup>



To a three necked flask, trimethyl(phenylethynyl)silane (3.63 g, 20.8 mmol) and dry hexane (5 mL) were added. To the reaction mixture was dropped DIBAL-H (1.0 M in hexane, 24 mL, 24 mmol). After stirring for 13 h at rt, the reaction mixture was quenched by 4 N HCl aq. (17 mL) at 0 °C. The mixture was extracted by  $Et_2O$  (20 mL x 3) and the collected organic layer was washed with sat. NaCl aq. (60 mL x 3). The organic layer was dried (MgSO<sub>4</sub>) and the solvent was evaporated and the residue was purified by distillation under reduced pressure (112 °C, 23 mmHg) to give the product (2.23 g, 60%). This is known compound and spectroscopic data were identical with those from literature.

(2i) trimethyl(prop-2-yn-1-yl)silane<sup>26</sup>

To a three necked flask, Mg (1.82 g, 75 mmol), Et<sub>2</sub>O (16 mL) and HgCl<sub>2</sub> (0.255 g, 0.94 mmol) were added. After vigolously stirring for 30 minutes at rt, propargyl bromide (1.1 M in Et<sub>2</sub>O, 2 mL, 2.2 mmol) was dropped to the mixture. The reaction mixture was cooled into 0 °C and propargyl bromide (1.1 M in Et<sub>2</sub>O, 31 mL, 34.1 mmol) was dropped to the reaction mixture. After stirring for 30 minutes at rt, the reaction mixture was cooled into 0 °C and Me<sub>3</sub>SiCl (3.34 g, 30.7 mmol) was dropped to the reaction mixture. After vigolously stirring for 9 h at rt, the reaction mixture was filtered through a celite pad and

the filtrate was purified by column chromatography (Et<sub>2</sub>O, column length 11 cm, diameter 26 mm, spherical silica gel) to give the product (51wt% in Et<sub>2</sub>O, 2.91 g, 36\%). This is known compound and spectroscopic data were identical with those from literature.

(2j) trimethyl(phenylthio)silane<sup>27,28</sup>

To a three necked flask, Et<sub>2</sub>O (32 mL) and benzenethiol (3.60 g, 33 mmol) were added. This solution was cooled at -78 °C and *n*-BuLi (1.6 M hexane solution, 52 mL, 82.5 mmol) was dropped to the mixture over 15 minutes, followed by Me<sub>3</sub>SiCl was dropped over 4 minutes. After stirring for 2 hours, the reaction mixture was quenched by dry hexane (18 mL). The solvent was evaporated and the residue was filtered and the residue was washed by hexane. The solvent was evaporated and the residue was purified by distillation under reduced pressure (110 °C, 20 mmHg) to give the product (4.82 g, 81%). This is known compound and spectroscopic data were identical with those from literature.

## General procedure for InI<sub>3</sub> catalyzed hydrofunctionalization of styrene derivatives 1 using organosilicon nucleophiles 2 (Table 1-3, Scheme 3A and Scheme 6)

To a solution of acid catalyst (10 mol %) in 1,2-dichloroethane (1 mL) was added styrene derivatives 1 (0.5 mmol). The mixture was stirred at 80 °C for 24 h and then cooled to rt. Organosilicon nucleophiles 2 (0.75 mmol) was added to the reaction mixture. After stirring at RT for 2 h, the mixture was quenched by diethyl ether (2 mL),  $Bu_4NF$  (1 M in THF, 1 mL) and sat.  $Na_2CO_3$  aq. (2 mL) and then extracted with diethyl ether (2 mL x 3). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (column length 11 cm, diameter 26 mm, spherical silica gel).

# Experimental procedure for InI<sub>3</sub> catalyzed hydroalkoxylation of styrene derivative 1a (Entry 7 in Table 1)

To a solution of  $InI_3$  (0.0250 g, 0.0504 mmol) in 1,2-dichloroethane (1 mL) was added 2-(2-vinylphenyl)ethan-1-ol **1a** (0.0777 g, 0.524 mmol). After stirring at 80 °C for 24 h, the mixture was quenched by diethyl ether (2 mL) and sat.  $Na_2CO_3$  aq. (2 mL), and then extracted with diethyl ether (2 mL x 3). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the yield of cyclic ether **4** was determined by <sup>1</sup>H-NMR.

# Experimental procedure for InI<sub>3</sub> catalyzed hydroalkoxylation of styrene with methanol (Scheme 3-B)

To a solution of  $InI_3$  (0.0241 g, 0.050 mmol) in 1,2-dichloroethane (1.0 mL) and methanol (20 µL, 0.50 mmol) was added styrene (56 µL, 0.50 mmol). After stirring at 80 °C for 24 h, the mixture was quenched by diethyl ether (2 mL) and sat. Na<sub>2</sub>CO<sub>3</sub> aq. (2 mL), and then extracted with diethyl ether (2 mL x 3). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the yield of benzyl ether **6** was determined by <sup>1</sup>H-NMR.

# Experimental procedure for InI<sub>3</sub> catalyzed hydroalkoxylation of styrene derivative 1a in the presence of proton scavenger (Scheme 4)

To a solution of  $InI_3$  (0.05 mmol) and proton scavenger (0.15 mmol) in 1,2-dichloroethane (1 mL) was added 2-(2-vinylphenyl)ethan-1-ol **1a** (0.50 mmol). After stirring at 80 °C for 24 h, the mixture was quenched by diethyl ether (2 mL) and sat. Na<sub>2</sub>CO<sub>3</sub> aq. (2 mL), and then extracted with diethyl ether (2 mL x 3). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the yield of the cyclic ether **4** was determined by <sup>1</sup>H-NMR.

### Hydrocyanation of styrene derivative with reported method



Scheme S1. Hydrocyanation of styrene derivative 1a with reported method.<sup>4b</sup>

### Hydroalkoxylation of various types of styrene derivaives

Table S1. Hydroalkoxylation of various types of styrene derivatives.<sup>[a]</sup>





[a] **1** (0.5 mmol), catalyst (0.05 mmol), ClCH<sub>2</sub>CH<sub>2</sub>Cl (1 mL), RT -80 °C, 7-96 h. [b] Yields were determined by NMR analysis.

#### **Product date**

Methyl 3-[2-(2-hydroxyethyl)phenyl]-2,2-dimethylbutanoate (3aa)



From InI<sub>3</sub> (0.0254 g, 0.0513 mmol), 2-(2-vinylphenyl)ethan-1-ol (0.0740 g, 0.499 mmol), and dimethylketene methyl trimethylsilylacetal (0.141 g, 0.809 mmol) following general procedure, **3aa** (0.089 g, 71%) was obtained after column chromatography (hexane/ethyl acetate = 70:30). IR: (neat) 3440 (OH) cm<sup>-1</sup>, 1728 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.25-7.10 (m, 4H, 4-H, 5-H, 6-H and 7-H), 3.87-3.69 (m, 2H, 1-H<sub>2</sub>), 3.63 (s, 3H, OMe), 3.54 (q, J = 7.2 Hz, 1H, 9-H), 3.27 (ddd, J = 13.5, 7.2, 6.8 Hz, 1H, 2-H<sup>A</sup>), 2.79 (ddd, J = 13.5, 7.7, 7.2 Hz, 1H, 2-H<sup>B</sup>), 1.47 (s, 1H, OH), 1.23 (d, J = 7.2 Hz, 3H, 10-H<sub>3</sub>), 1.18 (s, 3H, 11-Me<sup>A</sup>), 1.11 (s, 3H, 11-Me<sup>B</sup>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 178.3 (s, C-12), 141.5 (s, C-8), 136.8 (s, C-3), 130.1 (d, C-4), 128.0 (d), 126.3 (d), 126.2 (d), 63.6 (t, C-1), 51.7 (q, C-13), 47.0 (s, C-11), 39.9 (d, C-9), 36.6 (t, C-2), 24.0 (q, 11-Me<sup>B</sup>), 21.2 (q, 11-Me<sup>A</sup>), 17.5 (q, C-10); MS: (CI, 70 eV) *m/z* 251 ([M + 1]<sup>+</sup>, 100); HRMS: (EI, 70 eV) Calculated (C<sub>15</sub>H<sub>22</sub>O<sub>3</sub>) 250.1569 (M<sup>+</sup>) Found: 250.1571.

Methyl 1-(1-(2-(2-hydroxyethyl)phenyl)ethyl)cyclohexane-1-carboxylate (3ab)



From InI<sub>3</sub> (0.0495 g, 0.10 mmol), 2-(2-vinylphenyl)ethan-1-ol (0.0755 g, 0.509 mmol), and (cyclohexylidene(methoxy)methoxy)trimethylsilane (0.327 g, 1.53 mmol) following general procedure (rt, 3 h at 2<sup>nd</sup> step), **3ab** (0.109 g, 74%) was obtained after column chromatography (hexane/ethyl acetate = 70:30). IR: (neat) 3440 (OH) cm<sup>-1</sup>, 1716 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.19-7.11 (m, 4H, Ar), 3.78-3.67 (m, 2H, 1-H<sub>2</sub>), 3.60 (s, 3H, OMe), 3.32-3.19 (m, 2H, 9-H and 2-H<sup>A</sup>), 2.75 (dt, J = 14.0, 6.8 Hz, 1H, 2-H<sup>B</sup>), 2.34-2.28 (br, 1H), 2.21 (s, 1H, OH), 2.08-2.00 (br, 1H), 1.65-1.51 (br, 3H), 1.33-1.21 (m, 2H), 1.24 (d, J = 7.2 Hz, 3H, 10-H<sub>3</sub>), 1.21-0.95 (m, 3H); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 176.4 (s, C-17), 141.6 (s, C-8), 136.6 (s, C-3), 129.9 (d, C-4), 128.2 (d, C-7), 126.2 (d), 125.9 (d), 63.5 (t, C-1), 52.2 (s, C-11), 51.2 (q, OMe), 41.9 (d, C-9), 36.6 (t, C-2), 32.7 (t), 30.7 (t), 25.5 (t), 23.8 (t), 23.6 (t),

17.0 (q, C-10); MS: (EI, 70 eV) *m*/*z* 290 (M, 1); HRMS: (CI, 70 eV) Calculated (C<sub>18</sub>H<sub>27</sub>O<sub>3</sub>) 291.1960 ([M + 1]<sup>+</sup>) Found: 291.1957.

5-[2-(2-Hydroxyethyl)phenyl]-4-methylhexan-3-one (3ac)



From InI<sub>3</sub> (0.0226 g, 0.0456 mmol), 2-(2-vinylphenyl)ethan-1-ol (0.0765 g, 0.516 mmol), and (*Z*)-trimethyl(pent-2-en-3-yloxy)silane (0.216 g, 1.36 mmol) following general procedure (rt, 5 h at 2<sup>nd</sup> step), **3ac** (0.0837 g, 69%, d.r. = 54:46) was obtained after column chromatography (hexane/ethyl acetate = 80:20). IR: (neat) 3433 (OH) cm<sup>-1</sup>, 1709 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) diastereo mixture: 7.26-7.07 (m, 4H), 3.97-3.88 (m, 0.46H), 3.88-3.75 (m, 1.54H), 3.39-3.25 (m, 1H), 3.07 (dt, *J* = 13.5, 6.8 Hz, 0.46H), 3.02-2.78 (m, 2.54H), 2.61 (dq, J = 18.4, 7.2 Hz, 0.54H), 2.53-2.32 (m, 1H), 2.03 (dq, *J* = 17.8, 7.2 Hz, 0.46H), 1.82 (s, 0.54H), 1.20 (t, *J* = 6.8 Hz, 3H), 1.11 (t, *J* = 6.8 Hz, 1.62H), 1.10 (t, *J* = 7.2 Hz, 1.38H), 0.806 (t, *J* = 7.2 Hz, 1.62H), 0.788 (t, *J* = 6.8 Hz, 1.38H); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) diastereo mixture: 215.9 (s), 215.7 (s), 144.7 (s), 143.4 (s), 136.2 (s), 136.1 (s), 130.6 (d), 130.4 (d), 127.1 (d), 126.7 (d), 126.1 (d), 126.0 (d), 125.5 (d), 63.8 (t), 63.4 (t), 52.7 (d), 52.0 (d), 36.6 (d), 36.5 (d), 36.4 (t), 36.3 (t), 36.2 (t), 36.0 (t), 21.4 (q), 19.8 (q), 16.2 (q), 15.6 (q), 7.64 (q), 7.41 (q); MS: (CI, 70 eV) m/z 235 ([M + 1]<sup>+</sup>, 100); HRMS: (CI, 70 eV) Calculated (C<sub>15</sub>H<sub>23</sub>O<sub>2</sub>) 235.1698 ([M + 1]<sup>+</sup>) Found: 235.1698.

5-[2-(2-Hydroxyethyl)phenyl]-4-methylhexan-3-one (3ad)



From InI<sub>3</sub> (0.0565 g, 0.114 mmol, 20 mol %), 2-(2-vinylphenyl)ethan-1-ol (0.0765 g, 0.516 mmol), and trimethylsilyl cyanide (0.3 mL, 2.5 mmol) following general procedure (70 °C, 14 h at 2<sup>nd</sup> step), **3ad** (0.0284 g, 31%) was obtained after column chromatography (hexane/ethyl acetate = 80:20). IR: (neat) 3429 (OH) cm<sup>-1</sup>, 2241 (CN) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.54-7.47 (m, 1H), 7.34-7.20 (m, 3H, 4-H, 5-H and 6-H), 4.24 (q, J = 6.8 Hz, 1H, 9-H), 3.92-3.81 (m, 2H, 1-H<sub>2</sub>), 2.89 (t, J = 6.8 Hz, 2H,

2-H<sub>2</sub>), 1.71 (s, 1H, OH), 1.62 (d, *J* = 6.8 Hz, 3H, 10-H<sub>3</sub>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 136.0 (s), 135.6 (s), 130.5 (d, C-4), 128.3 (d), 127.6 (d), 127.3 (d), 122.2 (s, C-11), 63.2 (t, C-1), 35.2 (t, C-2), 27.7 (d, C-9), 21.0 (d, C-10); MS: (CI, 70 eV) *m*/*z* 176 ([M + 1]<sup>+</sup>, 25), 158 (M – OH, 100); HRMS: (EI, 70 eV) Calculated (C<sub>11</sub>H<sub>13</sub>NO) 175.0997 (M<sup>+</sup>) Found: 175.0998.

2-(2-(4-Methylpent-4-en-2-yl)phenyl)ethan-1-ol (3ae)



From InI<sub>3</sub> (0.0244 g, 0.0492 mmol), 2-(2-vinylphenyl)ethan-1-ol (0.0744 g, 0.502 mmol) and trimethyl(2-methylallyl)silane (0.115 g, 0.896 mmol) following general procedure (rt, 4 h at 2<sup>nd</sup> step), **3ae** (0.056 g, 55%) was obtained after column chromatography (hexane/ethyl acetate = 80:20). IR: (neat) 3410 (OH) cm<sup>-1</sup>, 1647 (C=C) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.27 (d, J = 7.2 Hz, 1H, 7-H), 7.22 (t, J = 7.2 Hz, 1H), 7.17 (d, J = 7.2 Hz, 1H, 4-H), 7.13 (t, J = 7.2 Hz, 1H), 4.74 (s, 1H, 13-H<sup>A</sup>), 4.68 (s, 1H, 13-H<sup>B</sup>), 3.83 (t, J = 6.8 Hz, 2H, 1-H<sub>2</sub>), 3.27-3.16 (m, 1H, 9-H), 3.03-2.88 (m, 2H, 2-H<sub>2</sub>), 2.32 (dd, J = 13.5, 6.3 Hz, 11-H<sup>A</sup>), 2.23 (dd, J = 13.5, 8.7 Hz, 11-H<sup>B</sup>), 1.70 (s, 3H, 14-H<sub>3</sub>), 1.50 (s, 1H, OH), 1.20 (d, J = 6.8 Hz, 3H, 10-H<sub>3</sub>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 146.0 (s, C-8), 144.0 (s, C-12), 135.0 (s, C-3), 130.1 (d, C-4), 126.9 (d), 126.1 (d, C-7), 125.8 (d), 112.2 (t, C-13), 63.5 (t, C-1), 46.6 (t, C-11), 35.9 (t, C-2), 32.1 (d, C-9), 22.4 (q, C-14), 21.6 (q, C-10); MS: (EI, 70 eV) *m/z* 204 (M, 2); HRMS: (CI, 70 eV) Calculated (C<sub>14</sub>H<sub>21</sub>O) 205.1592 ([M + 1]<sup>+</sup>) Found: 205.1589.

2-[2-(Pent-4-en-2-yl)phenyl]ethan-1-ol (3af)



From InI<sub>3</sub> (0.0303 g, 0.0611 mmol), 2-(2-vinylphenyl)ethan-1-ol (0.0745 g, 0.503 mmol), allyltrimethylsilane (0.0834 g, 0.730 mmol) and Me<sub>3</sub>SiBr (0.0097 g, 0.0634 mmol) following general procedure (rt, 3 h at 2<sup>nd</sup> step), **3af** (0.082 g, 86%) was obtained after column chromatography (hexane/ethyl acetate = 80:20). IR: (neat) 3344 (OH) cm<sup>-1</sup>, 1666 (C=C) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.29-7.10 (m, 4H, 4-H, 5-H, 6-H and 7-H), 5.77-5.64 (m, 1H, 12-H), 5.01 (d, J = 14.0 Hz, 1H, 13-H<sup>A</sup>), 4.95 (d, J = 10.1 Hz, 1H, 13-H<sup>B</sup>), 3.80 (t, J = 7.2 Hz, 2H, 1-H<sub>2</sub>), 3.09 (sextet, J = 7.0 Hz, 1H, 9-

H), 3.00-2.85 (m, 2H, 2-H<sub>2</sub>), 2.43-2.24 (m, 2H, 11-H<sub>2</sub>), 1.59 (s, 1H, OH), 1.23 (d, J = 7.0 Hz, 3H, 10-H<sub>3</sub>) <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 145.4 (s, C-8), 137.0 (d, C-12), 135.1 (s, C-3), 130.1 (d, C-4), 126.9 (d), 126.0 (d), 125.8 (d), 116.1 (t, C-13), 63.5 (t, C-1), 42.4 (t, C-11), 36.0 (t, C-2), 34.1 (d, C-9), 21.8 (q, C-10); MS: (CI, 70 eV) m/z 191 ([M + 1]<sup>+</sup>, 30), 173 (M - OH, 100), 149 (M - CH<sub>2</sub>CHCH<sub>2</sub>, 35); HRMS: (EI, 70 eV) Calculated (C<sub>13</sub>H<sub>18</sub>O) 190.1358 (M<sup>+</sup>) Found: 190.1357.

(*E*)-2-[2-(4-Phenylbut-3-en-2-yl)phenyl]ethan-1-ol (**3ag**)



From InI<sub>3</sub> (0.0229 g, 0.0462 mmol), 2-(2-vinylphenyl)ethan-1-ol (0.0743 g, 0.501 mmol), (*E*)-trimethyl(styryl)silane (0.2687 g, 1.52 mmol) and Me<sub>3</sub>SiBr (0.016 g, 0.105 mmol) following general procedure (rt, 10 h at 2<sup>nd</sup> step), **3ag** (0.0742 g, 59%) was obtained after column chromatography (hexane/ethyl acetate = 80:20). IR: (neat) 3402 (OH) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.35-7.14 (m, 9H, Ar), 6.37 (dd, J = 4.3, 15.9 Hz, 1H, 11-H), 6.35 (d, J = 15.9 Hz, 1H, 12-H), 3.94 (dq, J = 4.3, 6.8 Hz, 1H, 9-H), 3.85 (t, J = 6.8 Hz, 2H, 1-H<sub>2</sub>), 2.98 (t, J = 6.8 Hz, 2H, 2-H<sub>2</sub>), 1.58 (s, 1H, OH), 1.45 (d, J = 6.8 Hz, 3H, 10-H<sub>3</sub>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 143.9 (s, C-8), 137.4 (s, *i*), 135.4 (s, C-3), 135.2 (d, C-11), 130.2 (d, C-4), 128.5 (d), 127.2 (d), 127.04 (d), 127.01, 126.2 (d), 126.1 (d), 63.5 (t, C-1), 37.4 (d, C-9), 35.9 (t, C-2), 21.4 (q, C-10); MS: (CI, 70 eV) *m/z* 253 ([M + 1]<sup>+</sup>, 5), 235 (M – OH, 100); HRMS: (EI, 70 eV) Calculated (C<sub>18</sub>H<sub>20</sub>O) 252.1514 (M<sup>+</sup>) Found: 252.1514.

2-(2-(4-Phenylbut-3-yn-2-yl)phenyl)ethan-1-ol (3ah)



From InI<sub>3</sub> (0.0257 g, 0.0519 mmol), 2-(2-vinylphenyl)ethan-1-ol (0.0751 g, 0.507 mmol), trimethyl(phenylethynyl)silane (0.270 g, 1.55 mmol) and Me<sub>3</sub>SiBr (0.0143 g, 0.0934 mmol) following general procedure (rt, 1 h at 2<sup>nd</sup> step), **3ah** (0.0684 g, 54%) was obtained after column chromatography (hexane/ethyl acetate = 80:20). IR: (neat) 3301 (OH) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.64 (d, J = 7.2 Hz, 1H, Ar), 7.43-7.38 (m, 2H, Ar), 7.30-7.22 (m, 4H, Ar), 7.22-7.17 (m, 2H, Ar), 4.24 (q, J = 6.8 Hz,

1H, 9-H), 3.88 (t, *J* = 7.2 Hz, 2H, 1-H<sub>2</sub>), 3.04-2.92 (m, 2H, 2-H<sub>2</sub>), 1.72 (s, 1H, OH), 1.56 (d, *J* = 6.8 Hz, 3H, 10-H<sub>3</sub>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 141.8 (s, C-8), 134.9 (s, C-3), 131.5 (d), 130.1 (d, C-4), 128.2 (d), 127.7 (d), 127.5 (d), 127.2 (d), 126.9 (d), 123.6 (s, *i*), 93.1 (s, C-11), 81.7 (s, C-12), 63.4 (t, C-1), 35.6 (t, C-2), 28.5 (d, C-9), 23.8 (q, C-10); MS: (CI, 70 eV) *m/z* 251 ([M + 1]<sup>+</sup>, 20), 233 (M - OH, 100); HRMS: (CI, 70 eV) Calculated (C<sub>18</sub>H<sub>19</sub>O) 251.1436 ([M + 1]<sup>+</sup>) Found: 251.1436.

2-[2-(Penta-3,4-dien-2-yl)phenyl]ethan-1-ol (3ai)



From InI<sub>3</sub> (0.0274 g, 0.0553 mmol), 2-(2-vinylphenyl)ethan-1-ol (0.0745 g, 0.503 mmol), trimethyl(prop-2-yn-1-yl)silane (51 wt % in Et<sub>2</sub>O, 0.1653 g, 1.50 mmol) and Me<sub>3</sub>SiBr (0.0170 g, 0.111 mmol) following general procedure (35 °C, 2 h at 2<sup>nd</sup> step), **3ai** (0.07 g, 68%) was obtained after column chromatography (hexane/ethyl acetate = 70:30). IR: (neat) 3394 (OH) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.31 (d, J = 7.7 Hz, 1H, 7-H), 7.25-7.21 (m, 1H), 7.20-7.14 (m, 2H), 5.30 (dt, J = 6.8, 6.5 Hz, 1H, 11-H), 4.83-4.74 (m, 2H, 13-H<sub>2</sub>), 3.85 (dt, J = 6.3, 6.8 Hz, 2H, 1-H<sub>2</sub>), 3.82-3.74 (m, 1H, 9-H), 2.972 (t, J = 6.8 Hz, 1H, 2-H<sup>A</sup>), 2.966 (t, J = 6.8 Hz, 1H, 2-H<sup>B</sup>), 1.41 (t, J = 6.3 Hz, 1H, OH), 1.34 (d, J = 7.0 Hz, 3H, 10-H<sub>3</sub>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 207.8 (s, C-12), 144.3 (s, C-8), 135.2 (s, C-3), 130.1 (d, C-4), 127.1 (d), 127.0 (d), 126.2 (d), 96.0 (d, C-11), 76.7 (t, C-13), 63.5 (t, C-1), 35.9 (t, C-2), 33.6 (d, C-9), 21.5 (q, C-10); MS: (CI, 70 eV) *m*/*z* 189 ([M + 1]<sup>+</sup>, 10); HRMS: (EI, 70 eV) Calculated (C<sub>13</sub>H<sub>16</sub>O) 188.1201 (M<sup>+</sup>) Found: 188.1199.

2-[2-(1-(Phenylthio)ethyl)phenyl]ethan-1-ol (3aj)



From InI<sub>3</sub> (0.0245 g, 0.0494 mmol), 2-(2-vinylphenyl)ethan-1-ol (0.0739 g, 0.499 mmol) and trimethyl(phenylthio)silane (0.278 g, 1.52 mmol) following general procedure (rt, 13 h at 2<sup>nd</sup> step), **3aj** (0.0758 g, 59%) was obtained after column chromatography (hexane/ethyl acetate = 70:30). IR: (neat) 3394 (OH) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.51 (d, J = 7.7 Hz, 1H, 7-H), 7.42-7.36 (m, 2H), 7.32-

7.15 (m, 6H), 4.66 (q, 1H, 9-H), 3.88-3.73 (m, 2H, 1-H<sub>2</sub>), 3.03-2.82 (m, 2H, 2-H<sub>2</sub>), 1.77 (s, 1H, OH), 1.63 (d, J = 7.2 Hz, 3H, 10-H<sub>3</sub>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 141.1 (s, C-8), 135.7 (s), 134.7 (s), 133.2 (d), 130.1 (d, C-4), 128.8 (d), 127.6 (d), 127.2 (d), 127.1 (d), 127.0 (d), 63.4 (t, C-1), 43.3 (d, C-9), 35.7 (t, C-2), 22.3 (q, C-10); MS: (CI, 70 eV) m/z 259 ([M + 1]<sup>+</sup>, 1), 149 (M – SPh, 100); HRMS: (EI, 70 eV) Calculated (C<sub>16</sub>H<sub>18</sub>OS) 258.1078 (M<sup>+</sup>) Found: 258.1081.

2-[2-(1-Azidoethyl)phenyl]ethan-1-ol (3ak)



From InI<sub>3</sub> (0.0288 g, 0.0581 mmol), 2-(2-vinylphenyl)ethan-1-ol (0.0776 g, 0.524 mmol) and azidotrimethylsilane (0.2 mL, 1.5 mmol) following general procedure (50 °C, 4 h at 2<sup>nd</sup> step), **3ak** (0.0612 g, 54%) was obtained after column chromatography (hexane/ethyl acetate = 80:20). IR: (neat) 3437 (OH) cm<sup>-1</sup>, 2102 (N=N=N) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.35 (d, J = 7.2 Hz, 1H, 7-H), 7.26-7.12 (m, 3H, 4-H, 5-H and 6-H), 4.89 (q, J = 6.8 Hz, 1H, 9-H), 3.77 (t, J = 6.8 Hz, 2H, 1-H<sub>2</sub>), 2.87 (t, J = 6.8 Hz, 2H, 2-H<sub>2</sub>), 1.65 (s, 1H, OH), 1.47 (d, J = 6.8 Hz, 3H, 10-H<sub>3</sub>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 139.2 (s, C-8), 135.7 (s, C-3), 130.3 (d, C-4), 128.1 (d), 127.2 (d), 126.1 (d, C-7), 63.5 (t, C-1), 56.7 (d, C-9), 35.4 (t, C-2), 21.1 (q, C-10); MS: (CI, 70 eV) *m/z* 164 ([M + 1]<sup>+</sup> – N<sub>2</sub>, 30); HRMS: (CI, 70 eV) Calculated (C<sub>10</sub>H<sub>14</sub>N<sub>3</sub>O) 192.1137 ([M + 1]<sup>+</sup>) Found: 192.1134.

Methyl 3-[2-(2-hydroxyethyl)phenyl)]-2,2-dimethylpentanoate (3ba)



From InI<sub>3</sub> (0.0286 g, 0.0577 mmol), (*E*)-2-(2-(prop-1-en-1-yl)phenyl)ethan-1-ol (0.0799 g, 0.493 mmol) and dimethylketene methyl trimethylsilylacetal (0.2819 g, 1.62 mmol) following general procedure (80 °C, 15 h at 1<sup>st</sup> step), **3ba** (0.068 g, 52%) was obtained after column chromatography (hexane/ethyl acetate = 70:30). IR: (neat) 3452 (OH) cm<sup>-1</sup>, 1728 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.24-7.14 (m, 4H, 4-H, 5-H, 6-H and 7-H), 3.93-3.93 (m, 1H, 1-H<sup>A</sup>), 3.83-3.73 (m, 1H, 1-H<sup>B</sup>), 3.65 (s, 3H, 14-H<sub>3</sub>),

3.34 (dd, J = 11.8, 3.4 Hz, 1H, 9-H), 3.25 (dt, J = 14.0, 7.2 Hz, 1H, 2-H<sup>A</sup>), 2.83 (ddd, J = 14.0, 7.7, 7.5 Hz, 1H, 2-H<sup>B</sup>), 1.84-1.73 (m, 1H, 10-H<sup>A</sup>), 1.66-1.58 (m, 1H, 10-H<sup>B</sup>), 1.49-1.41 (m, 1H, OH), 1.16 (s, 3H, 12-Me<sup>A</sup>), 1.03 (s, 3H, 12-Me<sup>B</sup>), 0.677 (t, J = 7.2 Hz, 3H, 11-H<sub>3</sub>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 178.6 (s, C-13), 138.7 (s, C-8), 138.4 (s, C-3), 130.0 (d), 127.7 (d), 126.3 (d), 126.1 (d), 63.5 (t, C-1), 51.8 (q, C-14), 47.3 (s, C-12), 47.2 (d, C-9), 36.3 (t, C-2), 24.5 (q, 12-Me<sup>B</sup>), 24.1 (t, C-10), 20.7 (q, 12-Me<sup>A</sup>), 12.8 (q, C-11); MS: (CI, 70 eV) *m*/*z* 265 ([M + 1]<sup>+</sup>, 100); HRMS: (CI, 70 eV) Calculated (C<sub>16</sub>H<sub>25</sub>O<sub>3</sub>) 265.1804 ([M + 1]<sup>+</sup>) Found: 265.1803.

Methyl 3-[2-(2-hydroxyethyl)phenyl)]-2,2-dimethylpentanoate (3ca)



From InI<sub>3</sub> (0.0264 g, 0.0533 mmol), 2-(2-(prop-1-en-2-yl)phenyl)ethan-1-ol (0.0822 g, 0.507 mmol) and dimethylketene methyl trimethylsilylacetal (0.135 g, 0.774 mmol) following general procedure (80 °C, 6 h at 1<sup>st</sup> step), **3ca** (0.1047 g, 72%) was obtained after column chromatography (hexane/ethyl acetate = 70:30). IR: (neat) 3425 (OH) cm<sup>-1</sup>, 1720 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.39-7.30 (m, 1H, 7-H), 7.21-7.08 (m, 3H, 4-H, 5-H and 6-H), 3.87 (t, J = 7.2 Hz, 2H, 1-H<sub>2</sub>), 3.56 (s, 3H, OMe), 3.11 (t, J = 7.2 Hz, 2H, 2-H<sub>2</sub>), 1.59 (s, 4H, 9-Me and OH), 1.16 (s, 3H, 10-Me); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 177.3 (s, C-11), 144.1 (s, C-8), 137.3 (s, C-3), 132.4 (d, C-4), 130.9 (d, C-7), 126.3 (d), 125.4 (d), 64.8 (t, C-1), 51.4 (q, OMe), 49.9 (s, C-10), 45.4 (s, C-9), 39.6 (t, C-2), 28.2 (q, 9-Me), 22.5 (q, 10-Me); MS: (EI, 70 eV) *m*/*z* 232 (M - MeOH, 5); HRMS: (CI, 70 eV) Calculated (C<sub>16</sub>H<sub>25</sub>O<sub>3</sub>) 265.1804 ([M + 1]<sup>+</sup>) Found: 265.1801.

Methyl 7-hydroxy-2,2-dimethyl-3-phenylheptanoate (3ea)



From InI<sub>3</sub> (0.0278 g, 0.0561 mmol), (*E*)-5-phenylpent-4-en-1-ol (0.0864 g, 0.532 mmol) and dimethylketene methyl trimethylsilylacetal (0.149 g, 0.855 mmol) following general procedure (80 °C, 65 h at 1<sup>st</sup> step), **3ea** (0.040 g, 28%) was obtained after column chromatography (hexane/ethyl acetate =

80:20). IR: (neat) 3444 (OH) cm<sup>-1</sup>, 1728 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.25 (t, 2H, *m*), 7.22 (d, 1H, *p*), 7.15 (d, 2H, *o*), 3.64 (s, 3H, OMe), 3.51 (t, J = 6.8 Hz, 2H, 1-H<sub>2</sub>), 2.96 (dd, J = 12.6, 3.4 Hz, 1H, 5-H), 1.89-1.77 (m, 1H, 4-H<sup>A</sup>), 1.59-1.39 (m, 3H, 4-H<sup>B</sup> and 2-H<sub>2</sub>), 1.34 (s, 1H, OH), 1.18-1.05 (m, 2H, 3-H<sub>2</sub>), 1.14 (s, 3H, 6-Me<sup>A</sup>), 1.01 (s, 3H, 6-Me<sup>B</sup>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 178.3 (s, C-7), 140.1 (s, *i*), 129.7 (d, *m*), 127.8 (d, *o*), 126.6 (d, *p*), 62.7 (t, C-1), 52.8 (d, C-5), 51.7 (q, OMe), 46.7 (s, C-6), 32.5 (t, C-2), 29.7 (t, C-4), 24.5 (q, 6-Me<sup>A</sup>), 24.2 (t, C-3), 20.7 (q, 6-Me<sup>B</sup>); MS: (CI, 70 eV) *m/z* 265 ([M + 1]<sup>+</sup>, 74); HRMS: (CI, 70 eV) Calculated (C<sub>16</sub>H<sub>25</sub>O<sub>3</sub>) 265.1804 ([M + 1]<sup>+</sup>) Found: 265.1805.

Methyl 6-hydroxy-2,2-dimethyl-3-phenylhexanoate (3fa)



From InI<sub>3</sub> (0.0338 g, 0.068 mmol), (*E*)-4-phenylbut-3-en-1-ol (0.0762 g, 0.514 mmol) and dimethylketene methyl trimethylsilylacetal (0.120 g, 0.688 mmol) following general procedure (80 °C, 120 h at 1<sup>st</sup> step), **3fa** (0.038 g, 30%) was obtained after column chromatography (hexane/ethyl acetate = 80:20). IR: (neat) 3437 (OH) cm<sup>-1</sup>, 1728 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.38 (t, J = 7.2 Hz, 2H, *m*), 7.23 (t, J = 7.2 Hz, 1H, *p*), 7.16 (d, J = 7.2 Hz, 2H, *o*), 3.64 (s, 3H, 7-H<sub>3</sub>), 3.55 (t, J = 6.8 Hz, 2H, 1-H<sub>2</sub>), 2.99 (dd, J = 12.6, 2.9 Hz, 1H, 4-H), 1.92-1.79 (m, 1H, 3-H<sup>A</sup>), 1.66-1.47 (m, 2H, 3-H<sup>B</sup> and OH), 1.37-1.24 (m, 2H, 2-H<sub>2</sub>), 1.16 (s, 3H, 5-Me<sup>A</sup>), 1.02 (s, 3H, 5-Me<sup>B</sup>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 178.4 (s, C-6), 139.9 (s, *i*), 129.7 (d, *o*), 127.9 (d, *m*), 126.7 (d, *p*), 62.5 (t, C-1), 52.4 (d, C-4), 51.7 (q, C-7), 46.7 (s, C-5), 31.1 (t, C-2), 26.2 (t, C-3), 24.6 (q, 5-Me<sup>B</sup>), 20.6 (q, 5-Me<sup>A</sup>); MS: (CI, 70 eV) *m/z* 251 ([M + 1]<sup>+</sup>, 100); HRMS: (CI, 70 eV) Calculated (C<sub>15</sub>H<sub>23</sub>O<sub>3</sub>) 251.1647 ([M + 1]<sup>+</sup>) Found: 251.1649.

Methyl 7-hydroxy-2,2,3-trimethyl-3-phenylheptanoate (3ga)



From InI<sub>3</sub> (0.0253 g, 0.0511 mmol), 2-(2-vinylphenyl)ethan-1-ol (0.0911 g, 0.615 mmol) and dimethylketene methyl trimethylsilylacetal (0.125 g, 0.717 mmol) following general procedure (rt, 3 h at 1<sup>st</sup> step), **3ga** (0.084 g, 59%) was obtained after column chromatography (hexane/ethyl acetate =

70:30). IR: (neat) 3444 (OH) cm<sup>-1</sup>, 1720 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.32-7.15 (m, 5H, Ar), 3.55 (t, J = 7.0 Hz, 2H, 1-H<sub>2</sub>), 3.50 (s, 3H, OMe), 2.24 (td, J = 13.0, 4.8 Hz, 1H, 4-H<sup>A</sup>), 1.77 (s, 1H, OH), 1.63 (td, J = 13.0, 4.8 Hz, 1H, 4-H<sup>B</sup>), 1.54 (tt, J = 7.7, 7.0 Hz, 2H, 2-H<sub>2</sub>), 1.39 (s, 3H, 6-H<sub>3</sub>), 1.28-1.15 (m, 1H, 3-H<sup>A</sup>), 1.12 (s, 3H, 7-Me<sup>A</sup>), 1.02 (s, 3H, 7-Me<sup>B</sup>), 0.993-0.836 (m, 1H, 3-H<sup>B</sup>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 177.2 (s, C-8), 143.0 (s, *i*), 128.2 (d), 127.3 (d), 125.9 (d, *p*), 62.7 (t, C-1), 51.2 (q, OMe), 49.7 (s, C-7), 45.9 (s, C-5), 34.8 (t, C-4), 33.5 (t, C-2), 22.1 (q, 7-Me<sup>A</sup>), 21.7 (q, 7-Me<sup>B</sup>), 21.2 (q, C-6), 20.8 (t, C-3); MS: (CI, 70 eV) *m*/*z* 279 ([M + 1]<sup>+</sup>, 1), 247 (M - OMe, 3); HRMS: (CI, 70 eV) Calculated (C<sub>17</sub>H<sub>27</sub>O<sub>3</sub>) 279.1960 ([M + 1]<sup>+</sup>) Found: 279.1958.

Methyl 6-hydroxy-2,2,3-trimethyl-3-phenylhexanoate (3ha)



From InI<sub>3</sub> (0.0213 g, 0.0430 mmol), 4-phenylpent-4-en-1-ol (0.0862 g, 0.531 mmol) and dimethylketene methyl trimethylsilylacetal (0.139 g, 0.797 mmol) following general procedure (rt, 8 h at 1<sup>st</sup> step), **3ha** (0.092 g, 66%) was obtained after column chromatography (hexane/ethyl acetate = 70:30). IR: (neat) 3448 (OH) cm<sup>-1</sup>, 1720 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.32-7.22 (m, 4H, *o* and *m*), 7.22-7.16 (m, 1H, *p*), 3.59 (td, J = 6.8, 1.9 Hz, 2H, 1-H<sub>2</sub>), 3.50 (s, 3H, OMe), 2.31 (ddd, J = 14.0, 13.0, 3.38 Hz, 1H, 3-H<sup>A</sup>), 1.73-1.59 (m, 2H, 3-H<sup>B</sup> and OH), 1.48-1.34 (m, 1H, 2-H<sup>A</sup>), 1.40 (s, 3H, 5-H<sub>3</sub>) 1.22-1.09 (m, 1H, 2-H<sup>B</sup>), 1.13 (s, 3H, 6-Me<sup>A</sup>), 1.03 (s, 3H, 6-Me<sup>B</sup>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 177.2 (s, C-7), 142.8 (s, *i*), 128.2 (d), 127.3 (d), 126.0 (d, *p*), 63.4 (t, C-1), 51.2 (q, OMe), 49.7 (s, C-6), 45.6 (s, C-4), 31.0 (t, C-3), 28.0 (t, C-2), 22.1 (q, 6-Me<sup>B</sup>), 21.7 (q, 6-Me<sup>A</sup>), 21.3 (q, C-5); MS: (CI, 70 eV) *m/z* 265 ([M + 1]<sup>+</sup>, 100); HRMS: (EI, 70 eV) Calculated (C<sub>16</sub>H<sub>24</sub>O<sub>3</sub>) 264.1725 (M<sup>+</sup>) Found: 264.1724.

Methyl 7-hydroxy-2,2,3-trimethyl-3-(p-tolyl)heptanoate (3ia)



From InI<sub>3</sub> (0.0253 g, 0.0511 mmol), 5-phenylhex-5-en-1-ol (0.0908 g, 0.477 mmol) and dimethylketene methyl trimethylsilylacetal (0.123 g, 0.706 mmol) following general procedure (rt, 4 h at 1<sup>st</sup> step), **3ia** 

(0.097 g, 70%) was obtained after column chromatography (hexane/ethyl acetate = 70:30). IR: (neat) 3444 (OH) cm<sup>-1</sup>, 1720 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.11 (d, J = 8.7 Hz, 2H), 7.08 (d, J = 8.7 Hz, 2H), 3.56 (t, J = 6.8 Hz, 2H, 1-H<sub>2</sub>), 3.52 (s, 3H, 9-H<sub>3</sub>), 2.31 (s, 3H, *p*-Me), 2.22 (td, J = 13.0, 3.9 Hz, 1H, 4-H<sup>A</sup>), 1.61 (td, J = 12.6, 4.8 Hz, 1H, 4-H<sup>B</sup>), 1.54 (tt, J = 6.8, 7.3 Hz, 2H, 2-H<sub>2</sub>), 1.37 (s, 3H, 6-H<sub>3</sub>), 1.30 (s, 1H, OH), 1.27-1.15 (m, 1H, 3-H<sup>A</sup>), 1.10 (s, 3H, 7-Me<sup>A</sup>), 1.01 (s, 3H, 7-Me<sup>B</sup>), 0.99-0.87 (m, 1H, 3-H<sup>B</sup>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 177.3 (s, C-8), 139.9 (s, *i*), 135.4 (s, *p*), 128.1 (d), 128.0 (d), 62.8 (t, C-1), 51.2 (q, C-9), 49.7 (s, C-7), 45.6 (s, C-5), 34.8 (t, C-4), 33.6 (t, C-2), 22.2 (q, 7-Me<sup>A</sup>), 21.8 (q, 7-Me<sup>B</sup>), 21.3 (q, C-6), 20.9 (t, C-3), 20.8 (q, *p*-Me); MS: (CI, 70 eV) *m/z* 293 ([M + 1]<sup>+</sup>, 24); HRMS: (CI, 70 eV) Calculated (C<sub>18</sub>H<sub>29</sub>O<sub>3</sub>) 293.2117 ([M + 1]<sup>+</sup>) Found: 293.2114.

Methyl 7-hydroxy-3-(4-methoxyphenyl)-2,2,3-trimethylheptanoate (3ja)



From InI<sub>3</sub> (0.0266 g, 0.0537 mmol), 5-(4-methoxyphenyl)hex-5-en-1-ol (0.1075 g, 0.521 mmol) and dimethylketene methyl trimethylsilylacetal (0.121 g, 0.692 mmol) following general procedure (rt, 1 h at 1<sup>st</sup> step), **3ja** (0.092 g, 57%) was obtained after column chromatography (hexane/ethyl acetate = 60:40). IR: (neat) 3440 (OH) cm<sup>-1</sup>, 1720 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.14 (d, J = 8.7 Hz, 2H, o), 6.81 (d, J = 8.7 Hz, 2H, m), 3.79 (s, 3H, 10-H<sub>3</sub>), 3.55 (t, J = 7.2 Hz, 2H, 1-H<sub>2</sub>), 3.52 (s, 3H, 9-H<sub>3</sub>), 2.19 (ddd, J = 14.5, 13.5, 3.4 Hz, 1H, 4-H<sup>A</sup>), 1.66-1.50 (m, 3H, 4-H<sup>B</sup> and 2-H<sub>2</sub>), 1.36 (s, 3H, 6-H<sub>3</sub>), 1.27-1.13 (m, 2H, 3-H<sup>A</sup> and OH), 1.10 (s, 3H, 7-Me<sup>A</sup>), 1.01 (s, 3H, 7-Me<sup>B</sup>), 0.98-0.87 (m, 1H, 3-H<sup>B</sup>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 177.3 (s, C-8), 157.5 (s, p), 134.9 (s, i), 129.2 (d, o), 112.5 (d, m), 62.7 (t, C-1), 55.0 (q, C-10), 51.2 (q, C-9), 49.7 (s, C-7), 45.3 (s, C-5), 34.9 (t, C-4), 33.5 (t, C-2), 22.0 (q, 7-Me<sup>B</sup>), 21.7 (q, 7-Me<sup>A</sup>), 21.3 (q, C-6), 20.8 (t, C-3); MS: (EI, 70 eV) m/z 277 (M - OMe, 0.5), 207 (M - Me<sub>2</sub>CHCO<sub>2</sub>Me, 100); HRMS: (CI, 70 eV) Calculated (C<sub>18</sub>H<sub>29</sub>O<sub>4</sub>) 309.2066 ([M + 1]<sup>+</sup>) Found: 309.2064.

Methyl 3-(4-chlorophenyl)-6-hydroxy-2,2,3-trimethylhexanoate (3ka)



From InI<sub>3</sub> (0.0267 g, 0.0539 mmol), 4-(4-chlorophenyl)pent-4-en-1-ol (0.1014 g, 0.481 mmol) and dimethylketene methyl trimethylsilylacetal (0.137 g, 0.786 mmol) following general procedure (rt, 12 h at 1<sup>st</sup> step), **3ka** (0.064 g, 45%) was obtained after column chromatography (hexane/ethyl acetate = 70:30). IR: (neat) 3444 (OH) cm<sup>-1</sup>, 1724 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.25 (d, J = 8.7 Hz, 2H, m), 7.17 (d, J = 8.7 Hz, 2H, o), 3.64-3.56 (m, 2H, 1-H<sub>2</sub>), 3.52 (s, 3H, 8-H<sub>3</sub>), 2.27 (ddd, J = 14.5, 13.0, 3.4 Hz, 1H, 3-H<sup>A</sup>), 1.54 (ddd, J = 13.5, 13.0, 5.3 Hz, 1H, 3-H<sup>B</sup>), 1.45-1.34 (m, 1H, 2-H<sup>A</sup>), 1.38 (s, 3H, 5-H<sub>3</sub>), 1.32 (s, 1H, OH), 1.19-1.06 (m, 1H, 2-H<sup>B</sup>), 1.12 (s, 3H, 6-Me<sup>A</sup>), 1.04 (s, 3H, 6-Me<sup>B</sup>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 177.0 (s, C-7), 141.5 (s, *i*), 132.0 (s, *p*), 129.7 (d, *o*), 127.5 (d, *m*), 63.4 (t, C-1), 51.3 (q, C-8), 49.7 (s, C-6), 45.5 (s, C-4), 31.1 (t, C-3), 28.0 (t, C-2), 22.1 (q, 6-Me<sup>B</sup>), 21.8 (q, 6-Me<sup>A</sup>), 21.3 (q, C-5); MS: (CI, 70 eV) *m/z* 301 ([M + 1]<sup>+</sup> + 2, 34), 299 ([M + 1]<sup>+</sup>, 100); HRMS: (CI, 70 eV) Calculated (C<sub>16</sub>H<sub>24</sub>ClO<sub>3</sub>) 299.1414 ([M + 1]<sup>+</sup>) Found: 299.1410.

Methyl 3-(4-fluorophenyl)-6-hydroxy-2,2,3-trimethylhexanoate (3la)



From InI<sub>3</sub> (0.0229 g, 0.046 mmol), 4-(4-fluorophenyl)pent-4-en-1-ol (0.0905 g, 0.481 mmol) and dimethylketene methyl trimethylsilylacetal (0.139 g, 0.797 mmol) following general procedure (rt, 22 h at 1<sup>st</sup> step), **3la** (0.101 g, 77%) was obtained after column chromatography (hexane/ethyl acetate = 70:30). IR: (neat) 3456 (OH) cm<sup>-1</sup>, 1720 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.21 (dd,  $J^{d}_{HF}$  = 5.3 Hz,  $J^{3}_{HH}$  = 8.7 Hz, 2H, *o*), 6.98 (dd,  $J^{3}_{HF}$  = 9.2 Hz,  $J^{3}_{HH}$  = 8.7 Hz, 2H, *m*), 3.65-3.53 (m, 2H, 1-H<sub>2</sub>), 3.51 (s, 3H, 8-H<sub>3</sub>), 2.27 (ddd, *J* = 14.5, 13.5, 3.4 Hz, 1H, 3-H<sup>A</sup>), 1.67 (ddd, *J* = 13.5, 13.5, 5.3 Hz, 1H, 3-H<sup>B</sup>), 1.52 (s, 1H, OH), 1.46-1.34 (m, 1H, 2-H<sup>A</sup>), 1.39 (s, 3H, 5-H<sub>3</sub>), 1.22-1.13 (m, 1H, 2-H<sup>B</sup>), 1.12 (s, 3H, 6-Me<sup>A</sup>), 1.03 (s, 3H, 6-Me<sup>B</sup>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 177.1 (s, C-7), 161.2 (d, *J* = 255 Hz, *p*), 138.6 (d, *J* = 3.3 Hz, *i*), 129.8 (d, *J* = 7.4 Hz, *o*), 114.0 (d, *J* = 20.5 Hz, *m*), 63.3 (t, C-1), 51.3 (q, C-8), 49.8 (s, C-6), 45.3 (s, C-4), 31.1 (t, C-3), 27.9 (t, C-2), 22.0 (q, 6-Me<sup>B</sup>), 21.7 (q, 6-Me<sup>A</sup>), 21.4 (q, C-5); MS: (CI, 70 eV) *m*/*z* 283 ([M + 1]<sup>+</sup>, 100); HRMS: (CI, 70 eV) Calculated (C<sub>16</sub>H<sub>24</sub>FO<sub>3</sub>) 283.1709 ([M + 1]<sup>+</sup>) Found: 283.1708.

Methyl 3-(2-(2-hydroxyethyl)phenyl)-2,2,3-trimethylpentanoate (3ma)



From InI<sub>3</sub> (0.0298 g, 0.0601 mmol), 2-(2-(but-2-en-2-yl)phenyl)ethan-1-ol (0.0482 g, 0.273 mmol) and dimethylketene methyl trimethylsilylacetal (0.203 g, 1.16 mmol) following general procedure (70 °C, 4 h at 1<sup>st</sup> step), **3ma** (0.043 g, 56%) was obtained after column chromatography (hexane/ethyl acetate = 50:50). IR: (neat) 3417 (OH) cm<sup>-1</sup>, 1716 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.23 (dd, J = 7.3, 1.8 Hz, 1H, 7-H), 7.19 (dd, J = 6.9, 2.3 Hz, 1H, 4-H), 7.17-7.10 (m, 2H, 5-H and 6-H), 3.94-3.83 (m, 2H, 1-H<sub>2</sub>), 3.55 (s, 3H, OMe), 3.28 (dt, J = 13.7, 6.0 Hz, 1H, 2-H<sup>A</sup>), 2.91 (dt, J = 13.7, 6.8 Hz, 1H, 2-H<sup>B</sup>), 2.51 (dq, J = 14.7, 7.3 Hz, 1H, 10-H<sup>A</sup>), 1.60 (s, 3H, 9-Me), 1.60-1.51 (m, 1H, 10-H<sup>B</sup>), 1.25 (s, 1H, OH), 1.15 (s, 3H, 12-Me<sup>A</sup>), 1.13 (s, 3H, 12-Me<sup>B</sup>), 0.719 (t, J = 7.3 Hz, 3H, 11-H<sub>3</sub>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 177.4 (s, C-13), 140.6 (s, C-8), 138.6 (s, C-3), 132.2 (d, C-4), 131.6 (d, C-7), 126.2 (d), 125.1 (d), 64.6 (t, C-1), 51.4 (q, OMe), 50.8 (s, C-12), 49.2 (s, C-9), 39.1 (t, C-2), 31.3 (t, C-10), 24.4 (q, 9-Me), 22.5 (q, 12-Me<sup>A</sup>), 22.2 (q, 12-Me<sup>B</sup>), 9.3 (q, C-11); MS: (EI, 70 eV) *m/z* 249 (M – CH<sub>2</sub>CH<sub>3</sub>, 12); HRMS: (CI, 70 eV) Calculated (C<sub>17</sub>H<sub>27</sub>O<sub>3</sub>) 279.1960 ([M + 1]<sup>+</sup>) Found: 279.1958.

Methyl 6-hydroxy-2,2,3-trimethyl-3-(4-vinylphenyl)hexanoate (12oa)



From InI<sub>3</sub> (0.0251 g, 0.0507 mmol) in 1,2-dichloroethane (10 mL), 4-(4-vinylphenyl)pent-4-en-1-ol (0.080 g, 0.425 mmol) and dimethylketene methyl trimethylsilylacetal (0.302 g, 1.73 mmol) following general procedure (5 °C, 17 h at 1<sup>st</sup> step and rt, 12 h at 2<sup>nd</sup> step), **120a** (0.0568 g, 46%) was obtained after column chromatography (hexane/ethyl acetate = 70:30). IR: (neat) 3487 (OH) cm<sup>-1</sup>, 1716 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.33 (d, J = 8.6 Hz, 2H, m), 7.20 (d, J = 8.6 Hz, 2H, o), 6.69 (dd, J = 17.7, 10.9 Hz, 1H, 8-H), 5.73 (d, J = 17.7 Hz, 1H, 9-H<sup>A</sup>), 5.22 (d, J = 10.9 Hz, 1H, 9-H<sup>B</sup>), 3.62-3.55 (m, 2H, 1-H<sub>2</sub>), 3.52 (s, 3H, OMe), 2.30 (ddd, J = 14.0, 12.7, 3.2 Hz, 1H, 3-H<sup>A</sup>), 1.76-1.60 (br, 1H, OH), 1.66 (ddd, J = 13.1, 12.7, 5.0 Hz, 1H, 3-H<sup>B</sup>), 1.45-1.35 (m, 1H, 2-H<sup>A</sup>), 1.39 (s, 3H, 5-H<sub>3</sub>), 1.21-1.10 (m,

1H, 2-H<sup>B</sup>), 1.13 (s, 3H, 6-Me<sup>A</sup>), 1.04 (s, 3H, 6-Me<sup>B</sup>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 177.2 (s, C-7), 142.6 (s, *i*), 136.3 (d, C-8), 135.1 (s, *p*), 128.4 (d, *o*), 125.1 (d, *m*), 113.4 (t, C-9), 63.4 (t, C-1), 51.2 (q, OMe), 49.7 (s, C-6), 45.6 (s, C-4), 31.0 (t, C-3), 28.0 (t, C-2), 22.1 (q, 6-Me<sup>B</sup>), 21.7 (q, 6-Me<sup>A</sup>), 21.2 (q, C-5); MS: (EI, 70 eV) *m*/*z* 290 (M, 2); HRMS: (CI, 70 eV) Calculated (C<sub>18</sub>H<sub>27</sub>O<sub>3</sub>) 291.1960 ([M + 1]<sup>+</sup>) Found: 291.1958.

4-Azido-4-(4-vinylphenyl)pentan-1-ol (12-N<sub>3</sub>)



From InI<sub>3</sub> (0.0261 g, 0.0527 mmol) in 1,2-dichloroethane (10 mL), 4-(4-vinylphenyl)pent-4-en-1-ol (0.0964 g, 0.512 mmol) and trimethylsilyl azide (0.167 g, 1.45 mmol) following general procedure (5 °C, 5 h at 1<sup>st</sup> step and rt, 17 h at 2<sup>nd</sup> step), **12-N**<sub>3</sub> (0.0228 g, 20%) was obtained after column chromatography (hexane/ethyl acetate = 70:30). IR: (neat) 3398 (OH) cm<sup>-1</sup>, 2106 (N<sub>3</sub>) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.41 (d, J = 8.7 Hz, 2H), 7.35 (d, J = 8.7 Hz, 2H), 6.71 (dd, J = 17.9, 10.6 Hz, 1H, 6-H), 5.76 (d, J = 17.9 Hz, 1H, 7-H<sup>A</sup>), 5.26 (d, J = 10.6 Hz, 1H, 7-H<sup>B</sup>), 3.53 (t, J = 6.3 Hz, 2H, 1-H<sub>2</sub>), 1.96-1.85 (m, 2H, 3-H<sub>2</sub>), 1.69 (s, 3H, 5-H<sub>3</sub>), 1.57-1.32 (m, 3H, 2-H<sub>2</sub> and OH); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 142.8 (s, *i*), 136.6 (s, *p*), 136.0 (d, C-6), 126.3 (d), 125.7 (d), 114.2 (t, C-7), 66.6 (s, C-4), 62.6 (t, C-1), 38.5 (t, C-3), 27.5 (t, C-2), 25.7 (q, C-5); MS: (EI, 70 eV) *m/z* 171 (M – N<sub>3</sub> – H<sub>2</sub>O, 69); HRMS: (CI, 70 eV) Calculated (C<sub>13</sub>H<sub>18</sub>N<sub>3</sub>O) 232.1450 ([M + 1]<sup>+</sup>) Found: 232.1455.

5-Hydroxy-2-methyl-2-(4-vinylphenyl)pentanenitrile (12od)



From InI<sub>3</sub> (0.0256 g, 0.0517 mmol) in 1,2-dichloroethane (10 mL), 4-(4-vinylphenyl)pent-4-en-1-ol (0.0949 g, 0.504 mmol) and trimethylsilyl cyanide (0.3 mL, 2.5 mmol) following general procedure (5 °C, 19 h at 1<sup>st</sup> step and rt, 35 h at 2<sup>nd</sup> step), **12od** (0.055 g, 50%) was obtained after column chromatography (hexane/ethyl acetate = 70:30). IR: (neat) 3433 (OH) cm<sup>-1</sup>, 2237 (CN) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.42 (d, J = 8.7 Hz, 2H), 7.39 (d, J = 8.7 Hz, 2H), 6.70 (dd, J = 17.4, 10.6 Hz, 1H,

7-H), 5.76 (d, J = 17.4 Hz, 1H, 8-H<sup>A</sup>), 5.28 (d, J = 10.6 Hz, 1H, 8-H<sup>B</sup>), 3.61 (t, J = 6.3 Hz, 2H, 1-H<sub>2</sub>),
2.08-1.94 (m, 2H, 3-H<sub>2</sub>), 1.78-1.65 (m, 1H, 2-H<sup>A</sup>), 1.73 (s, 3H, 5-H<sub>3</sub>), 1.57-1.38 (m, 2H, 2-H<sup>B</sup> and OH);
<sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 139.3 (s, *i*), 137.1 (s, *p*), 135.8 (d, C-7), 126.7 (d), 125.6 (d), 123.3 (s, C-6), 114.6 (t, C-8), 62.0 (t, C-1), 42.1 (s, C-4), 38.3 (t, C-3), 28.6 (t, C-2), 27.8 (q, C-5); MS: (EI, 70 eV) *m/z* 215 (M, 21); HRMS: (CI, 70 eV) Calculated (C<sub>14</sub>H<sub>18</sub>NO) 216.1388 ([M + 1]<sup>+</sup>) Found: 216.1391.

4-Azido-4-(4-(1-azidoethyl)phenyl)butan-1-ol (13-N<sub>3</sub>)



To a branched reaction vessel, Fe<sub>2</sub>(ox)<sub>3</sub>·6H<sub>2</sub>O (1.19 g, 2.46 mmol) and H<sub>2</sub>O (20 mL) was added. After stirring for 1 h, the solution was cooled to 0 °C and degassed for 10 minutes. NaN<sub>3</sub> (0.26 g, 4.0 mmol) and THF (10 mL) were added and the vessel was purged by N<sub>2</sub>. To the mixture, 2-(2-(prop-1-en-2yl)phenyl)ethan-1-ol (0.0987 g, 0.524 mmol) in THF (10 mL) was slowly added at 0 °C, and then NaBH<sub>4</sub> (0.06 g, 1,60 mmol) was added at 0 °C. The reaction mixture was stirred at 0 °C for 5 minutes and NaBH<sub>4</sub> (0.06 g, 1,60 mmol) was added. After stirring for 50 minutes at 0 °C, the mixture was quenched by sat. NH<sub>4</sub>Cl aq. (8 mL) and then extracted with MeOH (10 wt % in CH<sub>2</sub>Cl<sub>2</sub> x 2). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 60:40, column length 11 cm, diameter 26 mm, spherical silica gel) to give the product 13-N<sub>3</sub> (0.035 g, 28%). IR: (neat) 3401 (OH) cm<sup>-1</sup>, 2106 (N<sub>3</sub>) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.40 (d, *J* = 7.7 Hz, 2H, *o*), 7.32 (d, *J* = 7.7 Hz, 2H, *m*), 4.62 (q, *J* = 7.0 Hz, 1H, 6-H), 3.58 (t, *J* = 6.0 Hz, 2H, 1-H<sub>2</sub>), 1.91 (t, *J* = 7.7 Hz, 2H, 3-H<sub>2</sub>), 1.70 (s, 3H, 5-H<sub>3</sub>), 1.60-1.46 (m, 1H,  $2-H^{A}$ ), 1.54 (d, J = 7.0 Hz, 3H, 7-H<sub>3</sub>), 1.46-1.31 (m, 1H, 2-H<sup>B</sup>), 1.31-1.18 (br, 1H, OH); <sup>13</sup>C NMR: (100) MHz, CDCl<sub>3</sub>) 143.2 (s, *i*), 139.9 (s, *p*), 126.5 (d, *m*), 125.9 (d, *o*), 66.5 (s, C-4), 62.6 (t, C-1), 60.6 (d, C-6), 38.5 (t, C-3), 27.5 (t, C-2), 25.7 (q, C-5), 21.5 (q, C-7); MS: (EI, 70 eV) m/z 232 (M - N<sub>3</sub>, 15); HRMS: (EI, 70 eV) Calculated (C13H18N6O) 274.1542 (M) Found: 274.1546.

Trimethyl(2-vinylphenethoxy)silane (TMS-1a)

TMS-1a in a crude product was identified by <sup>1</sup>H NMR of an authentic sample of TMS-1a (Table 1). The authentic sample was prepared by the following procedure.<sup>7b</sup> The solution of HN(SiMe<sub>3</sub>)<sub>2</sub> (0.80 mmol, 0.129 g) in CH<sub>2</sub>Cl<sub>2</sub> (0.5 mL) was slowly added to the solution of 1a (0.99 mmol, 0.147 g) and I<sub>2</sub> (0.010 mmol, 0.0025 g) at room temperature. After stirring for 1 hour at room temperature, Na<sub>2</sub>S<sub>2</sub>O<sub>3</sub> (1 g) was added to the reaction mixture and the resulting precipitation was stirred for 30 min. The supernatant liquid was directly purified by silica gel column chromatography (only CH<sub>2</sub>Cl<sub>2</sub> was used as a mobile-phase) to obtain trimethyl(2-vinylphenethoxy)silane TMS-1a (0.88 mmol, 0.194 g, 89%). IR: (neat) 3087, 3062, 3027, 1627, 1484, 1450, 1412, 1383 cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.49 (dd, *J* = 5.6, 4.1 Hz, 1H), 7.23-7.15 (m, 3H), 7.02 (dd, *J* = 17.4, 10.9 Hz, 1H), 5.65 (dd, *J* = 17.4, 1.2 Hz, 1 H), 5.30 (dd, *J* = 10.9, 1.2 Hz, 1H), 3.72 (t, *J* = 7.3 Hz, 2H), 2.93 (t, *J* = 7.3 Hz, 2H), 0.06 (s, 9H); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 137.0 (s), 136.0 (s), 134.5 (d), 130.4 (d), 127.7 (d), 126.7 (d), 125.7 (d), 115.7 (t), 63.2 (t), 36.7 (t), -0.61 (q); MS: (EI, 70 eV) m/z; HRMS: (EI, 70 eV) 220.1283 (M) Calculated Found: 220.1285.

#### 2-5. References

- (1) Comprehensive Organic Synthesis II, Vol. 4 Knochel, P., Mo lander, G. A., Eds.
- (2) For recent reviews: a) S. W. M. Crossley, C. Obradors, R. M. Martinez, R. A. Shenvi, *Chem. Rev.* 2016, *116*, 8912; b) V. Rodriguez-Ruiz, R. Carlino, S. Bezzenine-Lafollée, R. Gil, D. Prim, E. Schulza, J. Hannedouche, *Dalton Trans.* 2015, *44*, 12029; c) X. Zeng, *Chem. Rev.* 2013, *113*, 6864; d) *Hydrofunctionalization*; Ananikov, V. P., Tanaka, M., Eds.; Topics in Organometallic Chemistry; Springer-Verlag: Berlin, Heidelberg, 2013; p 43; e) N. T. Patil, R. D. Kavthe, V. S. Shinde, *Tetrahedron* 2012, *68*, 8079. For polymerization of vinylarenes catalyzed by indium salts: f) M. Bompart, J. Vergnaud, H. Strub, J.-F. Carpentier, *Polym. Chem.* 2011, *2*, 1638; g) C. Peppe, E. S. Lang, F. M. Andrade, L. B. Castro, *Synlett* 2004, 1723; h) T. Tsuchimoto, S. Kamiyama, R. Negoro, E. Shirakawa, Y. Kawakami, *Chem. Commun.* 2003, 852; i) J. Dai, J. Wu, G. Zhao, W.-M. Dai, *Chem. Eur. J.* 2011, *17*, 8290.
- (3) For selective examples: Allylsilane: a) S. E. Denmark, R. Y. Baizitov, J. Org. Chem. 2006, 71, 593.
  Alkenylsilane: b) Y. Nakao, J. Chen, H. Imaoka, T. Hiyama, Y. Ichikawa, W.-L. Duan, R. Shintani, T. Hayashi, J. Am. Chem. Soc. 2007, 129, 9137. Alkynylsilane: c) Y.-l. Xu, Y.-m. Pan, P. Liu, H.-S. Wang, X.-Y. Tian, G.-F. Su, J. Org. Chem. 2012, 77, 3557. Propargylsilane: d) M. Dziedzic, G. Lipner, J. M. Illangua, B. Furman, Tetrahedron 2005, 61, 8641. Silyl enolate: e) H. Tamagaki, Y. Nawate, R. Nagase, Y. Tanabe, Chem. Commun. 2010, 46, 5930. Silyl sulfide: f) M. Fiorenza, G. Reginato, A. Ricci, M. Taddei, J. Org. Chem. 1984, 49, 551. Silyl cyanide g) G. Strappaveccia, T. Angelini, L. Bianchi, S. Santoro, O. Piermatti, D. Lanari, L. Vaccaro, Adv. Synth. Catal. 2016, 358, 2134. Silyl azide: h) S.-G. Kim, T.-H. Park, Synth. Commun. 2007, 37, 1027.
- (4) Silyl cyanide: a) A. Falk, A.-L. Göderz, H.-G. Schmalz, Angew. Chem. Int. Ed. 2013, 52, 1576; b) A.

Yanagisawa, T. Nezu, S. Mohri, Org. Lett. 2009, 11, 5286. Silyl azide: c) C. D. Zotto, J. Michaux, A. Zarate-Ruiz, E. Gayon, D. Virieux, J.-M. Campagne, V. Terrasson, G. Pieters, A. Gaucher, D. Primd, J. Organomet. Chem. 2011, 696, 296; d) G. W. Breton, K. A. Daw, P. J. Kropp, J. Org. Chem. 1992, 57, 6646. Silyl enolate: e) Y. Nishimoto, M. Takeuchi, M. Yasuda, A. Baba, Chem. Eur. J. 2013, 19, 14411;
f) Y. Nishimoto, H. Ueda, Y. Inamoto, M. Yasuda, A. Baba, Org. Lett. 2010, 12, 3390.

- (5) A. S. Jones, J. F. Paliga, M. D. Greenhalgh, J. M. Quibell, A. Steven, S. P. Thomas, Org. Lett. 2014, 16, 5964.
- (6) R. Dorel, A. M. Echavarren, Chem. Rev. 2015, 115, 9028.
- (7) In general, the nucleophilic attack of allylsilanes to electrophiles occurs at γ-position of a silyl group. We reported the indium-catalyzed substitution reaction of hydroxy or trimethylsiloxy group with γ-substituted allylsilanes such as cinnamyl and prenyl silanes to give the corresponding products in γ-addition manner. a) T. Saito, Y. Nishimoto, M. Yasuda, A. Baba, *J. Org. Chem.* 2006, *71*, 8516; b) T. Saito, Y. Nishimoto, M. Yasuda, A. Baba, *J. Org. Chem.* 2007, *72*, 8588.
- (8) Our reported reactions for C-O bonds cleavage by indium-silicon combined Lewis acid catalyst. See: a) Y. Onishi, Y. Nishimoto, M. Yasuda, A. Baba, *Chem. Lett.* 2011, 40, 1223; b) Y. Onishi, Y. Nishimoto, M. Yasuda, A. Baba, *Org. Lett.* 2011, 13, 2762; c) Y. Nishimoto, Y. Onishi, M. Yasuda, A. Baba, *Angew. Chem. Int. Ed.* 2009, 48, 9131; d) T. Saito, M. Yasuda, A. Baba, *Synlett* 2005, 11, 1737; e) Y. Onishi, T. Ito, M. Yasuda, A. Baba, *Tetrahedron* 2002, 58, 8227.
- (9) For selective examples of anti-Markovnikov hydroallenation: a) U. Yokobori, H. Ohmiya, M. Sawamura, *Organometallics* 2012, *31*, 7909; b) C. Alameda-Angulo, B. Quiclet-Sire, S. Z. Zard, *Tetrahedron Lett.* 2006, *47*, 913; c) A. Sato, H. Ito, T. Taguchi, *J. Org. Chem.* 2005, *70*, 709.
- (10) Yields of hydrofunctionalization using Me<sub>3</sub>SiCN and Me<sub>3</sub>SiN<sub>3</sub> were not increased even by the addition of Me<sub>3</sub>SiBr.
- (11) Kobayashi et al have reported a catalytic amount of HI mediated hydroalkoxylation of styrene derivatives to give 6-membered cyclic ether. K. Kobayashi, K. Shikata, H. Maegawa, S. Fukamachi, M. Tanmatsu, H. Konishi, *Heterocycles* 2010, *81*, 2361.
- (12) R. F. Lambert, R. J. Hinkle, S. E. Ammann, Y. Lian, J. Liu, S. E. Lewis, R. D. Pike, J. Org. Chem. 2011, 76, 9269.
- (13) Y. Nishimoto, T. Saito, M. Yasuda, A. Baba, Tetrahedron 2009, 65, 5462.
- (14) The mixture of the silvl ether 11 and target alcohol product 3 was obtained with quenching by H<sub>2</sub>O instead of Bu<sub>4</sub>NF (1 M in THF).
- (15) A. Riera, M. Moreno, *Molecules* 2010, 15, 1041.
- (16) E. K. Leggans, T. J. Barker, K. K. Duncan, D. L. Boger, Org. Lett. 2012, 14, 1428.
- (17) G. W. Breton, K. A. Daus, P. J. Kropp, J. Org. Chem. 1992, 57, 6646.

- (18) R. R. III Cesati, J. D. Armas, A. H. Hoveyda, Org. Lett. 2002, 4, 395.
- (19) A. M. Palmer, S. Chrismann, G. Muench, C. Brehm, P. J. Zimmermann, W. Buhr, J. Senn-Bilfinger, M. P. Feth, W. A. Simon, *Bioorgan. Med. Chem.* 2009, 17, 368.
- (20) I. Yalavac, S. E. Lyons, M. R. Webb, D. J. Procter, Chem. Commun. 2014, 50, 12863.
- (21) J. Tummatorn, S. Ruchirawat, P. Ploypradith, Chem. Eur. J. 2010, 16, 1445.
- (22) C. Rosner, U. Hennecke, Org. Lett. 2015, 17, 3226.
- (23) R. E. Tirpak, M. W. Rathke, J. Org. Chem. 1982, 47, 5099.
- (24) Preparation: A. E. Sheshenev, M. S. Baird, I. G. Bolesov, A. S. Shashkov, *Tetrahedron* 2009, 65, 10552.
- (25) Characterization Data: J. R. McAtee, S. E. S. Martin, A. P. Cinderella, W. B. Reid, K. A. Johnson, D. A. Watson, *Tetrahedron* 2014, *70*, 4250.
- (26) H. Liu, J. Yu, X. Li, R. Yan, J.-C. Xiao, R. Hong, Org. Lett. 2015, 17, 4444.
- (27) Preparation: G. W. Craig, E. D. Sternberg, G. H. Jones, J. G. Moffatt, J. Org. Chem. 1986, 51, 1258.
- (28) Characterization: R. R. Dey, B. Paul, S. S. Dhar, S. Bhattacharjee, Chem. Lett. 2014, 43, 1545.

### Chapter 3

### Selective Oxymetalation of Terminal Alkynes *via* 6-*Endo* Cyclization: Mechanistic Investigation and Application to the Efficient Synthesis of 4-Substituted Isocoumarins

#### **3-1. Introduction**

Heterocyclic compounds have attracted much attention in pharmaceutical chemistry as well as in photochemistry, and also play a pivotal role as building blocks in organic synthetic transformation.<sup>1</sup> Therefore, a novel efficient synthetic method for heterocyclic frameworks is highly desired in various fields of chemistry. Many well-established methods are available in the literature.<sup>2</sup> Heterocyclization of  $\omega$ -heteroatom-substituted alkynes using  $\pi$  acidic metal salts is undoubtedly a powerful strategy for the preparation of heterocycles (Scheme 1).<sup>3</sup> This addition reaction uses readily available alkynes as a starting material. Furthermore, metal salt-mediated cyclization spontaneously forms a carbon-metal bond and a heterocyclic framework and produces organometallic intermediates leading to target heterocycles via appropriate synthetic reactions such as cross coupling. These features allowed us to directly access various substituted heterocycles from simple organic substrates.



Scheme 1. Metal salt-mediated heterocyclization of  $\omega$ -heteroatom-substituted alkynes.

Various heteroatom-containing alkynes have been investigated for use in the synthesis of heterocyclic compounds using  $\pi$  acidic metal salts. Alkyne **A** includes a carbonyl moiety and is a feasible and beneficial substrate to obtain 5- or 6-membered oxacyclic alkenylmetals (Scheme 2A). When **A** is treated with a metal salt (MX), oxymetalation proceeds to afford heterocyclic compounds through either 5-*exo* or 6-*endo* cyclization (Type *exo* or *endo*). Considering that the structure of **A** bears either an internal (R = alkyl, aryl etc.) or a terminal alkyne (R = H), oxymetalation can be distinguished by four types of reaction courses: Type *exo*-i, Type *exo*-t, Type *endo*-i, and Type *endo*-t. In Type *exo*-i and *exo*-t, the furan frameworks **B** and **C** have a metal carbenoid moiety and are obtained via the isomerization of zwitterion intermediates.<sup>4,5</sup> On the other hand, Type *endo*-i and *endo*-t lead to 1*H*-isochromen derivatives **D** and **E** via elimination of R'X from zwitterion intermediates.<sup>6,7</sup> Among these four types,

only Type *endo*-t is kinetically unfavorable due to an unstable cationic transition state via an *anti*-Markovnikov addition manner (Scheme 2B). Furthermore, recent theoretical researches about the regioselectivity of cyclization revealed that nucleophilic cyclization of alkynes displays *exo* selectivity intrinsically.<sup>8</sup> On the other hand, Lewis acidic metals can promote *endo* cyclization by decrease of the stereoelectronic penalty, but the *exo* cyclization was not disturbed, and, thus, the cyclization showed low selectivity.<sup>7,9</sup> For the above reasons, there is no report of a preparation method for species **E** via Type *endo*-t in contrast to the cases of Types *exo*-t, *exo*-t, and *endo*-i, for which target organometallic compounds (**B**, **C**, **D**) are well established.<sup>4b,5a,6c,6e</sup> If the reaction course of oxymetalation is realized from **A** to **E** in Type *endo*-t,<sup>10</sup> various 6-membered heterocycles based on **E**, which have been difficult to obtain and remain unknown, should be synthesized. Therefore, the establishment of a strategy for Type *endo*-t is an important challenge in heterocyclic chemistry.



**Scheme 2.** (A) Four types of metalated heterocycles from oxymetalation of alkyne A including carbonyl moiety. (B) Comparison of transition states of type *exo*-t with type *endo*-t.

Isocoumarins are an important class of oxygen-containing heterocycles that exhibit a wide range of pharmacological properties.<sup>11</sup> Thus, the development of their general synthetic method has attracted much attention. The reaction of Type *endo* would be a powerful tool for the synthesis of isocoumarins (Scheme 3A). In fact, reports have described the oxymetalation of 2-alkynylbenzoate  $\mathbf{1}$  (R = alkyl,

alkenyl, aryl) for Type *endo*-i and application to the synthesis of isocoumarins.<sup>6a,6b,6e</sup> Recently, Blum reported an excellent method for the construction of 4-borylated isocoumarins by oxyboration of the internal alkynes **1** in the Type *endo*-i reaction course (Scheme 3B).<sup>6e</sup> However, terminal alkyne **1** (R = H) gave only a 5-*exo* cyclization product according to the Markovnikov rule (blue path in Scheme 3C).<sup>6e</sup> This result prompted us to explore oxymetalation of the terminal alkynes **1** for Type *endo*-t. Oxymetalation of Type *endo*-t provides 3-unsubstituted and 4-substituted isocoumarins that are seldom investigated due to the lack of synthetic methods,<sup>12</sup> and limited substituents have been introduced at the 4-position despite well-known beneficial significant bioactivity characteristics such as antitumor,<sup>13</sup> antiangiogenic,<sup>14</sup> antifungal,<sup>15</sup> and antibiotic.<sup>16</sup>



Scheme 3. (A) Oxymetalation of 2-alkynylbenzoate 1 followed by transformation for the construction of isocoumarins. (B) Previously reported oxyboration of internal alkynes to generate 4-borylated isocoumarins. (C) Oxymetalation of terminal alkynes, reported oxyboration for 5-membered compounds (blue path), and our developed oxymetalation for 6-membered compounds (red path).

Our group developed the indium or gallium salt-mediated carbometalation of simple terminal alkynes with silyl ketene acetals by utilizing their high  $\pi$  electron affinity and moderate Lewis acidity.<sup>17</sup> In this context, we investigated the Type *endo*-t reaction of 2-ethynylbenzoate using indium or gallium trihalides for the synthesis of corresponding metalated isocoumarins. In this report, we successfully achieved a 6-*endo* selective oxymetalation of terminal alkynes and fully characterized the target organometallic species **E1** via NMR study and X-ray crystallographic analysis. Furthermore, the intermediate **F1** was isolated, which revealed that oxymetalation proceeds via the zwitterion intermediate **F1**, and elimination of the alkyl halide gives the target product **E1** (Scheme 4). While benzopyrylium species such as **F** are known as highly reactive intermediates in the proposed catalytic oxymetalation cycle,<sup>10a,18</sup> the isolation of species **F** is a challenge issue.<sup>10e,10f,19</sup> To the best of our knowledge, **F1** is the first example of a fully characterized benzopyrylium intermediate **F**. In addition, we performed full disclosure of the mechanism by combining experimental data and theoretical calculation. These mechanistic investigations were consistent with the achievement of isolation of the zwitterion intermediate and demonstrated that its stability is a crucial point in this remarkable cyclization regioselectivity.



Scheme 4. Oxyindation of alkynes for synthesis of isocoumarin framework via a zwitterion intermediate.

#### 3-2. Results and Discussion

#### **Optimization of reaction conditions**

First, we examined the effect of Lewis acids on oxymetalation using methyl 2-ethynylbenzoate **1a** (Table 1). The reaction of **1a** with metal halides was carried out in toluene at 50 °C, and the reaction mixture was quenched with acetic acid. The reaction using  $InCl_3$  afforded the target isocoumarin **2** via 6-*endo* cyclization, albeit in a low yield (entry 1). Gratifyingly,  $InBr_3$  and  $InI_3$  mediated oxymetalation smoothly proceeded in 6-*endo* cyclization fashion to give **2** in high yields (entries 2 and 3). In these cases, the reaction mixture was quenched by deuterated acetic acid to afford **2** bearing deuterium at the 4-position. We did not observe an isocoumarin bearing deuterium at the 3- position which could be produced through the generation of indium acetylide<sup>20</sup> followed by Lewis acid mediated cycloaddition. The reaction using  $InI_3$  showed a higher ratio of D/H than the case of  $InBr_3$ . This result suggested the

more efficient generation of the alkenylmetal intermediate **X** in the case of InI<sub>3</sub>. Gallium salts were also suitable for the 6-*endo* cyclization of **1a**, and GaI<sub>3</sub> gave a high yield (entries 4 and 5). On the other hand, typical Lewis acids such as AlCl<sub>3</sub>, AlI<sub>3</sub>, BBr<sub>3</sub> and TiCl<sub>4</sub> were ineffective (entries 6-9). Transition metal salts such as PdCl<sub>2</sub>, CuBr<sub>2</sub> and FeBr<sub>3</sub> provided no target product and resulted in a decomposition of **1a** (entries 10-12). Alkynophilic  $\pi$ -acids such as gold and silver salts were subjected into the present cyclization. It was found that AuCl<sub>3</sub>, AlCl, AgOTf and AuCl/AgOTf resulted in low yields (entries 13-16). A decrease in yield was observed at lower temperature (entry 17). The solvent effect was examined on oxyindation using InI<sub>3</sub>. Dichloroethane as a solvent provided a good yield while chlorobenzene and hexane afforded only moderate yields (entries 18-20). The yields were appreciably decreased in CH<sub>3</sub>CN and THF (entries 21 and 22) probably because the coordination of these solvents to InI<sub>3</sub> decreased the Lewis acidity. Finally, InI<sub>3</sub> was the most effective Lewis acid, and, therefore, we chose entry 3 to represent the optimal conditions.



0 OMe 1a (1.0 equiv)	+ MX <sub>n</sub> (1.0 equiv)	solvent 50 °C, 24 h	$X^{\text{O}}$ + $X_{n-1}M^{\text{O}}$	$ \left  \begin{array}{c} CH_{3}CO_{2}H \\ H \\ H \end{array} \right  $	+ +
	Entry	MX <sub>n</sub>	Solvent	Yield of <b>2</b> $(\%)^b$	
	1	InCl <sub>3</sub>	toluene	13	
	$2^c$	InBr <sub>3</sub>	toluene	82 (77% D)	
	3 <sup>c</sup>	InI <sub>3</sub>	toluene	79 (91% D)	
	4	GaBr <sub>3</sub>	toluene	30	
	5	GaI <sub>3</sub>	toluene	77	
	6	AlCl <sub>3</sub>	toluene	0	
	7	AlI <sub>3</sub>	toluene	0	
	8	$BBr_3$	toluene	0	
	9	TiCl <sub>4</sub>	toluene	0	
	10	PdCl <sub>2</sub>	toluene	0	
	11	CuBr <sub>2</sub>	toluene	0	
	12	FeBr <sub>3</sub>	toluene	0	
	13	AuCl <sub>3</sub>	toluene	7	
	14	AuCl	toluene	5	
	15	AgOTf	toluene	31	
	16	AuCl/AgOTf	toluene	18	
	17 <sup>d</sup>	InI <sub>3</sub>	toluene	61	
	18	InI <sub>3</sub>	ClCH <sub>2</sub> CH <sub>2</sub> Cl	78	
	19	InI <sub>3</sub>	ClC <sub>6</sub> H <sub>5</sub>	57	
	20	InI <sub>3</sub>	hexane	57	
	21	InI <sub>3</sub>	CH <sub>3</sub> CN	17	
	22	InI <sub>3</sub>	THF	0	

[a] Reaction conditions: 1a (0.5 mmol), Lewis acid  $MX_n$  (0.5 mmol), solvent (1 mL), 50 °C, 24 h. [b] The yield of 2 was determined by <sup>1</sup>H NMR. [c] The reaction mixture was quenched by CH<sub>3</sub>CO<sub>2</sub>D (30 equiv, 5 min) and a subsequent addition of H<sub>2</sub>O (10 mL). [d] 35 °C.

#### **Mechanistic Investigation**

To gain insight into the reaction mechanism, we used <sup>1</sup>H NMR spectroscopy to monitor the oxyindation. When 2-alkynylbenzoate **1a** was mixed with InI<sub>3</sub> in CDCl<sub>3</sub> at -30 °C, no reaction occurred. At -5 °C, some amount of a new product was observed (See Fig. S1 and S2 in Experimental Section). At room temperature, a large amount of white precipitation was formed. This white solid was also obtained in the reaction of 1a with InI<sub>3</sub> in toluene at room temperature (Eq. 1). X-ray crystallographic analysis revealed that the white solid was a 6-membered oxacycle zwitterion  $\mathbf{3}$  bearing a carbon-indium bond (Fig. 1). The bond lengths of two carbon-oxygen bonds (C1-O1 = 1.267 Å, C1-O2 = 1.298 Å) in the zwitterion 3 existed between a C-O double bond (1.203 Å) and the single bond (1.377 Å) of a typical isocoumarin derivative,<sup>21</sup> and, thus, the positive charge was delocalized in an ester moiety. The indium atom was coordinated with three iodines and showed a distorted tetrahedral structure with a formal negative charge. The formed zwitterionic alkenyl indium 3 was heated at 50 °C in toluene to give a neutral alkenylindium product 4a, quantitatively by elimination of MeI (Eq. 2). Although a suitable single crystal of 4a for X-ray analysis was not obtained, we successfully conducted X-ray diffraction analysis of nitro-substituted alkenylindium 4b produced from the 2-ethynyl-5-nitrobenzoate 1b (Eq. 3 and Fig. 2). The bond lengths of C1-O1 (1.211 Å) and C1-O2 (1.367 Å) were similar to those of reported isocoumarin framework.<sup>21</sup> The indium complex **4b** displayed trigonal bipyramidal coordination with two THF ligands in axial positions. These results indicated a two-step pathway including a fast cyclization and a slow elimination of MeI during the 6-endo oxyindation processing from 1 to 4.





**Figure 1.** The X-ray crystallographic structure of zwitterion intermediate **3** with the thermal ellipsoids shown at 50% probability (CCDC 1579824). Selected bond lengths (Å): C1-O1 = 1.267(9), C1-O2 = 1.298(11), C2-In = 2.171(7), In-I1 = 2.7392(8), In-I2 = 2.7219(8), In-I3 = 2.6915(7).



**Figure 2.** The X-ray crystallographic structure of isocoumarin including a carbon-indium bond at the 4-position,  $4b \cdot (THF)_2$ , with the thermal ellipsoids shown at 50% probability (CCDC 1576342). Selected bond lengths (Å) and angles (deg): C1-O1 = 1.211(4), C1-O2 = 1.367(5). C2-In = 2.162(3), In-I1 = 2.7148(4), In-I2 = 2.7005(4), In-O5 = 2.318(3), In-O6 = 2.371(3), O5-In-O6 = 175.44(10), I1-In-C2 = 116.70(11), C2-In-I2 = 125.51(11), I2-In-I1 = 117.621(12).

#### **Theoretical Calculation for Oxyindation**

A mechanism for the formation of the target isocoumarin 4a using InI<sub>3</sub> is proposed in Scheme 5 wherein InI<sub>3</sub> is coordinated by the alkyne moiety in **5**, oxyindation proceeds via 6-*endo* cyclization to give the zwitterion intermediate **3**, and, finally, the elimination of MeI affords **4a**.



Scheme 5. A proposed mechanism for formation of the isocoumarin 4a.

Density functional theory (DFT) calculations were performed to more thoroughly consider the reaction mechanism. Calculation of the potential energy profile for 6-endo cyclization (red) shows in Fig. 3. We selected 1a and  $In_2I_6$  as starting materials because  $InI_3$  exists in a dimer fashion.<sup>22</sup> The coordination of two 1a to In<sub>2</sub>I<sub>6</sub> dissociates the aggregation of InI<sub>3</sub> to give the complex 7, in which InI<sub>3</sub> is chelated by the alkyne moiety and carbonyl group of 1a (Fig. S3 in Experimental Section shows detail mechanism of generating the complex 7 from 1a and  $In_2I_6$ ). Dissociation of the carbonyl oxygen atom generates complex 5, in which  $InI_3$  directly activates the alkyne moiety. In this pathway, the anti-addition of InI<sub>3</sub> and the ester moiety into the alkyne moiety proceeds in a concerted mechanism to provide a stable 6-membered zwitterion intermediate 3. Elimination of MeI proceeds in an intermolecular fashion, because the intramolecular elimination of MeI requires a very unstable intermediate (Fig. S4 in Experimental Section shows the potential energy profile for the intramolecular elimination of MeI). Two zwitterions aggregate in a head-to-tail fashion to give complex  $\mathbf{8}$ , and then the elimination step starts from 8. Intermolecular nucleophilic substitution of the methyl group by I<sup>-</sup> proceeds in an S<sub>N</sub>2mechanism to give complex 10 and MeI, and then a subsequent elimination of MeI affords the target product 4a.<sup>23</sup> A carbonyl group of 4a coordinates to the indium atom of another 4a to give the stable dimeric product 13. The activation energy of the elimination step (8 to TS2-6-endo, 28.7 kcal/mol) is much higher than that of the cyclization step (7 to TS1-6-endo, 19.7 kcal/mol).<sup>24</sup> Therefore, the elimination of MeI is a rate-determining step. We also calculated the 5-exo cyclization pathway (blue) to investigate the regioselectivity. This process proceeds via concerted cyclization wherein the 5membered zwitterion 6 is much more unstable than the 6-membered version 3. Intermolecular elimination of MeI takes place in an S<sub>N</sub>2-manner (9 to 11). The energy level of the transition state (TS2-5-exo) shows the highest energy level, and it is higher even than the energy profile of the 6-endo cyclization process (red) due to the instability of the 5-membered zwitterion 6.

In order to clarify the unique 6-*endo* cyclization selectivity of oxyindation, the energy profiles of the two cyclization manners were compared. The activation energy of 5-*exo* cyclization is lower (7 to TS1-5-*exo*, 15.5 kcal/mol) than that of 6-*endo* cyclization (7 to TS1-6-*endo*, 19.7 kcal/mol). However, 5-*exo* cyclization is reversible because the activation energy for the elimination of MeI (9 to TS2-5-*exo*, 21.9 kcal/mol) is much higher than that of retro-cyclization (6 to TS1-5-*exo*, 9.3 kcal/mol) due to the instability of the zwitterion 6. On the other hand, during 6-*endo* cyclization, both activation energies of elimination (8 to TS2-6-*endo*, 28.7 kcal/mol) and retro-cyclization (3 to TS1-6-*endo*, 23.0 kcal/mol) are high because the 6-membered zwitterion intermediate 3 is thermodynamically stable. This result indicates that 6-*endo* cyclization is irreversible and the most thermodynamically stable form of intermediate 8 is exclusively generated to provide the target product 4a, which is consistent with the successful isolation of the zwitterion intermediate 3 (Fig. 1). Therefore, oxyindation proceeds under
thermodynamic control to afford the stable 6-membered product **4a**. We also calculated an energy profile of  $InCl_3$ -mediated oxyindation and found the same pathway with the case of  $InI_3$  (See Fig. S5 in Experimental Section). The activation energy of elimination step in the cases of  $InCl_3$  is higher than that of  $InI_3$  because of low nucleophilicity of  $Cl_3^-$  and it caused much less reactivity of  $InCl_3$  (entry 1, Table 1).



**Figure 3.** The energy profiles of 6-*endo* and 5-*exo* oxyindations and 3D molecular structures of transition states. DFT calculation was performed using wB97XD/6-31+G (d,p) for C, H, and O and using DGDZVP for In and I. Solvation effect was introduced using the IEFPCM model, and toluene was used as a solvent.

The remarkable regioselectivity of oxyindation is ascribed to the differences in stability between the 6-membered zwitterion **3** and the 5-membered **6**. Zwitterion **3** is much more stable than **6**, and this difference in stability originates from the aromaticity of these compounds, although ring strain is also a consideration. To verify this possibility, the aromaticity of zwitterions was evaluated via NICS(1)<sup>25</sup> (Fig. 4), and the 6-membered compound **3** showed a higher level of aromaticity than that of **6**.



**Figure 4.** NICS(1) values of 6-membered zwitterion **3** and 5-membered zwitterion **6**. The aromaticity was calculated using B3LYP/6-31G (d,p) for C, H, and O and using DGDZVP for In and I.

### **Theoretical Calculation for Oxyboration**

Blum and co-workers reported that oxyboration of **1a** using *B*-chlorocatecholborane (ClBcat) gave a 5-membered product<sup>6e</sup> rather than the 6-membered version (Scheme 6). ClBcat is coordinated by the carbonyl moiety of **1a**. Then, oxyboration proceeds via 5-*exo* cyclization to give the zwitterion intermediate **16**, and the elimination of MeCl gives the target product **18**. We also performed DFT calculation of oxyboration to investigate the striking change in the regioselectivity between oxyboration and oxyindation. First, the calculation of oxyboration was performed for a similar oxyindation mechanism via concerted cyclization and S<sub>N</sub>2-type elimination of MeCl from aggregated zwitterion intermediates (see Fig. S6 in Experimental Section). We considered another possibility for the elimination step, because recent theoretical investigation of ClBcat-mediated heterocyclization has shown other mechanisms,<sup>26</sup> whereby the Me group is attacked either by dissociated chloride<sup>26a</sup> or by [Cl<sub>2</sub>Bcat]<sup>-26b</sup>. Thus, we considered these additional two plausible elimination steps assisted either by free Cl<sup>-</sup> or [Cl<sub>2</sub>Bcat]<sup>-</sup> (See Fig. S7 in Experimental Section and Fig. 5 and 6). The result of comparison between these three pathways showed that the most probable path was the use of [Cl<sub>2</sub>Bcat]<sup>-</sup> (Details of the comparison are shown in Experimental Section).



Scheme 6. A proposed mechanism for oxyboration.

The total reaction profile of oxyboration is described in Fig. 5 and 6. In that profile, 5-*exo* cyclization from **1a** and 2ClBcat to **16** has an activation energy (27.9 kcal/mol) that is lower than that of 6-*endo* cyclization (**1a** and 2ClBcat to **15**, 34.4 kcal/mol). The chloride moiety of zwitterion **16** coordinates to another ClBcat to provide complex **19**. The chloride transfer process (**19** to **20**) has a low energy barrier (5.8 kcal/mol), and  $[Cl_2Bcat]^-$  is generated rapidly. Cl in  $[Cl_2Bcat]^-$  approaches the methyl group in the ester moiety (**20** $\rightarrow$ **21** $\rightarrow$ **22**), and an elimination of MeCl (**22** to **23**) in the S<sub>N</sub>2-mechanism occurs to give 5-membered product **18**. The activation energy of the elimination of MeCl (**22** to TS3-5*-exo*) is 20.7 kcal/mol, which allows the elimination of MeCl to proceed smoothly to give the final product **18**. The fast elimination step allows oxyboration to proceed under kinetic control to accomplish the 5*-exo* selective cyclization.



**Figure 5.** The energy profiles of 5-*exo* and 6-*endo* oxyborations. DFT calculation was performed by wB97XD/6-31+G (d,p) for C, H, O, B, and Cl. Solvation effect was introduced using the IEFPCM model, and toluene was used as a solvent.



Figure 6. 3D molecular structures of transition states in oxyborations.

# Comparing the Transition State of the Cyclization Step in Oxyindation with That in Oxyboration Based on an Electrostatic Potential Map

The significant difference between oxyindation and oxyboration was investigated because each showed a characteristic energy profile, particularly for the cyclization step. The energy barrier of cyclization in oxyindation (6-endo: 19.7 kcal/mol, 5-exo: 15.5 kcal/mol) is much lower than that of oxyboration (6-endo: 34.4 kcal/mol, 5-exo: 27.9 kcal/mol). Therefore, the electrostatic potential maps for the transition states of cyclization (TS1-6-endo and TS1-5-exo) were calculated (Fig. 7). The value of  $V_{\min}$ , which represents the most negative surface electrostatic potential, was investigated to evaluate the degree of localization for a negative charge.<sup>27</sup> The  $V_{min}$  of the organoindium species (left, in Fig. 7) was less negative than that of boron (right, in Fig. 7), which showed that the negative charge was delocalized in the transition state of oxyindation compared with oxyboration. The value of  $V_{\text{max}}$ , which is the most positive surface electrostatic potential, was also calculated and was less affected by the differences in the metals (see Table S2 in Experimental Section). The polarizability of the indium, boron and heteroatoms binding to a metal explained these results. Indium and iodine atoms have large polarizability ( $\alpha_{In} = 69$  a.u.,  $\alpha_I = 35.1$  a.u.),<sup>28</sup> and the increasing negative charge in the TS1 of oxyindation was efficiently delocalized to stabilize the zwitterionic TS1-6-endo.29 On the other hand, boron, chlorine and oxygen atoms have smaller polarizability ( $\alpha_B = 20.5$  a.u.,  $\alpha_{Cl} = 14.7$  a.u.,  $\alpha_O = 6.04$  a.u.)<sup>28</sup> than indium and iodine atoms, so that TS1-5-exo becomes unstable due to the localization of a negative charge. The difference in the fundamental features between indium and boron atoms imparts a significant amount of influence to the regioselectivity of oxymetalation.



Figure 7. Electrostatic potential maps were calculated on the 0.001 au isosurface of electron density for optimized structures of the transition states of oxyindation (left) and oxyboration (right). The potential is depicted by a color gradient from the most negative (red) to the most positive (blue) value (kcal/mol).  $V_{min}$  represents the most negative surface electrostatic potential.

### **Summary of DFT Calculation**

In oxyindation (Fig. 8A), the activation energy of 5-*exo* cyclization is much lower than that required for the elimination of MeI to lead reversible 5-*exo* cyclization. Therefore, the thermodynamically stable 6-membered zwitterion **3** was selectively produced to accomplish the remarkable 6-*endo* selectivity. The elimination step from **3** is a rate-determining step that provides the target metalated isocoumarin **4a**. On the other hand, the energy barrier for cyclization in oxyboration (Fig. 8B) is higher than that for the elimination of MeCl and the cyclization step is a rate-determining step, which leads to irreversible 5-*exo* cyclization to afford the 5-membered product **18** under kinetic control. Therefore, activation energies of cyclization as well as elimination are important factors to determine the regioselectivity in cyclization.



Figure 8. The summarized results of DFT calculation.

### Application to the Synthesis of Isocoumarin Derivatives

Our developed oxyindation was applied to the synthesis of isocoumarin derivatives. First, the gramscale synthesis of an organoindium species was carried out. Methyl ester **1a** (10 mmol) reacted with InI<sub>3</sub> to give organoindium **4a**, and 1.14 g of isocoumarin was isolated by the addition of H<sub>2</sub>O (Scheme 7).



Scheme 7. Gram-scale synthesis of isocoumarin including a carbon-indium bond.

Next, the oxidation of produced alkenylindium compounds was performed (Table 2). An oxyindation of **1a** using InI<sub>3</sub> was carried out, and the organoindium **4a** was oxidized by PhI(OAc)<sub>2</sub> in a one-pot procedure to give 4-iodoisocoumarin **24a** (entry 1). Subjecting InBr<sub>3</sub> to the oxidation reaction provided 4-bromoisocumarin **25a** in a high yield (entry 2). Therefore, various types of 2-alkynylbenzoates were surveyed in the sequential oxyindation/halogenation process to give 4-halogenated isocoumarins. Substrates with electron withdrawing groups such as nitro and carbonyl groups gave the target products **24b** and **24c** in high yields (entries 3 and 4). The structure of **24b** was characterized by X-ray crystallographic analysis. Substrates with methyl or aryl groups efficiently afforded the target isocoumarins **24d** and **24e** (entries 5 and 6). Also, 2-alkynylbenzoates, including halogen moieties (Br, Cl and F), were suitable for this reaction system to give the isocoumarins **25f-24h** in moderate yields (entries 7-9). The synthesis of isocoumarins from internal alkynes was also investigated. Optimization of the reaction conditions showed that gallium salts were more suitable than indium salts for the oxymetalation of an internal alkyne (See Table S3 in Experimental Section). Therefore, gallium salts were employed in the reactions of internal alkynes **1i**, **1j** and **1k** to provide the 3,4-disubstituted isocoumarins **24i**, **24j** and **25k** (entries 10-12).

One-pot syntheses of 4-substituted isocoumarins were performed via oxyindation followed by a palladium-catalyzed cross-coupling reaction (Table 3).<sup>30</sup> After the oxymetalation of **1a** using InBr<sub>3</sub>, the addition of a palladium catalyst, lithium chloride, organic halides **27**, and an additional solvent to the resultant toluene solution afforded the coupling product **28**. Iodobenzene **27a** and the aryl iodides bearing electron donating group **27b** or electron withdrawing group **27c** were applicable to give the 4-arylisocoumarins **28aa-28ac** in high yields (entry 1). Palladium-catalyzed cross coupling with acid chlorides also proceeded efficiently. Reactions using the benzoyl chloride derivatives **27d** and **27e**, as well as the alkanoyl chloride **27f**, afforded the isocoumarins **28ad-28af** with ketone moieties in good

yields (entries 2 and 3). The structure of **28ae** was characterized by X-ray crystallographic analysis. In this reaction system, alkyl halides such as the benzyl bromide **27g** and the allyl bromide **27h** were also suitable to give the 4-alkylisocoumarins **28ag** and **28ah**, respectively (entries 4 and 5). Various types of 4-substituted isocoumarins were obtained from an isocoumarin that included a carbon-indium bond by utilizing palladium-catalyzed cross coupling.



Table 2 Sequential Oxymetalation/Halogenation of Various Types of 2-Alkynylbenzoate 1.<sup>[a]</sup>



 Table 3. One-Pot Formation of 4-Substituted Isocoumarins by Palladium-Catalyzed Cross-coupling of

 Organoindium Species 26 with Organic Halides 27.<sup>[a]</sup>



[a] Basic reaction conditions of the first step: **1a** (0.5 mmol), InBr<sub>3</sub> (0.5 mmol), toluene (1 mL), 50 °C, 24 h. Second step: Pd<sub>2</sub>dba<sub>3</sub> (0.025 mmol), LiCl (1.0 mmol), **27** (1.0 mmol), NMP (2.5 mL), 50 °C, 24 h. [b] Isolated yields. [c] HMPA (2.5 mL), RT, 24 h. [d] HMPA (2.5 mL), 50 °C, 24 h. [e] *E*/*Z* = 90:10.

### **Formal Total Synthesis of Oosponol**

Finally, a formal total synthesis of oosponol, which exhibits strong antifungal activity,<sup>15</sup> was conducted (Scheme 8). Firstly, iodination of commercially available compound **29** proceeded via a method found in the literature.<sup>31</sup> During the initial investigation, **30** was transformed into methyl 2-ethynyl-6-methoxybenzoate, and then we attempted the synthesis of the precursor of oosponol via oxyindation and cross-coupling, but the reaction returned a complicated mixture. Therefore, in another synthetic route, the OMe group of **29** was converted to an OAc group with less ability to donate electrons. The OMe moiety of **30** was completely deprotected by BBr<sub>3</sub>. Acid-catalyzed esterification and acetylation of the phenol moiety gave methyl 6-iodoacetylsalicylate **33** in a high yield. Sonogashira coupling followed by the removal of a silyl moiety afforded the desired 2-alkynylbenzoate derivative **35**. Oxymetalation of **35** using InBr<sub>3</sub> and sequential palladium-catalyzed cross coupling with acid chloride **27i** produced the key intermediate **36**, and the hydrolyzation of **36** yielded oosponol.<sup>16b</sup> Our method used a readily available starting material and gave a higher yield than previous works.<sup>16b,32</sup>



Scheme 8. Formal Total Synthesis of Oosponol. Reagents and reaction conditions: a) BBr<sub>3</sub> (1 M in CH<sub>2</sub>Cl<sub>2</sub>, 2.0 equiv), CH<sub>2</sub>Cl<sub>2</sub>, RT, 20 h, 100%. b) H<sub>2</sub>SO<sub>4</sub> (20 mol %), MeOH, Reflux, 20 h, 87%. c) AcCl (1.04 equiv), Pyridine (1.04 equiv), Acetone, RT, 14 h, 97%. d) Ethynyltrimethylsilane (1.1 equiv), PdCl<sub>2</sub>(PPh<sub>3</sub>)<sub>2</sub> (2.0 mol %), CuI (20 mol %), NEt<sub>3</sub>, RT, 17 h, 100%. e) 1M KF aq. (1.65 equiv), DMF, RT, 0.5 h, 76%. f) InBr<sub>3</sub> (1.0 equiv), Toluene, 50 °C, 24 h. g) Pd<sub>2</sub>dba<sub>3</sub> (5.0 mol %), LiCl (2.0 equiv), 2-(acetyloxy)acetyl chloride **27i** (2.0 equiv), HMPA, RT, 9 h, 44%.

### 3-3. Conclusion

We achieved the synthesis of isocoumarins bearing a metal-carbon bond at the 4-position via 6endo selective oxymetalation of 2-alkynylbenzoate 1 (Type endo-t). Indium and gallium salts showed high activity for the oxymetalation of 2-ethynylbenzoate 1a. Both the metalated isocoumarin 4b and the zwitterion intermediate 3 were identified by X-ray crystallographic analysis. This is the first example of the isolation of the product E and the benzopyrylium intermediate F proposed in the mechanism of oxymetalation (Scheme 2A). The elimination of MeI from zwitterion 3 occurred under heating conditions to give the target product 4a, which means the rate-determining step was the elimination step. DFT calculation suggested that thermodynamic control led to 6-endo selective oxyindation, while kinetic control led to 5-exo selective oxyboration. The 6-membered product proved much more stable than the 5-membered product due to a difference in the degree of aromatic stability. The investigation of electrostatic potential of the transition state in the cyclization pathway suggested that a delocalization of negative charge by large atomic radii of In and I atoms stabilizes the zwitterionic transition state. In contrast, small atomic radii of B, Cl, and O atoms causes a localization of negative charge to destabilize the corresponding transition state. The difference in stability between the 6- and 5-membered zwitterions and the elemental character of InI<sub>3</sub> both played important roles in the unique regioselectivity of oxymetalation and in the facile preparation of the E species.

These isocoumarins bearing a carbon-metal bond at the 4-position were applied to organic synthesis. Oxymetalation provided isocoumarins on a gram scale. The oxidation of organoindium or gallium species yielded various types of 4-halogenated isocoumarins. Palladium-catalyzed cross coupling with aryl iodide, acid chloride, and alkyl bromide gave a wide range of 4-substituted isocoumarins in a one-pot reaction. Therefore, the unprecedented regioselectivity of the present oxymetalation contributed to the synthesis of new types of isocoumarins. We accomplished a formal total synthesis of oosponol to demonstrate the utility of our reaction system.

### 3-4. Experimental Section

#### General

New compounds were characterized by <sup>1</sup>H and <sup>13</sup>C NMR, <sup>13</sup>C off-resonance techniques, COSY, HMQC, HMBC, IR, MS, HRMS. <sup>1</sup>H (400 MHz) and <sup>13</sup>C NMR (100 MHz) spectra were obtained with TMS as internal standard. IR spectra were recorded as thin films or as solids in KBr pellets. High-resolution mass spectra were obtained by magnetic sector type mass spectrometer. Column chromatography was performed on silica gel (MERK C60 or Fuji Silysia FL100DX). All reactions were carried out under nitrogen. Yields were determined by <sup>1</sup>H NMR using internal standards (bromoform). All calculations (Fig. 3-6) were performed with Gaussian09 rev.E.01. The wB97XD density functional and a mixed basis

set of DGDZVP for In and I and 6-31+G (d,p) for other atoms were used. All molecular geometries were fully optimized and Gibbs free energies including contribution of vibrational entropy at an appropriate temperature were described in energy profiles. Solvation effect was introduced using the IEFPCM model and toluene was used as a solvent. Aromaticity of 6- and 5-membered zwitterions (**3** and **6**) was calculated by the B3LYP density functional and a mixed basis set of 6-31G (d,p) for C, H, O and DGDZVP for In, I for their optimized structures. Electrostatic potential maps were calculated on the 0.001 au isosurface of electron density for optimized structures of transition states of oxyindation and oxyboration. The potential is depicted with a color gradient from the most negative (red) to the most positive (blue) value (kcal/mol).

### Materials

Dehydrated toluene was purchased and used without further purification. 2-Alkynylbenzoate derivatives **1a**, **1c**, **1f**, **1g**, **1i**, **1k**, **1l**, **1m** were synthesized by literature procedures and spectral data of these compounds are shown below. Methyl 2-alkynylbenzoate derivatives **1b**, **1d**, **1e**, **1h**, **1j** are new compounds, and synthetic method and spectral data of these compounds are shown below. All metal salt, organohalides **27a-i** and the starting material of oosponol **29** are commercially available.

(1a) Methyl 2-ethynylbenzoate



This compound was synthesized by a reported method<sup>33</sup> and the NMR data was agreement with the literature<sup>33</sup>.

(1b) methyl 2-ethynyl-5-nitrobenzoate



To a three necked flask, methyl 5-nitro-2-((trimethylsilyl)ethynyl)benzoate (2.83 g, 10.2 mmol), MeOH (20 mL) and  $K_2CO_3$  (0.198 g, 1.43 mmol) were added. After stirring for 3 h at RT, the solvent was evaporated and the residue was diluted by  $CH_2Cl_2$  (20 mL). The solution was washed by water (20 mL) and sat. NaCl aq. (20 mL). The organic layer was dried (MgSO<sub>4</sub>) and the solvent was evaporated. The

residue was purified by column chromatography (hexane/ethyl acetate = 80:20, column length 11 cm, diameter 26 mm, spherical silica gel) to give the product (1.56 g, 74%). IR: (KBr) 1701 (C=O) cm<sup>-1</sup>, 1523 (NO<sub>2</sub>) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 8.80 (d, J = 2.3 Hz, 1H, 7-H), 8.33 (dd, J = 8.6, 2.3 Hz, 1H, 5-H), 7.80 (d, J = 8.6 Hz, 1H, 4-H), 4.00 (s, 3H, OMe), 3.73 (s, 1H, 9-H); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 164.2 (s, C-1), 146.9 (s, C-6), 136.0 (d, C-4), 133.7 (s, C-2), 128.9 (s, C-3), 126.0 (d, C-5), 125.5 (d, C-7), 87.9 (d, C-9), 80.3 (s, C-8), 52.8 (q, OMe); MS: (EI, 70 eV) m/z 205 (M, 100), 174 (81), 147 (51), 128 (72), 100 (49), 74 (41); HRMS: (CI, 70 eV) Calculated (C<sub>10</sub>H<sub>8</sub>NO<sub>4</sub>) 206.0453 [M+H]<sup>+</sup> Found: 206.0451.

(1c) dimethyl 2-ethynylterephthalate



This compound was synthesized by a reported method<sup>33</sup> and the NMR data was agreement with the literature<sup>34</sup>.

(1d) methyl 2-ethynyl-5-methylbenzoate

To a three necked flask, methyl 5-methyl-2-((trimethylsilyl)ethynyl)benzoate (7.96 g, 32.3 mmol), MeOH (30 mL) and K<sub>2</sub>CO<sub>3</sub> (0.41 g, 3.0 mmol) were added. After stirring for 15 h at RT, the solvent was evaporated and the residue was diluted by CH<sub>2</sub>Cl<sub>2</sub> (20 mL). The solution was washed by water (20 mL) and sat. NaCl aq. (20 mL). The organic layer was dried (MgSO<sub>4</sub>) and the solvent was evaporated. The residue was purified by column chromatography (hexane/ethyl acetate = 80:20, column length 11 cm, diameter 26 mm, spherical silica gel) to give the product (2.33 g, 40%). IR: (KBr) 1720 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.74 (d, J = 1.4 Hz, 1H, 7-H), 7.50 (d, J = 7.7 Hz, 1H, 4-H), 7.26 (dd, J = 7.7, 1.4 Hz, 1H, 5-H), 3.91 (s, 3H, OMe), 3.36 (s, 1H, 10-H), 2.37 (s, 3H, 8-H<sub>3</sub>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 166.4 (s, C-1), 138.6 (s, C-6), 134.6 (d, C-4), 132.4 (d, C-5), 132.1 (s, C-2), 130.7 (d, C-7), 119.5 (s, C-3), 82.0 (s, C-9), 81.2 (d, C-10), 51.9 (q, OMe), 21.1 (q, C-8); MS: (EI, 70 eV) m/z 174 (M, 94), 143 (92), 115 (100); HRMS: (CI, 70 eV) Calculated (C<sub>11</sub>H<sub>11</sub>O<sub>2</sub>) 175.0759 [M+H]<sup>+</sup> Found: 175.0762.

(Precursor of 1e) 4'-(tert-butyl)-4-((trimethylsilyl)ethynyl)-[1,1'-biphenyl]-3-carboxylate



To a three necked flask, methyl 4'-(*tert*-butyl)-4-iodo-[1,1'-biphenyl]-3-carboxylate (1.20 g, 3.04 mmol), THF (4.5 mL), CuI (0.0128 g, 0.127 mmol), PPh<sub>3</sub> (0.0202 g, 0.0770 mmol), PdCl<sub>2</sub>(PPh<sub>3</sub>)<sub>2</sub> (0.101 g, 0.144 mmol), NEt<sub>3</sub> (0.475 g, 4.69 mmol) and trimethylsilylacetylene (0.45 g. 4.58 mmol) was added. After stirring for 57 h at RT, the reaction mixture was filtered through a celite pad. The filtrate was evaporated and the residue was purified by column chromatography (hexane, column length 11 cm, diameter 26 mm, spherical silica gel) to give the product (0.984 g, 90%). IR: (KBr) 1736 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 8.16 (d, J = 1.9 Hz, 1H, 7-H), 7.66 (dd, J = 8.1, 1.9 Hz, 1H, 5-H), 7.62 (d, J = 8.1 Hz, 1H, 4-H), 7.55 (d, J = 8.7 Hz, 2H, o), 7.46 (d, J = 8.7 Hz, 2H, m), 3.94 (s, 3H, OMe). 1.35 (s, 9H, 10-Me<sub>3</sub>), 0.289 (s, 9H, SiMe<sub>3</sub>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 166.9 (s, C-1), 151.2 (s, p), 140.8 (s, C-6), 136.2 (s, i), 134.9 (d, C-4), 132.9 (s, C-2), 129.6 (d, C-5), 128.6 (d, C-7), 126.6 (d, o), 125.9 (d, m), 121.6 (s, C-3), 103.4 (s, C-8), 100.0 (s, C-9), 52.0 (q, OMe), 34.6 (s, C-10), 31.3 (q, 10-Me<sub>3</sub>), -0.091 (q, SiMe<sub>3</sub>); MS: (EI, 70 eV) m/z 364 (M, 89), 349 (93), 319 (100); HRMS: (EI, 70 eV) Calculated (C<sub>23</sub>H<sub>28</sub>O<sub>2</sub>Si) 364.1859 [M]<sup>+</sup> Found: 364.1855.

(1e) methyl 4'-(tert-butyl)-4-ethynyl-[1,1'-biphenyl]-3-carboxylate



To a three necked flask, methyl 4'-(*tert*-butyl)-4-((trimethylsilyl)ethynyl)-[1,1'-biphenyl]-3-carboxylate (0.73 g, 2.00 mmol), MeOH (2.0 mL) and K<sub>2</sub>CO<sub>3</sub> (0.028 g, 0.203 mmol) were added. After stirring for 18 h at RT, the solvent was evaporated and the residue was diluted by CH<sub>2</sub>Cl<sub>2</sub> (5 mL). The solution was washed by water (5 mL) and sat. NaCl aq. (5 mL). The organic layer was dried (MgSO<sub>4</sub>) and the solvent was evaporated. The residue was purified by column chromatography (hexane/ethyl acetate = 90:10, column length 11 cm, diameter 26 mm, spherical silica gel) to give the product (0.385 g, 65%). IR: (KBr) 1724 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 8.18 (d, *J* = 1.9 Hz, 1H, 7-H), 7.70 (dd, *J* = 8.2, 1.9 Hz, 1H, 5-H), 7.67 (d, *J* = 8.2 Hz, 1H, 4-H), 7.56 (d, *J* = 8.7 Hz, 2H, *o*), 7.49 (d, *J* = 8.7 Hz, 2H, *m*), 3.96 (s, 3H, OMe). 3.43 (s, 1H, 9-H), 1.36 (s, 9H, 10-Me<sub>3</sub>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 166.5 (s, C-

1), 151.4 (s, *p*), 141.2 (s, C-6), 136.2 (s, *i*), 135.4 (d, C-4), 132.9 (s, C-2), 129.9 (d, C-5), 128.7 (d, C-7), 126.7 (d, *o*), 125.9 (d, *m*), 121.0 (s, C-3), 82.5 (d, C-9), 82.1 (s, C-8), 52.2 (q, OMe), 34.6 (s, C-10), 31.3 (q, 10-Me<sub>3</sub>); MS: (EI, 70 eV) m/z 292 (M, 61), 277 (100); HRMS: (EI, 70 eV) Calculated (C<sub>20</sub>H<sub>20</sub>O<sub>2</sub>) 292.1463 [M]<sup>+</sup> Found: 292.1463.

(1f) methyl 5-bromo-2-ethynylbenzoate



This compound was synthesized by a reported method<sup>33</sup> and the NMR data was agreement with the literature<sup>35</sup>.

(1g) methyl 5-chloro-2-ethynylbenzoate



This compound was synthesized by a reported method<sup>33</sup> and the NMR data was agreement with the literature<sup>36</sup>.

(1h) methyl 2-ethynyl-5-fluorobenzoate

$$\mathsf{F}_{\overbrace{\substack{6\\5\\4}}^{7}}, \overbrace{\substack{2\\1\\4}^{2}}, 0 \mathsf{Me}$$

To a three necked flask, methyl 5-fluoro-2-((trimethylsilyl)ethynyl)benzoate (1.05 g, 4.20 mmol), MeOH (4.2 mL) and K<sub>2</sub>CO<sub>3</sub> (0.063 g, 0.457 mmol) were added. After stirring for 5 h at RT, the solvent was evaporated and the residue was diluted by CH<sub>2</sub>Cl<sub>2</sub> (10 mL). The solution was washed by water (10 mL) and sat. NaCl aq. (10 mL). The organic layer was dried (MgSO<sub>4</sub>) and the solvent was evaporated. The residue was purified by column chromatography (hexane/ethyl acetate = 80:20, column length 11 cm, diameter 26 mm, spherical silica gel) to give the product (0.502 g, 67%). IR: (KBr) 1724 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.67-7.56 (m, 2H, 4-H and 7-H), 7.19 (td,  $J^3_{HH}$  = 8.2 Hz,  $J^3_{HF}$  = 8.2 Hz,  $J^4_{HH}$  = 2.9 Hz 1H, 5-H), 3.93 (s, 3H, OMe), 3.40 (s, 1H, 9-H); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 165.0 (s, C-1), 161.7 (d, *J* = 250.7 Hz, C-6), 136.7 (d, *J* = 8.2 Hz, C-4), 134.3 (d, *J* = 8.2 Hz, C-2), 119.1 (d, *J* = 22.1, C-5), 118.6 (d, *J* = 4.1 Hz, C-3), 117.2 (d, *J* = 23.8 Hz, C-7), 82.0 (d, C-9), 80.8 (s, C-8), 52.3

(q, OMe); MS: (EI, 70 eV) m/z 178 (M, 84), 147 (100), 119 (95); HRMS: (CI, 70 eV) Calculated ( $C_{10}H_8O_2F$ ) 179.0508 [M + H]<sup>+</sup> Found: 179.0511.

(1i) methyl 2-(phenylethynyl)benzoate



This compound was synthesized by a reported method<sup>33</sup> and the NMR data was agreement with the literature<sup>37</sup>.

(1j) methyl 2-((4-fluorophenyl)ethynyl)benzoate



To a three necked flask, methyl 2-iodebenzoate (1.30 g, 4.96 mmol), NEt<sub>3</sub> (5 mL), PdCl<sub>2</sub>(PPh<sub>3</sub>)<sub>2</sub> (0.04 g, 0.057 mmol), CuI (0.050 g, 0.50 mmol) was added and 1-ethynyl-4-fluorobenzene (0.70 g. 5.83 mmol) was dropwised at 0 °C. After stirring for 24 h at RT, the reaction mixture was filtered through a celite pad. The filtrate was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 90:10, column length 11 cm, diameter 26 mm, spherical silica gel) to give the product (0.70 g, 55%). IR: (neat) 1732 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.97 (dd, J = 8.2, 1.4 Hz, 1H, 7-H), 7.62 (dd, J = 7.7, 1.4 Hz, 1H, 4-H), 7.56 (dd, J = 9.1, 5.4 Hz, 2H, o), 7.48 (td, J = 7.7, 1.4 Hz, 1H, 5-H), 7.37 (td, J = 7.7, 1.4 Hz, 1H, 6-H), 7.05 (dd, J = 9.1, 8.6 Hz, 2H, m). 3.95 (s, 3H, OMe); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 166.5 (s, C-1), 162.6 (s, d, <sup>1</sup> $J_{CF} = 250$  Hz, p), 133.8 (d), 133.6 (d), 133.5 (d), 131.7 (d, C-5), 131.6 (s, C-2), 130.4 (d, C-7), 127.9 (d, C-6), 123.5 (s, C-3), 119.3 (s, d, <sup>4</sup> $J_{CF} = 2.9$  Hz, i), 115.6 (d, d<sup>2</sup> $_{CF} = 21.9$  Hz, m), 93.2 (s, C-8), 87.9 (s, C-9), 52.1 (q, OMe); MS: (EI, 70 eV) m/z 254 (M, 100), 239 (87), 211 (42), 183 (40); HRMS: (EI, 70 eV) Calculated (C<sub>20</sub>H<sub>20</sub>O<sub>2</sub>) 254.0743 [M]<sup>+</sup> Found: 254.0743.

(1k) methyl 2-(hex-1-yn-1-yl)benzoate



This compound was synthesized by a reported method<sup>33</sup> and the NMR data was agreement with the literature<sup>38</sup>.

### (11) ethyl 2-ethynylbenzoate



This compound was synthesized by a reported method<sup>34</sup> and the NMR data was agreement with the literature<sup>39</sup>.

(1m) isopropyl 2-ethynylbenzoate



This compound was synthesized by a reported method<sup>33</sup> and the NMR data was agreement with the literature<sup>40</sup>.

# Experimental Procedure for the oxymetalation of methyl 2-alkynylbenzoate followed by protonolysis (Table 1 and Scheme 7)

In glove box filled for nitrogen, to a sealed vial,  $InI_3$  (0.5 mmol), toluene (1 mL) and methyl 2ethynylbenzoate **1a** (0.5 mmol) were added. The solution was stirred at 50 °C and the reaction mixture was quenched by deuterated acetic acid (1 mL) or water. After addition of water (10 mL), the solution was extracted with  $CH_2Cl_2$  (5 mL x 3). The collected organic layer was dried (MgSO<sub>4</sub>). In the investigation shown in Table 1, the solvent was evaporated and the yield of deuterated isocoumarin **2** was determined by <sup>1</sup>H NMR (400 MHz, in CDCl<sub>3</sub>). In Scheme 7, the solvent was evaporated and the residue was purified by column chromatography.

# Experimental Procedure for the oxymetalation of 2-alkynylbenzoate and quantitation of alkyl halide (Table S1)

In glove box filled for nitrogen, to a sealed vial,  $InI_3$  (0.5 mmol), toluene- $d_8$  (1 mL) and 2ethynylbenzoate **1** (0.5 mmol) were added. After stirring at 50 °C for 1 h or 3 h, internal standard (CHBr<sub>3</sub>) was added to the reaction mixture in glove box filled for nitrogen and then the yield of alkyl halide and the conversion of 2-ethynylbenzoate was determined by <sup>1</sup>H-NMR.

# Experimental Procedure for the oxymetalation of 2-alkynylbenzoate followed by halogenation (Table 2)

In glove box filled for nitrogen, to a sealed vial,  $InI_3$  (0.5 mmol), toluene (1 mL) and methyl 2alkynylbenzoate **1** (0.5 mmol) were added. After stirring at 50 °C for 24 h, the slurry was diluted by  $Et_2O$  (2 mL) and  $PhI(OAc)_2$  (1.0 mmol) was added to the solution in glove box filled for nitrogen. The reaction mixture was stirred at RT for 12 h and then quenched by 1 N HCl aq. (5 mL) and  $CH_2Cl_2$  (2 mL). The solution was extracted with  $CH_2Cl_2$  (2 mL x 3) and the collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography.

# Experimental Procedure for the oxymetalation of 2-alkynylbenzoate followed by palladium catalyzed cross coupling (Table 3)

In glove box filled for nitrogen, to a sealed vial,  $InBr_3$  (0.5 mmol), toluene (1 mL) and methyl 2-ethynylbenzoate **1a** (0.5 mmol) were added. After stirring at 50 °C for 24 h, Pd<sub>2</sub>dba<sub>3</sub> (0.025 mmol), LiCl (1.0 mmol), organohalide **27** (1.0 mmol) and solvent (NMP or HMPA, 2.5 mL) was added to the slurry in glove box filled for nitrogen. The mixture was stirred at RT or 50 °C for 24 h and then quenched by water (10 mL) and Et<sub>2</sub>O (5 mL). The solution was extracted with Et<sub>2</sub>O (10 mL x 3) and the collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography.

### Monitoring of Oxyindation by <sup>1</sup>H NMR Spectroscopy

(1) at - 30 °C



Fig. S1 Monitoring of oxyindation at - 30 °C using <sup>1</sup>H NMR (400 MHz, in CDCl<sub>3</sub>) spectroscopy.

When methyl 2-alkynylbenzoate 1a was mixed with InI<sub>3</sub> in CDCl<sub>3</sub> at - 30 °C, no reaction occurred.

(2) at - 5 °C



**Fig. S2** Monitoring of oxyindation at - 5 °C using <sup>1</sup>H NMR (400 MHz, in CDCl<sub>3</sub>) spectroscopy. A small amount of zwitterion intermediate **3** was observed.

### The Effect of O-alkyl Groups on the Elimination of Alkyl Halide

As shown in Eq. 2, elimination of alkyl halide would be rate determining step. Accordingly, the reactivity of *O*-alkyl esters toward the elimination step was compared. Oxyindation of three types of alkyl esters was conducted in a short reaction time and the yields of alkyl halides and conversions of 2-alkynylbenzoate **1** were investigated by <sup>1</sup>H NMR (Table S1). The yield of zwitterion intermediate cannot be determined due to its insolubility to toluene. In all cases, fast conversion of **1** and formation of zwitterion species were observed. Methyl ester **1a** and ethyl ester **1l** gave MeI and EtI in 40% and 10% yields, respectively (entries 1 and 2). No alkyl halide was obtained in the case of isopropyl ester **1m** in spite of long reaction time (entry 3). These results indicated that the steric hindrance of *O*-alkyl group inhibited the elimination of alkyl halide.

#### Table S1 The effect of O-alkyl groups on the elimination of alkyl halide<sup>[a]</sup>

1a, 1l-1m (	O OR	+ InI <sub>3</sub> <sup>—</sup> (1.0 equiv) /)	toluen 50 °C, t	$e-d_8$ time $\left[ \begin{array}{c} O^{R} \\ I^{+} \\ I^{-} \\$		+ RI
	entry	R	time	NMR Yield (%) of RI	Conversion (%) of <b>1</b>	
	1	Me ( <b>1a</b> )	1 h	40	94	
	2	Et ( <b>1I</b> )	1 h	10	83	
	3	<sup>i</sup> Pr ( <b>1m</b> )	3 h	0	100	

[a] 1 (0.5 mmol), InI<sub>3</sub> (0.5 mmol), toluene-*d*<sub>8</sub> (1 mL), 50 °C.

## Product



A two necked flask was dried over and filled with nitrogen. To this flask, the InI<sub>3</sub> (4.98 g, 10.0 mmol), toluene (20 mL) and methyl 2-ethynylbenzoate **1a** (1.60 g, 10.0 mmol) were added. The solution was stirred at 50 °C for 21 h and the reaction mixture was quenched by water (30 mL). This solution was extracted with  $CH_2Cl_2$  (30 mL x 3) and the collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 90:10, column length 11 cm, diameter 26 mm, spherical silica gel) to give the product (1.14 g, 78%). The NMR data was agreement with the literature<sup>41</sup>.

(24a) 4-iodo-1*H*-isochromen-1-one



To a solution of  $InI_3$  (0.255 g, 0.515 mmol) in toluene (1 mL), methyl 2-ethynylbenzoate (0.0824 g, 0.514 mmol) was added. The mixture was stirred at 50 °C for 24 h. Et<sub>2</sub>O (1 mL) and PhI(OAc)<sub>2</sub> (0.32 g, 0.993 mmol) was added to the reaction mixture at RT. After stirring at RT for 12 h, the mixture was quenched by  $CH_2Cl_2$  (2 mL) and 1 N HCl aq. (5 mL), and then extracted with  $CH_2Cl_2$  (2 mL x 3). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 90:10, column length 11 cm, diameter 26 mm, spherical

silica gel) to give the product (0.090 g, 64%). IR: (KBr) 1718 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 8.26 (d, J = 7.7 Hz, 1H, 8-H), 7.82 (t, J = 7.6 Hz, 1H, 6-H), 7.59 (m, 3H, 2-H, 5-H, 7-H); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 161.0 (s, C-1), 147.4 (d, C-2), 136.3 (s, C-4), 135.6 (d, C-8), 130.1 (d), 130.0 (d), 129.7 (d), 121.9 (s, C-9), 75.9 (s, C-3); MS: (EI, 70 eV) *m/z* 272 (M<sup>+</sup>, 100), 244 (89), 89 (97), 63 (26); HRMS: (CI, 70 eV) Calculated (C<sub>9</sub>H<sub>6</sub>IO<sub>2</sub>) 272.9412 [M + H]<sup>+</sup> Found 272.9415.

(25a) 4-bromo-1*H*-isochromen-1-one



To a solution of InI<sub>3</sub> (0.255 g, 0.515 mmol) in toluene (1 mL), methyl 2-ethynylbenzoate (0.0820 g, 0.512 mmol) was added. The mixture was stirred at 50 °C for 24 h. Et<sub>2</sub>O (1 mL) and PhI(OAc)<sub>2</sub> (0.335 g, 1.04 mmol) was added to the reaction mixture at RT. After stirring at RT for 12 h, the mixture was quenched by CH<sub>2</sub>Cl<sub>2</sub> (2 mL) and 1 N HCl aq. (5 mL), and then extracted with CH<sub>2</sub>Cl<sub>2</sub> (2 mL x 3). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 90:10, column length 11 cm, diameter 26 mm, spherical silica gel) to give the product (0.0845 g, 73%). IR : (KBr) 1736 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 8.31 (d, *J* = 7.5 Hz, 1H, 8-H), 7.85 (t, *J* = 7.5 Hz, 1H, 6-H), 7.74 (d, *J* = 7.5 Hz, 1H, 5-H), 7.62 (t, *J* = 7.5 Hz, 1H, 7-H), 7.51 (s, 1H, 2-H); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 160.8 (s, C-1), 143.0 (d, C-2), 135.4 (d, C-6), 135.0 (s, C-4), 130.1 (d) 130.0 (d) 125.8 (d, C-5), 121.6 (s, C-9) 103.6 (s, C-3); MS: (EI, 70 eV) *m/z* 224 (M<sup>+</sup>, 55), 226 (M<sup>+</sup> + 2, 54), 196 (59), 89 (100), 63 (22); HRMS: (CI, 70 eV) Calculated (C<sub>9</sub>H<sub>6</sub>BrO<sub>2</sub>) 224.9551 [M + H]<sup>+</sup> Found 224.9548.

(24b) 4-iodo-7-nitro-1*H*-isochromen-1-one



To a solution of  $InI_3$  (0.248 g, 0.500 mmol) in toluene (1 mL), methyl 2-ethynyl-5-nitrobenzoate (0.105 g, 0.512 mmol) was added. The mixture was stirred at 50 °C for 24 h. Et<sub>2</sub>O (1 mL) and PhI(OAc)<sub>2</sub> (0.350 g, 1.09 mmol) was added to the reaction mixture at RT. After stirring at RT for 12 h, the mixture was quenched by  $CH_2Cl_2$  (2 mL) and 1 N HCl aq. (5 mL), and then extracted with  $CH_2Cl_2$  (2 mL x 3). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 80:20, column length 11 cm, diameter 26 mm, spherical

silica gel) to give the product (0.0978 g, 61%). This product was recrystallized from CH<sub>2</sub>Cl<sub>2</sub> and hexane to give a single crystal and the structure was determined by X-ray crystallographic analysis (CCDC 1576343). IR: (KBr) 1720 (C=O) cm<sup>-1</sup>, 1342 (NO<sub>2</sub>) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 9.10 (d, J = 2.7 Hz, 1H, 8-H), 8.61 (dd, J = 9.2, 2.7 Hz, 1H, 6-H), 7.84 (d, J = 9.2 Hz, 1H, 5-H), 7.76 (s, 1H, 2-H); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 159.1 (s, C-1), 150.4 (d, C-2), 147.8 (s, C-7), 141.3 (s, C-4), 132.0 (d, C-5), 129.5 (d, C-6), 125.7 (d, C-8), 122.5 (s, C-9), 73.6 (s, C-3); MS: (EI, 70 eV) m/z 317 (M, 100), 289 (68); HRMS: (CI, 70 eV) Calculated (C<sub>9</sub>H<sub>5</sub>NO<sub>4</sub>I) 317.9263 [M + H]<sup>+</sup> Found: 317.9263.

(24c) methyl 4-iodo-1-oxo-1H-isochromene-6-carboxylate



To a solution of InI<sub>3</sub> (0.255 g, 0.515 mmol) in toluene (1 mL), dimethyl 2-ethynylterephthalate (0.11 g, 0.504 mmol) was added. The mixture was stirred at 50 °C for 24 h. Et<sub>2</sub>O (1 mL) and PhI(OAc)<sub>2</sub> (0.330 g, 1.02 mmol) was added to the reaction mixture at RT. After stirring at RT for 12 h, the mixture was quenched by CH<sub>2</sub>Cl<sub>2</sub> (2 mL) and water (5 mL), and then extracted with CH<sub>2</sub>Cl<sub>2</sub> (2 mL x 3). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 90:10, column length 11 cm, diameter 26 mm, spherical silica gel) to give the product (0.1171 g, 70%). IR: (KBr) 1736 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 8.35 (d, *J* = 8.0 Hz, 1H, 8-H), 8.27 (d, *J* = 1.2 Hz, 1H, 5-H), 8.20 (dd, *J* = 8.0, 1.2 Hz, 1H, 7-H), 7.64 (s, 1H, 2-H), 4.02 (s, 3H, OMe); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 165.3 (s, C-10), 160.3 (s, C-1), 148.1 (d, C-2), 136.7 (s), 136.4 (s), 131.6 (d, C-5), 130.5 (d, C-8), 130.0 (d, C-7), 125.0 (s), 75.2 (s, C-3), 52.9 (q, OMe); MS: (EI, 70 eV) m/z 330 (M, 100), 302 (71), 147 (82); HRMS: (CI, 70 eV) Calculated (C<sub>11</sub>H<sub>8</sub>O<sub>4</sub>I) 330.9467 [M + H]<sup>+</sup> Found: 330.9469.

(24d) 4-iodo-7-methyl-1*H*-isochromen-1-one



To a solution of  $InI_3$  (0.260 g, 0.525 mmol) in toluene (1 mL), methyl 2-ethynyl-5-methylbenzoate (0.087 g, 0.500 mmol) was added. The mixture was stirred at 50 °C for 24 h. Et<sub>2</sub>O (1 mL) and PhI(OAc)<sub>2</sub> (0.335 g, 1.04 mmol) was added to the reaction mixture at RT. After stirring at RT for 12 h, the mixture

was quenched by CH<sub>2</sub>Cl<sub>2</sub> (2 mL) and 1 N HCl aq. (5 mL), and then extracted with CH<sub>2</sub>Cl<sub>2</sub> (2 mL x 3). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 90:10, column length 11 cm, diameter 26 mm, spherical silica gel) to give the product (0.0774 g, 54%). IR: (KBr) 1751 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 8.07 (s, 1H, 8-H), 7.61 (d, J = 8.2 Hz, 1H, 6-H), 7.54 (s, 1H, 2-H), 7.48 (d, J = 8.2 Hz, 1H, 5-H), 2.50 (s, 3H, 10-H); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 161.3 (s, C-1), 146.6 (d, C-2), 140.3 (s, C-7), 136.7 (d, C-6), 134.0 (s, C-4), 130.0 (d), 129.8 (d), 121.8 (s, C-9), 75.9 (s, C-3), 21.2 (q, C-10); MS: (EI, 70 eV) m/z 286 (M, 100), 258 (66), 103 (88); HRMS: (CI, 70 eV) Calculated (C<sub>10</sub>H<sub>8</sub>O<sub>2</sub>I) 286.9569 [M + H]<sup>+</sup> Found: 286.9567.

(24e) 7-(4-(tert-butyl)phenyl)-4-iodo-1H-isochromen-1-one



To a solution of InI<sub>3</sub> (0.259 g, 0.523 mmol) in toluene (1 mL), methyl 4'-(*tert*-butyl)-4-ethynyl-[1,1'biphenyl]-3-carboxylate (0.1453 g, 0.497 mmol) was added. The mixture was stirred at 50 °C for 24 h. Et<sub>2</sub>O (1 mL) and PhI(OAc)<sub>2</sub> (0.335 g, 1.04 mmol) was added to the reaction mixture at RT. After stirring at RT for 12 h, the mixture was quenched by CH<sub>2</sub>Cl<sub>2</sub> (2 mL) and 1 N HCl aq. (5 mL), and then extracted with CH<sub>2</sub>Cl<sub>2</sub> (2 mL x 3). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 90:10, column length 11 cm, diameter 26 mm, spherical silica gel) to give the product (0.109 g, 54%). IR: (KBr) 1728 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 8.51 (d, *J* = 1.9 Hz, 1H, 8-H), 8.05 (dd, *J* = 8.5, 1.9 Hz, 1H, 6-H), 7.67 (d, *J* = 8.5 Hz, 1H, 5-H), 7.64 (d, *J* = 8.7 Hz, 2H, *o*), 7.60 (s, 1H, 2-H), 7.53 (d, *J* = 8.7 Hz, 2H, *m*); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 161.3 (s, C-1), 151.7 (s, *p*), 147.1 (d, C-2), 142.5 (s, C-7), 135.6 (s, *i*), 135.0 (s, C-4), 134.0 (d, C-6), 130.7 (d, C-5), 127.7 (d, C-8), 126.8 (d, *o*), 126.1 (d, *m*), 122.3 (s, C-9), 75.7 (s, C-3), 34.6 (s, C-10), 31.3 (q, 10-Me x 3); MS: (EI, 70 eV) m/z 404 (M, 56), 389 (100); HRMS: (CI, 70 eV) Calculated (C<sub>19</sub>H<sub>18</sub>O<sub>2</sub>I) 405.0351 [M + H]<sup>+</sup> Found: 405.0352.

(25f) 4,7-dibromo-1*H*-isochromen-1-one



To a solution of InBr<sub>3</sub> (0.190 g, 0.536 mmol) in toluene (1 mL), methyl 5-bromo-2-ethynylbenzoate (0.120 g, 0.502 mmol) was added. The mixture was stirred at 50 °C for 24 h. Et<sub>2</sub>O (1 mL) and PhI(OAc)<sub>2</sub> (0.322 g, 1.00 mmol) was added to the reaction mixture at RT. After stirring at RT for 12 h, the mixture was quenched by CH<sub>2</sub>Cl<sub>2</sub> (2 mL) and 1 N HCl aq. (5 mL), and then extracted with CH<sub>2</sub>Cl<sub>2</sub> (2 mL x 3). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 90:10, column length 11 cm, diameter 26 mm, spherical silica gel) to give the product (0.071 g, 47%). IR: (KBr) 1732 (C=O) cm<sup>-1; 1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 8.43 (d, *J* = 1.8 Hz, 1H, 8-H), 7.94 (dd, *J* = 8.6, 1.8 Hz, 1H, 6-H), 7.62 (d, *J* = 8.6 Hz, 1H, 5-H), 7.53 (s, 1H, 2-H); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 159.5 (s, C-1), 143.3 (d, C-2), 138.5 (d, C-6), 133.9 (s, C-4), 132.6 (d, C-8), 127.6 (d, C-5), 123.5 (s, C-7), 122.8 (s, C-9), 102.8 (s, C-3); MS: (EI, 70 eV) m/z 306 (M + 4, 39), 304 (M + 2, 100), 302 (M, 40), 169 (65), 167 (70); HRMS: (CI, 70 eV) Calculated (C<sub>9</sub>H<sub>5</sub>Br<sub>2</sub>O<sub>2</sub>) 302.8656 [M + H]<sup>+</sup> Found: 302.8654.

(24g) 7-chloro-4-iodo-1*H*-isochromen-1-one



To a solution of InI<sub>3</sub> (0.256 g, 0.516 mmol) in toluene (1 mL), methyl 5-chloro-2-ethynylbenzoate (0.098 g, 0.505 mmol) was added. The mixture was stirred at 50 °C for 24 h. Et<sub>2</sub>O (1 mL) and PhI(OAc)<sub>2</sub> (0.321 g, 0.997 mmol) was added to the reaction mixture at RT. After stirring at RT for 12 h, the mixture was quenched by CH<sub>2</sub>Cl<sub>2</sub> (2 mL) and 1 N HCl aq. (5 mL), and then extracted with CH<sub>2</sub>Cl<sub>2</sub> (2 mL x 3). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 90:10, column length 11 cm, diameter 26 mm, spherical silica gel) to give the product (0.1036 g, 67%). IR: (KBr) 1724 (C=O) cm<sup>-1; 1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 8.24 (d, J = 2.3 Hz, 1H, 8-H), 7.75 (dd, J = 8.7, 2.3 Hz, 1H, 6-H), 7.59 (s, 1H, 2-H), 7.57 (d, J = 8.7 Hz, 1H, 5-H); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 159.9 (s, C-1), 147.6 (d, C-2), 135.78 (d, C-6), 135.76 (s, C-7), 134.9 (s, C-4), 131.8 (d, C-5), 129.4 (d, C-8), 122.9 (s, C-9), 74.6 (s, C-3); MS: (EI, 70 eV) m/z 308 (M + 2, 35), 306 (M, 100), 278 (64), 123 (84); HRMS: (CI, 70 eV) Calculated (C<sub>9</sub>H<sub>5</sub>ClO<sub>2</sub>I) 306.9023 [M + H]<sup>+</sup> Found: 306.9022.

(24h) 7-fluoro-4-iodo-1H-isochromen-1-one



To a solution of InI<sub>3</sub> (0.255 g, 0.515 mmol) in toluene (1 mL), methyl 5-fluoro-2-ethynylbenzoate (0.0913 g, 0.512 mmol) was added. The mixture was stirred at 50 °C for 24 h. Et<sub>2</sub>O (1 mL) and PhI(OAc)<sub>2</sub> (0.330 g, 1.02 mmol) was added to the reaction mixture at RT. After stirring at RT for 12 h, the mixture was quenched by CH<sub>2</sub>Cl<sub>2</sub> (2 mL) and 1 N HCl aq. (5 mL), and then extracted with CH<sub>2</sub>Cl<sub>2</sub> (2 mL x 3). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 90:10, column length 11 cm, diameter 26 mm, spherical silica gel) to give the product (0.091 g, 61%). IR: (KBr) 1728 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.93 (d, <sup>3</sup>*J*<sub>HF</sub> = 8.2 Hz, 1H, 8-H), 7.64 (dd, *J* = 8.9 Hz, <sup>4</sup>*J*<sub>HF</sub> = 5.1 Hz, 1H, 5-H), 7.56 (s, 1H, 2-H), 7.52 (dd, *J* = 8.9 Hz, <sup>3</sup>*J*<sub>HF</sub> = 8.2 Hz, 1H, 6-H); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 162.6 (s, d, <sup>1</sup>*J*<sub>CF</sub> = 252 Hz, C-7), 160.1 (s, C-1), 146.6 (d, C-2), 132.9 (s, d, <sup>4</sup>*J*<sub>CF</sub> = 2.5 Hz, C-4), 132.7 (d, d, <sup>3</sup>*J*<sub>CF</sub> = 7.4 Hz, C-5), 123.6 (d, d, <sup>2</sup>*J*<sub>CF</sub> = 22.9 Hz, C-6), 123.3 (s, d, <sup>3</sup>*J*<sub>CF</sub> = 8.2 Hz, C-9), 115.5 (d, d, <sup>2</sup>*J*<sub>CF</sub> = 23.8 Hz, C-8), 74.5 (s, C-3); MS: (EI, 70 eV) m/z 290 (M, 96), 262 (55), 107 (100); HRMS: (CI, 70 eV) Calculated (C<sub>9</sub>H<sub>5</sub>FO<sub>2</sub>I) 290.9318 [M + H]<sup>+</sup> Found: 290.9315.

### (24i) 4-iodo-3-phenyl-1*H*-isochromen-1-one



To a solution of GaI<sub>3</sub> (0.230 g, 0.511 mmol) in toluene (1 mL), methyl 2-(phenylethynyl)benzoate (0.116 g, 0.491 mmol) was added. The mixture was stirred at 50 °C for 24 h. Et<sub>2</sub>O (2 mL) and PhI(OAc)<sub>2</sub> (0.322 g, 1.00 mmol) was added to the reaction mixture at RT. After stirring at RT for 12 h, the mixture was quenched by CH<sub>2</sub>Cl<sub>2</sub> (2 mL) and 1 N HCl aq. (5 mL), and then extracted with CH<sub>2</sub>Cl<sub>2</sub> (2 mL x 3). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 90:10, column length 11 cm, diameter 26 mm, spherical silica gel) to give the product (0.103 g, 60%). The NMR data was agreement with the literature<sup>38</sup>.

(24j) 3-(4-fluorophenyl)-4-iodo-1H-isochromen-1-one



To a solution of GaI<sub>3</sub> (0.231 g, 0.513 mmol) in toluene (1 mL), methyl 2-((4-fluorophenyl)ethynyl)benzoate (0.129 g, 0.507 mmol) was added. The mixture was stirred at 50 °C for 24 h. Et<sub>2</sub>O (2 mL) and PhI(OAc)<sub>2</sub> (0.330 g, 1.02 mmol) was added to the reaction mixture at RT. After stirring at RT for 12 h, the mixture was quenched by CH<sub>2</sub>Cl<sub>2</sub> (2 mL) and 1 N HCl aq. (5 mL), and then extracted with CH<sub>2</sub>Cl<sub>2</sub> (2 mL x 3). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 90:10, column length 11 cm, diameter 26 mm, spherical silica gel) to give the product (0.104 g, 56%). IR: (KBr) 1728 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CD<sub>2</sub>Cl<sub>2</sub>) 8.13 (d, *J* = 8.2 Hz, 1H, 8-H), 7.77 (d, *J* = 8.2 Hz, 1H, 5-H), 7.72 (dd, *J* = 8.2, 7.7 Hz, 1H, 6-H), 7.60 (dd, *J* = 8.6 Hz, <sup>4</sup>*J*<sub>HF</sub> = 5.4 Hz, 1H, *o*), 7.48 (dd, *J* = 8.2, 7.7 Hz, 1H, 7-H), 7.09 (dd, *J* = 8.6 Hz, <sup>3</sup>*J*<sub>HF</sub> = 9.1 Hz, 1H, m); <sup>13</sup>C NMR: (100 MHz, CD<sub>2</sub>Cl<sub>2</sub>) 163.9 (s, d, <sup>1</sup>*J*<sub>CF</sub> = 250 Hz, *p*), 161.5 (s, C-1), 154.1 (s, C-2), 138.3 (s, C-4), 136.1 (d, C-6), 132.6 (d, d, <sup>3</sup>*J*<sub>CF</sub> = 8.6 Hz, *o*), 132.0 (s, d, <sup>4</sup>*J*<sub>CF</sub> = 3.8 Hz, *i*), 131.9 (d, C-5), 129.9 (d, C-8), 129.8 (d, C-7), 120.6 (s. C-9), 115.6 (d, d, <sup>2</sup>*J*<sub>CF</sub> = 11.4 Hz, *m*), 77.0 (s, C-3); MS: (EI, 70 eV) m/z 366 (M, 100), 338 (57), 211 (60), 183 (65), 123 (50), 95 (48); HRMS: (CI, 70 eV) Calculated (C<sub>15</sub>H<sub>9</sub>FIO<sub>2</sub>) 366.9631 [M+H]<sup>+</sup> Found: 366.9630.

(25k) 4-bromo-3-butyl-1*H*-isochromen-1-one



To a solution of GaBr<sub>3</sub> (0.169 g, 0.546 mmol) in toluene (1 mL), methyl 2-(hex-1-yn-1-yl)benzoate (0.110 g, 0.509 mmol) was added. The mixture was stirred at 50 °C for 24 h. Et<sub>2</sub>O (2 mL) and PhI(OAc)<sub>2</sub> (0.322 g, 1.00 mmol) was added to the reaction mixture at RT. After stirring at RT for 12 h, the mixture was quenched by  $CH_2Cl_2$  (2 mL) and 1 N HCl aq. (5 mL), and then extracted with  $CH_2Cl_2$  (2 mL x 3). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 90:10, column length 11 cm, diameter 26 mm, spherical silica gel) to give the product (0.0852 g, 60%). The NMR data was agreement with the

literature<sup>42</sup>.

(28aa) 4-phenyl-1*H*-isochromen-1-one



To a solution of InBr<sub>3</sub> (0.185 g, 0.522 mmol) in toluene (1 mL), methyl 2-ethynylbenzoate (0.082 g, 0.512 mmol) was added. The mixture was stirred at 50 °C for 24 h. Pd<sub>2</sub>dba<sub>3</sub> (0.026 g, 0.0284 mmol), LiCl (0.044 g, 1.04 mmol), iodebenzene (0.255 g, 1.25 mmol) and NMP (2.5 mL) was added to the reaction mixture at RT. After stirring at 50 °C for 24 h, the mixture was quenched by Et<sub>2</sub>O (5 mL) and water (10 mL), and then extracted with Et<sub>2</sub>O (10 mL x 3). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 90:10, column length 11 cm, diameter 26 mm, spherical silica gel) to give the product (0.0922 g, 81%). The NMR data was agreement with the literature<sup>43</sup>.

(28ab) 4-(4-methoxyphenyl)-1H-isochromen-1-one



To a solution of InBr<sub>3</sub> (0.179 g, 0.505 mmol) in toluene (1 mL), methyl 2-ethynylbenzoate (0.083 g, 0.518 mmol) was added. The mixture was stirred at 50 °C for 24 h. Pd<sub>2</sub>dba<sub>3</sub> (0.03 g, 0.033 mmol), LiCl (0.059 g, 1.39 mmol), 1-iodo-4-methoxybenzene (0.234 g, 1.0 mmol) and NMP (2.5 mL) was added to the reaction mixture at RT. After stirring at 50 °C for 24 h, the mixture was quenched by Et<sub>2</sub>O (5 mL) and water (10 mL), and then extracted with Et<sub>2</sub>O (10 mL x 3). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 90:10, column length 11 cm, diameter 26 mm, spherical silica gel) to give the product (0.0931 g, 71%). IR: (KBr) 1728 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 8.39 (d, *J* = 8.0 Hz, 1H, 8-H), 7.70 (t, *J* = 8.0 Hz, 1H, 6-H), 7.55 (t, *J* = 8.0 Hz, 1H, 7-H), 7.41 (d, *J* = 8.0 Hz, 1H, 5-H), 7.32 (d, *J* = 8.7 Hz, 2H, *o*), 7.23 (s, 1H, 2-H), 7.02 (d, *J* = 8.7 Hz, 2H, *m*); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>)

162.1 (s, C-1), 159.7 (s, *p*), 142.0 (d, C-2), 137.1 (s, C-4), 134.6 (d, C-6), 130.9 (d, *o*), 130.0 (d, C-8), 128.4 (d, C-7), 125.1 (s, *i*), 124.6 (d, C-5), 121.3 (s, C-9), 120.2 (s, C-3), 114.2 (d, *m*), 55.3 (q, OMe); MS: (EI, 70 eV) m/z 252 (M, 100), 224 (36), 181 (31), 152 (33); HRMS: (CI, 70 eV) Calculated (C<sub>16</sub>H<sub>13</sub>O<sub>3</sub>) 253.0865 [M + H]<sup>+</sup> Found: 253.0861.

(28ac) 4-(1-oxo-1H-isochromen-4-yl)benzonitrile



To a solution of InBr<sub>3</sub> (0.188 g, 0.530 mmol) in toluene (1 mL), methyl 2-ethynylbenzoate (0.081 g, 0.506 mmol) was added. The mixture was stirred at 50 °C for 24 h. Pd<sub>2</sub>dba<sub>3</sub> (0.026 g, 0.0284 mmol), LiCl (0.0511 g, 1.21 mmol), 1-iodo-4-cyanobenzene (0.250 g, 1.09 mmol) and NMP (2.5 mL) was added to the reaction mixture at RT. After stirring at 50 °C for 24 h, the mixture was quenched by Et<sub>2</sub>O (5 mL) and water (10 mL), and then extracted with Et<sub>2</sub>O (10 mL x 3). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 90:10, column length 11 cm, diameter 26 mm, spherical silica gel) to give the product (0.090 g, 72%). IR: (KBr) 2225 (CN) cm<sup>-1</sup>, 1720 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 8.42 (d, *J* = 7.8 Hz, 1H, 8-H), 7.81 (d, *J* = 7.3 Hz, 2H, *o*), 7.74 (t, *J* = 7.8 Hz, 1H, 6-H), 7.61 (t, *J* = 7.8 Hz, 1H, 7-H), 7.56 (d, *J* = 7.3 Hz, 2H, *m*), 7.33 (d, *J* = 7.8 Hz, 1H, 5-H), 7.30 (s, 1H, 2-H); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 161.4 (s, C-1), 142.8 (d, C-2), 138.1 (s, C-3), 135.6 (s, C-4), 135.0 (d, C-6), 132.7 (d, *o*), 130.5 (d, *m*), 130.4 (d, C-8), 129.1 (d, C-7), 124.0 (d, C-5), 121.3 (s, C-9), 119.3 (s, *i*), 118.3 (s, *p*), 112.5 (s, CN); MS: (EI, 70 eV) m/z 247 (M, 63), 219 (100), 190 (66); HRMS: (CI, 70 eV) Calculated (C<sub>16</sub>H<sub>10</sub>NO<sub>2</sub>) 248.0712 [M + H]<sup>+</sup> Found: 248.0710.

(28ad) 4-benzoyl-1H-isochromen-1-one



To a solution of InBr<sub>3</sub> (0.180 g, 0.508 mmol) in toluene (1 mL), methyl 2-ethynylbenzoate (0.0824 g, 0.514 mmol) was added. The mixture was stirred at 50 °C for 24 h. Pd<sub>2</sub>dba<sub>3</sub> (0.027 g, 0.0295 mmol), LiCl (0.052 g, 1.23 mmol), benzoyl chloride (0.142 g, 1.01 mmol) and NMP (2.5 mL) was added to the reaction mixture at RT. After stirring at 50 °C for 24 h, the mixture was quenched by Et<sub>2</sub>O (5 mL) and water (10 mL), and then extracted with Et<sub>2</sub>O (10 mL x 3). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 90:10, column length 11 cm, diameter 26 mm, spherical silica gel) to give the product (0.1058 g, 82%). IR: (KBr) 1735 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 8.39 (d, *J* = 8.0 Hz, 1H, 8-H), 8.16 (d, *J* = 8.2 Hz, 1H, 5-H), 7.88 (d, *J* = 8.5 Hz, 2H, *o*), 7.82 (t, *J* = 8.2 Hz, 1H, 6-H), 7.67 (s, 1H, 2-H), 7.69-7.60 (m, 2H, *p* and 7-H), 7.53 (t, *J* = 8.0 Hz, 2H, *m*); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 192.4 (s, C-10), 160.5 (s, C-1), 151.7 (d, C-2), 137.6 (s, *i*), 135.3 (d, C-6), 133.9 (s, C-4), 133.5 (d), 129.9 (d, C-8), 129.8 (d, *o*), 129.3 (d), 128.7 (d, *m*), 125.2 (d, C-5), 120.7 (s, C-9), 117.7 (s, C-3); MS: (EI, 70 eV) m/z 250 (M, 97), 249 (100), 105 (78), 77 (52); HRMS: (CI, 70 eV) Calculated (C<sub>16</sub>H<sub>11</sub>O<sub>3</sub>) 251.0708 [M + H]<sup>+</sup> Found: 251.0709.

(28ae) 4-(4-chlorobenzoyl)-1H-isochromen-1-one



To a solution of InBr<sub>3</sub> (0.179 g, 0.505 mmol) in toluene (1 mL), methyl 2-ethynylbenzoate (0.080 g, 0.499 mmol) was added. The mixture was stirred at 50 °C for 24 h. Pd<sub>2</sub>dba<sub>3</sub> (0.026 g, 0.0284 mmol), LiCl (0.049 g, 1.16 mmol), 4-chlorobenzoyl chloride (0.180 g, 1.03 mmol) and NMP (2.5 mL) was added to the reaction mixture at RT. After stirring at 50 °C for 24 h, the mixture was quenched by Et<sub>2</sub>O (5 mL) and water (10 mL), and then extracted with Et<sub>2</sub>O (10 mL x 3). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 90:10, column length 11 cm, diameter 26 mm, spherical silica gel) to give the product (0.091 g, 64%). This product was recrystallized from CH<sub>2</sub>Cl<sub>2</sub> to give a single crystal and the structure was determined by X-ray crystallographic analysis (CCDC 1576344). IR: (KBr) 1763 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 8.37 (d, *J* = 8.0 Hz, 1H, 8-H), 8.10 (d, *J* = 8.0 Hz, 1H, 5-H), 7.86-7.77 (m, 3H, *o* and 6-H), 7.65 (s, 1H, 2-H), 7.63 (t, *J* = 8.0 Hz, 1H, 7-H), 7.50 (d, *J* = 8.7 Hz, 2H, *m*); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 191.2 (s, C-10), 160.3 (s, C-1), 151.5 (d, C-2), 140.1 (s), 136.0 (s), 135.4

(d, C-6), 133.7 (s, C-4), 131.2 (d, *o*), 130.0 (d, C-8), 129.5 (d, C-7), 129.1 (d, *m*), 125.2 (d, C-5), 120.8 (s, C-9), 117.6 (s, C-3); MS: (EI, 70 eV) m/z 286 (M +2, 32), 284 (M, 93), 283 (49), 249 (40), 139 (100), 111 (55); HRMS: (CI, 70 eV) Calculated (C<sub>16</sub>H<sub>10</sub>ClO<sub>3</sub>) 285.0318 [M + H]<sup>+</sup> Found: 285.0318.

(28af) 4-isobutyryl-1*H*-isochromen-1-one



To a solution of InBr<sub>3</sub> (0.185 g, 0.522 mmol) in toluene (1 mL), methyl 2-ethynylbenzoate (0.0857 g, 0.535 mmol) was added. The mixture was stirred at 50 °C for 24 h. Pd<sub>2</sub>dba<sub>3</sub> (0.030 g, 0.0328 mmol), LiCl (0.043 g, 1.01 mmol), isobutyryl chloride (0.115 g, 1.08 mmol) and HMPA (2.5 mL) was added to the reaction mixture at RT. After stirring at RT for 24 h, the mixture was quenched by Et<sub>2</sub>O (5 mL) and water (10 mL), and then extracted with Et<sub>2</sub>O (10 mL x 3). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 80:20, column length 11 cm, diameter 26 mm, spherical silica gel) to give the product (0.070 g, 61%). IR: (KBr) 1739 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 8.47 (d, *J* = 8.2 Hz, 1H, 5-H), 8.34 (dd, *J* = 7.8, 1.4 Hz, 1H, 8-H), 8.03 (s, 1H, 2-H), 7.81 (ddd, *J* = 8.2, 7.8, 1.4 Hz, 1H, 6-H), 7.60 (t, *J* = 7.8 Hz, 1H, 7-H), 3.29-3.17 (m, 1H, 11-H), 1.25 (d, *J* = 6.9 Hz, 6H, 11-Me<sub>2</sub>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 160.6 (s, C-10), 150.5 (d, C-2), 135.4 (d, C-7), 133.7 (s, C-4), 129.9 (d, C-5), 129.2 (d, C-6), 125.8 (d, C-8), 120.9 (s, C-9), 117.1 (s, C-3), 37.8 (d, C-11), 19.0 (q, 11-Me<sub>2</sub>); MS: (EI, 70 eV) m/z 216 (M, 49), 173 (100), 89 (52); HRMS: (CI, 70 eV) Calculated (C<sub>13</sub>H<sub>13</sub>O<sub>3</sub>) 217.0865 [M + H]<sup>+</sup> Found: 217.0863.

(28ag) 4-benzyl-1*H*-isochromen-1-one



To a solution of InBr<sub>3</sub> (0.188 g, 0.530 mmol) in toluene (1 mL), methyl 2-ethynylbenzoate (0.0824 g, 0.514 mmol) was added. The mixture was stirred at 50 °C for 24 h. Pd<sub>2</sub>dba<sub>3</sub> (0.0223 g, 0.0242 mmol),

LiCl (0.043 g, 1.01 mmol), benzyl bromide (0.176 g, 1.03 mmol) and HMPA (2.5 mL) was added to the reaction mixture at RT. After stirring at 50 °C for 24 h, the mixture was quenched by Et<sub>2</sub>O (5 mL) and water (10 mL), and then extracted with Et<sub>2</sub>O (10 mL x 3). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 90:10, column length 11 cm, diameter 26 mm, spherical silica gel) to give the product (0.062 g, 51%). IR: (KBr) 1720 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 8.34 (d, J = 8.2 Hz, 1H, 8-H), 7.67 (dd, J = 8.2, 7.8 Hz, 1H, 6-H), 7.51 (dd, J = 8.2, 7.8 Hz, 1H, 7-H), 7.44 (d, J = 8.2 Hz, 1H, 5-H), 7.35-7.28 (m, 2H, *m*), 7.28-7.21 (m, 3H, *o* and *p*), 7.11 (s, 1H, 2-H), 3.93 (s, 2H, 10-H<sub>2</sub>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 162.4 (s, C-1), 142.9 (d, C-2), 137.7 (s, *i*), 136.7 (s, C-4), 134.6 (d, C-6), 130.1 (d, C-8), 128.8 (d), 128.4 (d), 128.3 (d, C-7), 126.8 (d, *p*), 123.4 (d, C-5), 121.6 (s, C-9), 115.7 (s, C-3), 33.6 (t, C-10); MS: (EI, 70 eV) m/z 236 (M, 100), 208 (63), 178 (40); HRMS: (CI, 70 eV) Calculated (C<sub>13</sub>H<sub>13</sub>O<sub>3</sub>) 237.0916 [M + H]<sup>+</sup> Found: 237.0916.

(28ah) 4-(but-2-en-1-yl)-1H-isochromen-1-one



To a solution of InBr<sub>3</sub> (0.185 g, 0.522 mmol) in toluene (1 mL), methyl 2-ethynylbenzoate (0.078 g, 0.487 mmol) was added. The mixture was stirred at 50 °C for 24 h. Pd<sub>2</sub>dba<sub>3</sub> (0.026 g, 0.0284 mmol), LiCl (0.057 g, 1.34 mmol), 1-bromobut-2-ene (0.149 g, 1.10 mmol) and HMPA (2.5 mL) was added to the reaction mixture at RT. After stirring at 50 °C for 24 h, the mixture was quenched by Et<sub>2</sub>O (5 mL) and water (10 mL), and then extracted with Et<sub>2</sub>O (10 mL x 3). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 90:10, column length 11 cm, diameter 26 mm, spherical silica gel) to give the product (0.0441 g, 45%, *E/Z* = 90:10). IR: (KBr) 1728 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 8.34 (d, *J* = 7.8 Hz, 1H, 8-H), 7.76 (t, *J* = 7.8 Hz, 1H, 6-H), 7.59-7.50 (m, 2H, 5-H and 7-H), 7.21 (s 1H, 2-H), 5.73-5.50 (m, 2H, 11-H and 12-H), 3.26 (d, *J* = 4.6 Hz, 2H, 10-H<sub>2</sub>), 1.70 (d, *J* = 5.0 Hz, 3H, 13-H<sub>3</sub>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 162.6 (s, C-1), 141.8 (d, C-2), 136.9 (s, C-4), 134.5 (s, C-6), 130.0 (d, C-8), 128.4 (d), 128.2 (d), 126.9 (d), 123.1 (d, C-5), 121.5 (s, C-9), 115.9 (s, C-3), 30.3 (t, C-10), 17.9 (q, C-13); MS: (EI, 70 eV) m/z 200 (M, 100), 171 (33), 157 (39), 128 (46), 115 (30); HRMS: (CI, 70 eV) Calculated (C<sub>13</sub>H<sub>13</sub>O<sub>3</sub>) 201.0916 [M+H]<sup>+</sup> Found: 201.0913.

(30) 2-iodo-6-methoxybenzoic acid



This compound was synthesized by a reported method<sup>44</sup> and the NMR data was agreement with the literature<sup>44</sup>.

(31) 2-hydroxy-6-iodobenzoic acid



To a three necked flask, 2-iodo-6-methoxybenzoic acid (0.141 g, 0.507 mmol), CH<sub>2</sub>Cl<sub>2</sub> (4 mL) and BBr<sub>3</sub> (1 M in CH<sub>2</sub>Cl<sub>2</sub>, 1.0 mL, 1.0 mmol) were added. After stirring for 20 h at RT, the reaction mixture was evaporated and the residue was diluted with CH<sub>2</sub>Cl<sub>2</sub> (10 mL). The solution was washed with 0.5 N HCl aq. (10 mL) and the water layer was extracted by CH<sub>2</sub>Cl<sub>2</sub> (10 mL x 3). The corrected organic layer was dried (MgSO<sub>4</sub>) and the solvent was evaporated to give the product (100% NMR yield). The product was used in next step without further purification. IR: (neat) 3568 (OH) cm<sup>-1</sup>, 1635 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, acetone-*d*<sub>6</sub>) 7.62 (dd, *J* = 7.7, 0.97 Hz, 1H), 7.19 (t, *J* = 7.7 Hz, 1H, 5-H), 7.10 (dd, *J* = 7.7, 0.97 Hz, 1H); <sup>13</sup>C NMR: (100 MHz, acetone-*d*<sub>6</sub>) 168.9 (s, C-1), 158.5 (s, C-7), 133.5 (d, C-5), 132.0 (d), 123.5 (s, C-2), 116.8 (d), 93.1 (s, C-3); HRMS: (CI, 70 eV) Calculated (C<sub>7</sub>H<sub>6</sub>O<sub>3</sub>I) 264.9362 [M + H]<sup>+</sup> Found: 264.9364.

(32) methyl 2-hydroxy-6-iodobenzoate



This compound was synthesized by a reported method<sup>45</sup> and the NMR data was agreement with the literature<sup>46</sup>.

(33) methyl 2-acetoxy-6-iodobenzoate



To a three necked flask, methyl 2-hydroxy-6-iodobenzoate (0.136 g, 0.491 mmol), acetone (0.7 mL) and pyridine (0.042 mL, 0.52 mmol) were added. The mixture was cooled to -5 °C and acetyl chloride (37  $\mu$ L. 0.52 mmol) was added. After stirring for 14 h at RT, the reaction mixture was quenched by H<sub>2</sub>O (2 mL). The solution was extracted by CH<sub>2</sub>Cl<sub>2</sub> (2 mL x 2). The corrected organic layer was dried (MgSO<sub>4</sub>) and the solvent was evaporated to give the product (97% NMR yield). The product was used in next step without further purification. IR: (KBr) 1766 (C=O) cm<sup>-1</sup>, 1732 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.72 (dd, *J* = 6.8, 1.9 Hz, 1H), 7.16 (dd, *J* = 8.2, 1.9 Hz, 1H), 7.13 (dd, *J* = 8.2, 6.8 Hz, 1H, 5-H), 3.93 (s, 3H, OMe), 2.26 (s, 3H, 9-H<sub>3</sub>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 168.4 (s, 8), 166.2 (s, 1), 147.7 (s, C-7), 136.7 (d), 132.9 (s, C-2), 131.6 (d, C-5), 122.8 (d), 92.2 (s, C-3), 52.7 (q, OMe), 20.6 (q, C-9); MS: (EI, 70 eV) m/z 320 (M, 19), 278 (100), 246 (100); HRMS: (CI, 70 eV) Calculated (C<sub>10</sub>H<sub>10</sub>O<sub>4</sub>I) 320.9624 [M + H]<sup>+</sup> Found: 320.9620.

(34) methyl 2-acetoxy-6-((trimethylsilyl)ethynyl)benzoate



To a three necked flask, methyl 2-acetoxy-6-iodobenzoate (0.156 g, 0.487 mmol), NEt<sub>3</sub> (0.5 mL), PdCl<sub>2</sub>(PPh<sub>3</sub>)<sub>2</sub> (0.0072 g, 0.0103 mmol) and CuI (0.0096 g, 0.0956 mmol) were added. The mixture was cooled to 0 °C and trimethylsilylacetylene (0.0538 g. 0.548 mmol) was added. After stirring for 17 h at RT, the reaction mixture was filtered through a celite pad. The filtrate was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 90:10, column length 11 cm, diameter 26 mm, spherical silica gel) to give the product (0.1581 g, 100%). IR: (neat) 1778 (C=O) cm<sup>-1</sup>, 1736 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.41 (dd, J = 7.8, 1.8 Hz, 1H, 4-H), 7.38 (t, J = 7.8 Hz, 1H, 5-H), 7.11 (dd, J = 7.8, 1.8 Hz, 1H, 6-H), 3.90 (s, 3H, OMe), 2.29 (s, 3H, 9-H<sub>3</sub>), 0.236 (s, 9H, SiMe<sub>3</sub>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 168.9 (s, C-8), 165.5 (s, C-1), 148.2 (s, C-7), 130.7 (d), 130.6 (d), 128.9 (s, C-2), 123.3 (d, C-6), 123.2 (s, C-3), 101.3 (s, C-10), 99.5 (s, C-11), 52.3 (q, OMe), 20.8 (q, C-9), - 0.241 (q, SiMe<sub>3</sub>); MS: (EI, 70 eV) m/z 290 (M, 8), 248 (79), 233 (32), 216 (100), 203 (85); HRMS: (CI, 70 eV) Calculated (C<sub>15</sub>H<sub>19</sub>O<sub>4</sub>Si) 291.1053 [M + H]<sup>+</sup> Found: 291.1054.

(35) methyl 2-acetoxy-6-ethynylbenzoate



To a three necked flask, methyl 2-acetoxy-6-((trimethylsilyl)ethynyl)benzoate (0.0488 g, 0.168 mmol), DMF (1.7 mL) and 1 M KF aq. (0.28 mL, 0.28 mmol) were added. After stirring for 0.5 h at RT, the reaction mixture was extracted by CHCl<sub>3</sub> (5 mL). The organic layer was washed by 10% NH<sub>4</sub>Cl aq. (5 mL) and then was dried (MgSO<sub>4</sub>). The solvent was evaporated to give the product (0.0279 g, 76%). The product was used in next step without further purification. IR: (neat) 1774 (C=O) cm<sup>-1</sup>, 1732 (C=O) cm<sup>-1</sup>; <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 7.46 (dd, J = 8.0, 1.4 Hz, 1H, 4-H), 7.42 (t, J = 8.0 Hz, 1H, 5-H), 7.15 (dd, J = 8.0, 1.4 Hz, 1H, 4-H), 2.29 (s, 3H, 9-H<sub>3</sub>); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 168.8 (s, C-8), 165.3 (s, C-1), 148.2 (s, C-7), 131.1 (d), 130.9 (d), 128.9 (s, C-2), 123.8 (d, C-6), 122.3 (s, C-3), 81.8 (s, C-11), 80.2 (d, C-11), 52.5 (q, OMe), 20.7 (q, C-9); MS: (EI, 70 eV) m/z 218 (M, 7), 176 (98), 144 (100), 116 (36); HRMS: (CI, 70 eV) Calculated (C<sub>12</sub>H<sub>11</sub>O<sub>4</sub>) 219.0657 [M + H]<sup>+</sup> Found: 219.0654.

(36) 2-(8-acetoxy-1-oxo-1H-isochromen-4-yl)-2-oxoethyl acetate



To a solution of InBr<sub>3</sub> (0.114 g, 0.322 mmol) in toluene (0.6 mL), methyl 2-acetoxy-6-ethynylbenzoate (0.069 g, 0.316 mmol) was added. The mixture was stirred at 50 °C for 24 h. Pd<sub>2</sub>dba<sub>3</sub> (0.0171 g, 0.0187 mmol), LiCl (0.0284 g, 0.670 mmol), 2-chloro-2-oxoethyl acetate (0.087 g, 0.637 mmol) and HMPA (1.5 mL) was added to the reaction mixture at RT. After stirring at RT for 9 h, the mixture was quenched by Et<sub>2</sub>O (5 mL) and water (10 mL), and then extracted with Et<sub>2</sub>O (10 mL x 3). The collected organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 60:40, column length 11 cm, diameter 26 mm, spherical silica gel) to give the product (0.0420 g, 44%). The NMR data was agreement with the literature<sup>47</sup>.

### **Isolation of Organoindium Compounds**

(3) zwitterion intermediate



To a sealed vial, InI<sub>3</sub> (0.152 g, 0.307 mmol) in toluene (0.6 mL) and methyl 2-ethynylbenzoate (0.048 g, 0.30 mmol) were added. The mixture was stirred at RT for 24 h. The solvent was removed by decantation to obtain a white solid and the solid was washed by CHCl<sub>3</sub> (3 mL x 6). The residue was dried under vacuum to give the product as a white solid including a small amount of a stating ester (0.150 g, 98wt%, 73% yield). This compound was recrystallized from CH<sub>2</sub>Cl<sub>2</sub> and hexane to give a single crystal. The structure was determined by X-ray crystallographic analysis (CCDC 1579824). Characterization by NMR study was also carried out and the spectra is shown below. <sup>1</sup>H NMR: (400 MHz, CD<sub>2</sub>Cl<sub>2</sub>) 8.68 (d, J = 8.2 Hz, 1H, 8-H), 8.55 (d, J = 8.2 Hz, 1H, 5-H), 8.35 (t, J = 8.2 Hz, 1H, 6-H), 8.10 (s, 1H, 2-H), 7.97 (t, J = 8.2 Hz, 1H, 7-H), 4.68 (s, 3H, 10-H<sub>3</sub>); <sup>13</sup>C NMR: (100 MHz, CD<sub>2</sub>Cl<sub>2</sub>) 171.8 (s, C-1), 146.7 (d, C-2), 144.0 (s, C-4), 141.8 (d, C-6), 132.3 (d, C-7), 130.5 (d, C-5), 129.8 (d, C-8), 127.6 (s, C-3), 115.8 (s, C-9), 62.1 (q, C-10).

(4a) (1-oxo-1H-isochromen-4-yl)indium(III) iodide



To a sealed vial, zwitterion intermediate **3** (0.190 g, 98wt%, 0.284 mmol) in toluene (0.6 mL) was added. The mixture was stirred at 50 °C for 10 h. The solvent was removed by decantation to obtain a white solid and the solid was washed by CHCl<sub>3</sub> (3 mL x 2). The residue was dried under vacuum to give the product as a white solid (0.150 g, 100%). This compound was characterization by NMR study and the spectra is shown below. <sup>1</sup>H NMR: (400 MHz, THF- $d_8$ ) 8.26 (d, J = 7.7 Hz, 1H, 8-H), 8.09 (d, J = 7.7 Hz, 1H, 5-H), 7.75 (t, J = 7.7 Hz, 1H, 6-H), 7.53 (t, J = 7.7 Hz, 1H, 7-H), 7.17 (s, 1H, 2-H); <sup>13</sup>C NMR: (100 MHz, THF- $d_8$ ) 161.9 (s, C-1), 149.8 (d, C-2), 141.3 (s, C-4), 135.2 (d, C-6), 130.4 (d, C-8), 129.1 (d, C-7), 128.7 (d, C-5), 124.3 (s, C-9), 120.2 (s, C-3). (4b) (7-nitro-1-oxo-1H-isochromen-4-yl)indium(III) iodide



To a sealed vial, InI<sub>3</sub> (0.250 g, 0.505 mmol) in toluene (1.0 mL) and methyl 2-ethynyl-5-nitrobenzoate (0.089 g, 0.434 mmol) were added. The mixture was stirred at 50 °C for 12 h. The solvent was removed by decantation to obtain a white solid and the solid was washed by CH<sub>2</sub>Cl<sub>2</sub> (3 mL x 2). The residue was dried under vacuum to give the product as a white solid (0.218 g, 90%). This compound was recrystallized from THF and hexane to give a single crystal. The structure was determined by X-ray crystallographic analysis (CCDC 1576342). Characterization by NMR study was also carried out and the spectra is shown below. <sup>1</sup>H NMR: (400 MHz, THF-*d*<sub>8</sub>) 8.99 (s, 1H, 8-H), 8.61 (d, *J* = 7.2 Hz, 1H, 6-H), 8.25 (d, *J* = 7.2 Hz, 1H, 5-H), 7.35 (s, 1H, 2-H); <sup>13</sup>C NMR: (100 MHz, THF-*d*<sub>8</sub>) 160.8 (s, C-1), 152.7 (d, C-2), 148.1 (s, C-7), 146.5 (s, C-4), 130.2 (d), 129.4 (d), 125.7 (d, C-8), 124.8 (s, C-9), 119.3 (s, C-3).

### The energy profile of dissociation of In<sub>2</sub>I<sub>6</sub> and In<sub>2</sub>Cl<sub>6</sub>



**Reaction Process** 

Fig. S3 The energy profile of dissociation of In<sub>2</sub>I<sub>6</sub> and In<sub>2</sub>Cl<sub>6</sub>.

Coordination of two molecules of 1a to  $In_2X_6$  gives S1 or S2. The aggregation of  $In_2X_6$  was dissociated to generate 7 or S3. The all steps are exergonic to afford the complex 7 or S3 initiatively.





Fig. S4 The energy profile of intramolecular elimination of MeI.

In this pathway, oxyindation proceeds in a concerted mechanism, and then dissociation of iodide from zwitterion **3** gives complex **S4**. Elimination of MeI takes places to give **S5**. The activation energy of dealkylation step (**S4** to **S5**) is only 7.9 kcal/mol. However, the intermediate **S4** is too unstable to be generated.



The energy profile of oxyindation using InCl<sub>3</sub>

**Reaction Process** 

**Fig. S5** The energy profile of oxyindation using InCl<sub>3</sub>.
We found that oxyindation using InCl<sub>3</sub> proceeds through the same fashion with InI<sub>3</sub>. In the case of InCl<sub>3</sub>, the energy level of all structures became lower than that of the case with InI<sub>3</sub>. The remarkable difference between the cases of InCl<sub>3</sub> and InI<sub>3</sub> was the activation energy of elimination step. The elimination of MeCl (**S9** to TS2-6-*endo*) showed higher activation energy (32.7 kcal/mol) than that of the elimination of MeI (**8** to TS2-6-*endo*, 28.7 kcal/mol). This inhibition of elimination step, which is a rate determining step, was rationalized by the low nucleophilicity of Cl<sup>-</sup>, and it caused much less reactivity of InCl<sub>3</sub>.

#### The comparison of elimination steps in oxyboration





Fig. S6 The energy profile of elimination of MeCl by another zwitterion intermediate.

After 5-*exo* cyclization, two zwitterions aggregate in head to tail fashion to give the complex **S16**. The elimination step starts from **S16**. Intermolecular nucleophilic substitution of methyl group by chloride proceeds in  $S_N 2$  manner to give complex **S17** and MeCl, and then further elimination of MeCl affords target product.

#### (ii) The energy profile of elimination of MeCl by free chloride



Fig. S7 The energy profile of elimination of MeCl by free chloride.

After 5-*exo* cyclization, dissociation of chloride from zwitterion 16 provides free chloride **S18**. After the chloride move toward the methyl moiety (**S19**), elimination of MeCl proceeds to give **S20**.

40 CIBca 35.0 30 27 9 24.9 TS1-5 20 19.1 DG [kcal/mol] at 398 K 22 20 ¥ ⊿G<sup>‡</sup>= 5.8 14.3 ⊿G<sup>‡</sup>= 20.7 10 0.0 0 ⊿G<sup>‡</sup>= 27.9 23 14 -9.2 -10 TS2-5-exc +MeCl +CIBca -20 22.3 Cyclization Elimination of MeCI (intermolecular reaction with another CIBcat) -30 **Reaction Process** 

(iii) The energy profile of elimination of MeCl by [Cl<sub>2</sub>Bcat]<sup>-</sup> (the most probable path)

Fig. S8 The energy profile of elimination of MeCl by [Cl<sub>2</sub>Bcat]<sup>-</sup>.

In the case of path i, the activation energy of elimination step (S16 to S17, 21.4 kcal/mol) is higher than that of elimination in the path iii (22 to 23, 20.7 kcal/mol). In the path ii, the activation energy of dealkylation step (S19 to S20) is only 4.4 kcal/mol. However, the intermediate S18, which has naked Cl-, is too unstable to be generated. Therefore, path iii is the most probable reaction process of

oxyboration.

#### The value of electrostatic potential

The value of  $V_{\text{max}}$ ,  $V_{\text{min}}$  and the difference  $\Delta V$  was summarized in Table S2. The value of  $V_{\text{max}}$  was higher in transition state of oxyindation than that of oxyboration. However, the transition state of oxyboration showed much more negative value of  $V_{\text{min}}$  than that of oxyindation, and thus the value of difference  $\Delta V$ was small in the case of oxyindation compared with oxyboration. As the total result, the transition state of oxyindation delocalized the increasing charge more efficiently than that of oxyboration.

**Table S2** The value of  $V_{\text{max}}$ ,  $V_{\text{min}}$  and the difference  $\Delta V$  for the electrostatic potential of transition state of cyclization step

	Oxyindation	Oxyboration	
_	TS1-6-endo	TS1-5-exo	
V <sub>max</sub> (kcal/mol)	43.0	33.2	
V <sub>min</sub> (kcal/mol)	-21.3	-43.8	
⊿V (kcal/mol)	64.3	77.0	

#### **Optimization of Reaction Conditions for Oxymetalation of Internal Alkyne**

Table S3 The effect of Lewis acids on the oxymetalation of internal alkyne 1i.<sup>a</sup>



<sup>a</sup>Reaction conditions: 1i (0.5 mmol), MX<sub>3</sub> (0.5 mmol), Toluene (1 mL).

The internal alkyne **1i** was subjected to the oxyindation/protonation process to afford a target product in low yield (entries 1 and 2), but the cases using gallium salts gave high yields (entries 3 and 4).

#### 3-5. References

1 a) D. B. Kitchen, H. Decornez, J. R. Furr, J. Bajorath, *Nat. Rev. Drug Discovery* 2004, *3*, 935;
b) R. E. Ziegert, J. Toräng, K. Knepper, S. Bräse, *J. Comb. Chem.* 2005, *7*, 147; c) R. A. Ward,

J. G. Kettle, J. Med. Chem. 2011, 54, 4670; d) P. S. Mariano, Tetrahedron 1983, 39, 3879; e)
N. A. Romero, D. A. Nicewicz, Chem. Rev. 2016, 116, 10075; f) B. H. Lipshutz, Chem. Rev. 1986, 86, 795; g) A. R. Katritzky, Chem. Rev. 1989, 89, 827; h) Y.-G. Zhou, Acc. Chem. Res. 2007, 40, 1357.

- 2 a) R. Dorel, A. M. Echavarren, *Chem. Rev.* 2015, *115*, 9028; b) N. T. Patil, Y. Yamamoto, *Chem. Rev.* 2008, *108*, 3395; c) F. Alonso, I. P. Beletskaya, M. Yus, *Chem. Rev.* 2004, *104*, 3079; d)
  A. Deiters, S. F. Martin, *Chem. Rev.* 2004, *104*, 2199; e) V. Nair, C. Rajesh, A. U. Vinod, S. Bindu, A. R. Sreekanth, J. S. Mathen, L. Balagopal, *Acc. Chem. Res.* 2003, *36*, 899.
- 3 For selective examples: a) J. Liu, X. Xie, Y. Liu, Chem. Commun. 2013, 49, 11794; b) Y.-J. Feng, F.-Y. Tsai, S.-L. Huang, Y.-H. Liu, Y.-C. Lin, Eur. J. Inorg. Chem. 2014, 5406; c) S. S. Racharlawar, D. Shankar, M. V. Karkhelikar, B. Sridhar, P. R. Likhar, J. Organomet. Chem. 2014, 757, 14; d) J. J. Hirner, D. J. Faizi, S. A. Blum, J. Am. Chem. Soc. 2014, 136, 4740; e) D. Shankar, K. Jaipal, B. Sridhar, R. N. Chary, S. Prabhakar, L. Giribabu, P. R. Likhar, RSC Adv. 2015, 5, 20295; f) E. Chong, S. A. Blum, J. Am. Chem. Soc. 2015, 137, 10144; g) K. N. Tu, J. J. Hirner, S. A. Blum, Org. Lett. 2016, 18, 480; h) D. J. Faizi, N. A. Nava, M. Al-Amin, S. A. Blum, Org. Synth. 2016, 93, 228; i) D. J. Faizi, A. J. Davis, F. B. Meany, S. A. Blum, Angew. Chem. Int. Ed. 2016, 55, 14286; j) A. J. Warner, A. Churn, J. S. McGough, M. J. Ingleson, Angew. Chem. Int. Ed. 2017, 56, 354; k) N. Chaisan, W. Kaewsri, C. Thongsornkleeb, J. Tummatorn, S. Ruchirawat, Tetrahedron Lett. 2018, 59, 675.
- 4 a) Y. Kato, K. Miki, F. Nishino, K. Ohe, S. Uemura, Org. Lett. 2003, 5, 2619; b) C. P. Casey, N. A. Strotman, I. A. Guzei, Organometallics 2004, 23, 4121; c) R. Vicente, J. González, L. Riesgo, J. González, L. A. López, Angew. Chem. Int. Ed. 2012, 51, 8063; d) T. Murata, M. Murai, Y. Ikeda, K. Miki, K. Ohe, Org. Lett. 2012, 14, 2296; e) Y. Xia, S. Qu, Q. Xiao, Z.-X. Wang, P. Qu, L. Chen, Z. Liu, L. Tian, Z. Huang, Y. Zhang, J. Wang, J. Am. Chem. Soc. 2013, 135, 13502; f) J.-M. Yang, Z.-Q. Li, M.-L. Li, Q. He, S.-F. Zhu, Q.-L. Zhou, J. Am. Chem. Soc. 2017, 139, 3784.
- 5 a) K. Miki, T. Yokoi, F. Nishino, K. Ohe, S. Uemura, J. Organomet. Chem. 2002, 645, 228; b)
  K. Miki, T. Yokoi, F. Nishino, Y. Kato, Y. Washitake, K. Ohe, S. Uemura, J. Org. Chem. 2004, 69, 1557; c) K. Okamoto, M. Watanabe, A. Mashida, K. Miki, K. Ohe, Synlett 2013, 24, 1541.
- 6 a) R. C. Larock, L. W. Harrison, J. Am. Chem. Soc. 1984, 106, 4218; b) A. Nagarajan, T. R. Balasubramanian, Indian J. Chem. B, 1987, 26B, 917; c) Y. Zhu, B. Yu, Angew. Chem. Int. Ed. 2011, 50, 8329; d) Y. Tang, J. Li, Y. Zhu, Y. Li, B. Yu, J. Am. Chem. Soc. 2013, 135, 18396; e) D. J. Faizi, A. Issaian, A. J. Davis, S. A. Blum, J. Am. Chem. Soc. 2016, 138, 2126; f) B.

Akkachairin, J. Tummatorn, N. Supantanapong, P. Nimnual, C. Thongsornkleeb, S. Ruchirawat, *J. Org. Chem.* **2017**, *82*, 3727.

- 7 The metal species E was proposed as intermediate for oxymetalation/protonation sequential reactions, and these cyclic reactions showed low regioselectivity to afford mixture of 5- and 6-membered rings. a) E. Marchal, P. Uriac, B. Legouin, L. Toupet, P. v. d. Weghe, *Tetrahedron* 2007, *63*, 9979; b) B. Y.-W. Man, A. Knuhtsen, M. J. Page, B. A. Messerle, *Polyhedron* 2013, *61*, 248.
- 8 a) I. V. Alabugin, K. Gilmore, M. Manoharan, J. Am. Chem. Soc. 2011, 133, 12608. b) K. Gilmore, R. K. Mohamed, I. V. Alabugin, WIREs: Comput. Mol. Sci., 2016, 6, 487.
- 9 a) I. V. Alabugin, K. Gilmore, *Chem. Commun.* 2013, 49, 11246. b) P. W. Peterson, R. K. Mohamed, I. V. Alabugin, *Eur. J. Org. Chem.*, 2013, 2505. c) T. Yao, R. C. Larock, *J. Org. Chem.* 2003, 68, 5936.
- 10 Although some groups reported 6-endo cyclic oxymetalation using terminal alkynes, the formed 6-membered pyrylium intermediate F was trapped by a nucleophile to give not E but heterocyclic compounds or naphthalene derivatives, and F has not been isolated. a) N. Iwasawa, M. Shido, H. Kusama, J. Am. Chem. Soc. 2001, 123, 5814; b) N. Asao, T. Nogami, K. Takahashi, Y. Yamamoto, J. Am. Chem. Soc. 2002, 124, 764; c) H. Kusama, H. Funami, J. Takaya, N. Iwasawa, Org. Lett. 2004, 6, 605; d) N. T. Patil, Y. Yamamoto, J. Org. Chem. 2004, 69, 5139;
  e) H. Kusama, H. Funami, M. Shido, Y. Hara, J. Takaya, N. Iwasawa, J. Am. Chem. Soc. 2005, 127, 2709; f) H. Kusama, H. Funami, N. Iwasawa, Synthesis 2007, 13, 2014; g) R. Yanada, K. Hashimoto, R. Tokizane, Y. Miwa, H. Minami, K. Yanada, M. Ishikura, T. Takemoto, J. Org. Chem. 2008, 73, 5135; h) X.-L. Fang, R.-Y. Tang, X.-G. Zhang, P. Zhong, C.-L. Deng, J.-H. Li, J. Organomet. Chem. 2011, 696, 352; i) S. Zhu, Z. Zhang, X. Huang, H. Jiang, Z. Guo, Chem. Eur. J. 2013, 19, 4695; j) G. Mariaule, G. Newsome, R. Y. Toullec, P. Belmont, V. Michelet, Org. Lett. 2014, 16, 4570; k) J.-F. Cui, H.-M. Ko, K.-P. Shing, J.-R. Deng, N. C.-H. Lai, M.-K. Wong, Angew. Chem. Int. Ed. 2017, 56, 3074.
- 11 a) R. D. Barry, Chem. Rev. 1964, 64, 229; b) R. A. Hill in Progress in the Chemistry of Organic Natural Products, ed. W. Herz, Vol. 49, pp 1–78. c) I.-U. Rahman, M. Arfan, G. A. Khan, J. Chem. Soc. Pak. 1998, 20, 76; d) V. Rukachaisirikul, A. Rodglin, Y. Sukpondma, S. Phongpaichit, J. Buatong, J. Sakayaroj, J. Nat. Prod. 2012, 75, 853; e) M. E. Riveiro, A. Moglioni, R. Vazquez, N. Gomez, G. Facorro, L. Piehl, E. R. De Celis, C. Shayo, C. Davio, Bioorg. Med. Chem. 2008, 16, 2665; (f) L. M. Bedoya, E. Del Olmo, R. Sancho, B. Barboza, M. Beltrán, A. E. García-Cadenas, S. Sánchez-Palomino, J. L. López-Pérez, E. Muñoz, A. S. Feliciano, J. Alcamí, Bioorg. Med. Chem. Lett. 2006, 16, 4075; g) N. Agata, H. Nogi, M.

Milhollen, S. Kharbanda, D. Kufe, *Cancer Res.* 2004, 64, 8512; h) T. Nakashima, S. Hirano, N. Agata, H. Kumagai, K. Isshiki, T. Yoshioka, M. Ishizuka, K. Maeda, T. Takeuchi, *J. Antibiot.* 1999, 52, 426; i) M. Yoshikawa, E. Harada, Y. Noritoh, K. Inoue, H. Matsuda, H. Shimoda, J. Yamahara, N. Murakami, *Chem. Pharm. Bull.* 1994, 42, 2225; j) T. Furuta, Y. Fukuyama, Y. Asakawa, *Phytochemistry* 1986, 25, 517.

- 12 a) N. Panda, P. Mishra, I. Mattan, J. Org. Chem. 2016, 81, 1047; b) K. Kobayashi, W. Miyatani, M. Kuroda, Helv. Chim. Acta. 2013, 96, 2173; c) M. Lessi, T. Masini, L. Nucara, F. Bellina, R. Rossi, Adv. Synth. Catal. 2011, 353, 501; d) M. P. Pavan, M. Chakravarty, K. C. K. Swamy, Eur. J. Org. Chem. 2009, 5927; e) Z. He, A. K. Yudin, Org. Lett. 2006, 8, 5829; f) M. Chakravarty, K. C. K. Swamy, J. Org. Chem. 2006, 71, 9128; g) S. Kim, G.-J. Fan, J. Lee, J. J. Lee and D. Kim, J. Org. Chem. 2002, 67, 3127; h) B. A. Kowalczyk, Synthesis 2000, 8, 1113; i) P. Babin, J. Dunoguès, Tetrahedron Lett. 1984, 25, 4389; j) K. Beautement, J. M. Clough, Tetrahedron Lett. 1984, 25, 3025; k) O. S. Wolfbeis, Liebigs Ann. Chem. 1981, 819.
- 13 M. Kimura, I. Waki, M. Kokubo, Jpn. J. Pharmacol, 1978, 28, 693.
- 14 J. H. Lee, Y. J. Park, H. S. Kim, Y. S. Hong, K.-W. Kim, J. J. Lee, J. Antibiot., 2001, 54, 463.
- 15 a) J. Sonnenbichler, T. Kovács, Eur. J. Biochem. 1997, 246, 45; b) T. Kovács, J. Sonnenbichler, Liebigs Ann-Recl 1997, 1, 211; c) K. Nozawa, M. Yamada, Y. Tsuda, K. Kawai, S. Nakajima, Chem. Pharm. Bull. 1981, 29, 2689.
- 16 a) S. Nakajima, K. Kawai, S. Yamada, *Phytochemistry* 1976, 15, 327. b) T. Kovács, I. Sonnenbichler, J. Sonnenbichler, *Liebigs Ann-Recl* 1997, 4, 773.
- 17 a) Y. Nishimoto, H. Ueda, M. Yasuda, A. Baba, *Chem. Eur. J.* 2011, *17*, 11135; b) Y. Nishimoto,
  R. Moritoh, M. Yasuda, A. Baba, *Angew. Chem. Int. Ed.* 2009, *48*, 4577.
- 18 For selective examples: a) H. Wang, Y. Kuang, J. Wu, Asian J. Org. Chem. 2012, 1, 302; b) E. Tomás-Mendivil, F. C. Heinrich, J.-C. Ortuno, J. Starck, V. Michelet, ACS Catal. 2017, 7, 380.
- Benzopyrylium intermediate in oxymetalation was detected by NMR studies or MS sepctra: a)
   Y. Zhu, B. Yu, *Chem. Eur. J.* 2015, 21, 8771; b) B. N. Nguyen, L. A. Adrin, E. M. Barreiro, J.
   B. Brazier, P. Haycock, K. K. Hii, M. Nachtegaal, M. A. Newton, J. Szlachetko, *Organometallics* 2012, 31, 2395. Other types of zwitterion intermediate have been characterized by X-ray crystallographic analysis: c) O. A. Egorova, H. Seo, Y. Kim, D. Moon, Y. M. Rhee, K. H. Ahn, *Angew. Chem. Int. Ed.* 2011, 50, 11446; d) Y. Yu, G. Chen, L. Zhu, Y. Liao, Y. Wu and X. Huang, *J. Org. Chem.* 2016, 81, 8142.
- 20 I. Braun, A. M. Asiri, A. S. K. Hashmi, ACS Catal. 2013, 3, 1902.
- 21 T. M. Baber, G. Qadeer, G. S. Khan, N. H. Rama, W.-Y. Wong, *Acta Cryst.* 2006, *E62*, o5612o5613.

- 22 J. D. Forrester, A. Zalkin, D. H. Templeton, Inorg. Chem. 1964, 3, 63.
- 23 S<sub>N</sub>2-type elimination of MeI agrees with inhibition of the elimination step by bulky *O*-alkyl group in ester moiety (see Table S1 in Experimental Section).
- 24 The values of activation energy of cyclization step shown in Fig. 3 (6-*endo*: 39.4 kcal/mol, 5*exo*: 31.0 kcal/mol) equivalent to two molecules. The activation energy, which equivalents to one molecule, is a half of the values (6-*endo*: 19.7 kcal/mol, 5-*exo*: 15.5 kcal/mol).
- 25 a) A. R. Katritzky, *Chem. Rev.* 2001, 101, 1421; b) P. v. R. Schleyer, C. Maerker, A. Dransfeld,
  H. Jiao, N. J. R. v. E. Hommes, *J. Am. Chem. Soc.* 1996, 118, 6317.
- 26 a) A. Issaian, D. J. Faizi, J. O. Bailey, P. Mayer, G. Berionni, D. A. Singleton, S. A. Blum, J. Org. Chem. 2017, 82, 8165; b) J. Jiang, Z. Zhang, Y. Fu, Asian J. Org. Chem., 2017, 6, 282.
- 27 X. Shi, N. El Hassan, A. Ikni, W. Li, N. Guiblin, A. Spasojević de-Biré, N. E. Ghermani, *CrystEngComm.* 2016, 18, 3289.
- 28 P. Schwerdtfeger, Atomic Static Dipole Polarizabilities, in Atoms, Molecules and Clusters in Electric Fields, ed. G. Maroulis, Imperial College Press, London, 2006, pp 1-32.
- 29 Silyl anion is stable than carboanion because of the larger atomic radius and higher dipole polarizability of silicon. A. C. Hopkinson, M. H. Lien, *Tetrahedron* **1981**, *37*, 1105.
- 30 K. Zhao, L. Shen, Z.-L. Shen, T.-P. Loh, Chem. Soc. Rev. 2017, 46, 586.
- 31 T.-H. Nguyen, A.-S. Castanet, J. Mortier, Org. Lett. 2006, 8, 765.
- 32 M. Uemura, T. Sakan, J. Chem. Soc. D 1971, 16, 921.
- 33 L. C. Wilkins, B. A. R. Günther, M. Walther, J. R. Lawson, T. Wirth, R. L. Melen, *Angew. Chem. Int. Ed.* **2016**, *55*, 11292.
- 34 Y.-X. Hu, L.-F. Wang, W.-W. Zhang, Y.-Z. Li, X.-M. Ren, J. Bai, *Inorg. Chem. Commun.* 2012, 17, 173.
- 35 H. Cho, Y. Iwama, K. Okano, H. Tokuyama, Chem. Pharm. Bull. 2014, 62, 354.
- 36 K. W. Kuntz, R. Chesworth, K. W. Duncan, H. Keihack, N. Warholic, C. Klaus, W. Zheng, M. Seki, S. Shirotori, S. Kawano, PCT Int. Appl. (2012), WO 2012142504 A1
- 37 V. G. Landge, G. Jaiswal, E. Balaraman, Org. Lett. 2016, 18, 812.
- 38 R. Mancuso, C. C. Pomelli, F. Malafronte, A. Maner, N. Marino, C. Chiappe, B. Gabriele, Org. Biomol. Chem. 2017, 15, 4831.
- 39 T. Katsumata, M. Shiotsuki, F. Sanda, X. Sauvage, L. Delaude, T. Masuda, *Macromol. Chem. Phys.* **2009**, *210*, 1891.
- 40 H. Kusama, H. Funami, J. Takaya, N. Iwasawa, Org. Lett. 2004, 6, 605.
- 41 F. M. Hauser, V. M. Baghdanov, J. Org. Chem. 1988, 53, 4676.

- 42 M. Peuchmaur, V. Lisowski, C. Gandreuil, L. T. Maillard, J. Martinez, J.-F. Hermandez, J. Org. Chem. 2009, 74, 4158.
- 43 M. Lessi, T. Masini, L. Nucara, F. Bellina, R. Rossi, Adv. Synth. Catal. 2011, 353, 501.
- 44 T.-H. Nguyen, A.-S. Castanet, J. Mortier, Org. Lett. 2006, 8, 765.
- 45 J. L. Seidel, W. W. Epstein, D. W. Davidson, J. Chem. Ecol. 1990, 16, 1791.
- 46 Y. C. Fan, O. Kwon, Org. Lett. 2012, 14, 3264.
- 47 T. Kovács, I. Sonnenbichler, J. Sonnenbichler, Liebigs Ann./Recueil 1997, 773.

### Chapter 4

# Synthesis of 2,4,6-Triarylpyryliums Bearing a Carbon-Metal Bond and their Application to Bifunctional Photocatalyst

#### 4-1. Introduction

Pyrylium is an important class of oxygen-containing heteroaromatic compounds in photochemistry and organic synthetic chemistry. Plant pigments such as anthocyanidin include a pyrylium framework.<sup>1</sup> The application of 2,4,6-triarylpyryliums to a photoredox catalyst is well-established<sup>2</sup> because 2,4,6triarylpyryliums absorb visible light and the corresponding excited state works as a strong oxidant due to its cationic character. Pyryliums also work as a synthetic building block to give various heterocycles.<sup>3</sup> Especially, pyryliums bearing a carbon-metal bond have been utilized as an useful intermediate to synthesize oxacycles and polycyclic compounds.<sup>4</sup> The organometallic pyryliums are usually unstable, thus, isolation of them has been seldom reported (Scheme 1A).<sup>5</sup> Iwasawa and Uemura synthesized pyryliums bearing a carbon-tungsten bond and applied it in Dield-Alder reaction.<sup>5b,5c</sup> Shaw investigated redox properties of pyryliums including a carbon-manganese or carbon-iron bond.<sup>5d,5e</sup> However, the structure and physicochemical properties of organometallic pyryliums have been much less investigated than that of no-metalated pyryliums, and there have been no reports for the application to a photocatalsyt. Herein, the synthesis of two types of 2,4,6-triphenylpyryliums bearing a carbon-metal bond (2 and 3) by intramolecular oxymetalation of carbonyl-ene-yne 1 and anion abstraction is described (Scheme 1B). Determination of their structures and evaluation of physicochemical properties were conducted, and their application as a photocatalyst possessing a Lewis acidic metal moiety<sup>6</sup> was also investigated.





**Scheme 1.** (A) Previously developed organometallic pyryliums. (B) This work: synthesis and evaluation of physicochemical properties of 2,4,6-triphenylpyryliums bearing a caebon-metal bond and thier application to a bifunctional catalyst.

#### 4-2. Results and Discussion

Investigation of suitable metal salts for synthesis of 2,4,6-triphenylpyryliums bearing a carbon-metal bond was carried out (Table 1). In chapter 3, it was revealed that the intramolecular oxymetalation of 2-alkynylbenzoate with indium or gallium trihalides afforded zwitterionic benzopyryliums bearing a carbon-metal bond.<sup>7</sup> Therefore, the oxymetalation of the carbonyl-ene-yne 1 using indium or gallium salts was conducted to access triphenylpyryliums. Oxyindation of 1 with InI<sub>3</sub> provided 2,4,6-triphenylpyrylium 2a in high yield (entry 1). GaI<sub>3</sub> and GaBr<sub>3</sub> also gave organometallic pyrylium 2b and 2c, respectively, in moderate yields (entries 2 and 3). Subjecting AgSbF<sub>6</sub> and AuSbF<sub>6</sub>, which work as an excellent  $\pi$ -electrophilic Lewis acid,<sup>8</sup> into the oxymetalation of 1 also gave metalated triphenylpyryliums (entries 4 and 5). Other Lewis acids such as boron and aluminum salts were not effective (entries 6-8).

<b>Fable 1.</b> Investigation of suitable metal sa	lts for oxymetalation of	the carbonyl-ene-yne 1. <sup>[a]</sup>
--	--------------------------	--

Ph Ph	n Mtx	$\begin{array}{c} \begin{array}{c} \text{Ph} \\ \hline \\ \hline \\ toluene \\ RT, 24 \text{ h} \end{array} \end{array} \xrightarrow{\text{Ph}} \begin{array}{c} \begin{array}{c} \text{Ph} \\ \hline \\ \\ \end{array} \\ \begin{array}{c} \\ \\ \\ \end{array} \\ \begin{array}{c} \\ \\ \end{array} \\ \begin{array}{c} \\ \\ \end{array} \\ \begin{array}{c} \\ \\ \end{array} \\ \begin{array}{c} \\ \\ \\ \end{array} \\ \begin{array}{c} \\ \\ \end{array} \\ \begin{array}{c} \\ \\ \\ \end{array} \\ \begin{array}{c} \\ \\ \end{array} \\ \end{array} \\ \begin{array}{c} \\ \\ \end{array} \\ \begin{array}{c} \\ \\ \end{array} \\ \end{array} \\ \begin{array}{c} \\ \end{array} \\ \end{array} \\ \begin{array}{c} \\ \\ \end{array} \\ \end{array} \\ \begin{array}{c} \\ \\ \end{array} \\ \end{array} \\ \begin{array}{c} \\ \end{array} \\ \end{array} \\ \begin{array}{c} \\ \end{array} \\ \end{array} \\ \begin{array}{c} \\ \\ \end{array} \\ \end{array} \\ \begin{array}{c} \\ \end{array} \\ \end{array} \\ \begin{array}{c} \\ \end{array} \\ \end{array} \\ \begin{array}{c} \\ \\ \end{array} \\ \end{array} \\ \end{array} \\ \begin{array}{c} \\ \end{array} \\ \end{array} \\ \end{array} \\ \begin{array}{c} \\ \end{array} \\ \end{array} \\ \end{array} \\ \end{array} \\ \begin{array}{c} \\ \end{array} \\ \end{array} \\ \end{array} \\ \end{array} \\ \end{array} \\ \begin{array}{c} \\ \end{array} \\ $	
Entry	MtX <sub>n</sub>	Yield of <b>2</b> (%) <sup>[b]</sup>	
1	InI <sub>3</sub>	<b>2a</b> : 83	
2	GaI <sub>3</sub>	<b>2b</b> : 66	
3	GaBr <sub>3</sub>	<b>2c</b> : 54	
4	AgSbF <sub>6</sub>	<b>2d</b> : 81	
5	AuSbF <sub>6</sub>	<b>2e</b> : 67	
6	AlI <sub>3</sub>	<b>2f</b> : 0	
7	AlBr <sub>3</sub>	<b>2g</b> : 12	
8	BBr <sub>3</sub>	<b>2h</b> : 0	
[a] Reaction conditions: 1 (1 equiv), $MtX_n$ (1 equiv), toluene, RT, 24 h. [b] Isolated yield.			

As shown in Figure 1, X-ray crystallographic analysis of **2a-2c** revealed that the formation of zwitteionic 2,4,6-triphenylpyryliums including a carbon-metal bond at the 3-position. Torsion angles of C2-C1-C4-C5 and C2-C3-C6-C7 for **2a-2c** (average  $38.58^{\circ}$ ) are much larger than that of a pyrylium without a carbon-metal bond (average  $15.44^{\circ}$ , see Figure S1 in Experimental Section for X-ray structure of the pyrylium without a carbon-metal bond). This result suggested that the two phenyl groups at the 2 and 4-positions rotated to decrease a steric repulsion between the phenyl groups and MtX<sub>3</sub> moiety.



**Figure 1.** The X-ray crystallographic structure of the zwitterionic metalated pyryliums **2a-2c** with the thermal ellipsoids shown at 50% probability. Selected bond lengths (Å) and torsion angles (deg) for **2a**: C2-In = 2.223(4), In-I1 = 2.7324(3), In-I2 = 2.7296(3), In-I3 = 2.7236(3), C2-C1-C4-C5 = 43.97, C2-C3-C6-C7 = 44.91. **2b**: C2-Ga = 2.037(7), Ga-I1 = 2.6108(11), Ga-I2 = 2.5267(11), Ga-I3 = 2.5498(9), C2-C1-C4-C5 = 34.92, C2-C3-C6-C7 = 38.34. **2c**: C2-Ga = 2.016(3), Ga-Br1 = 2.4062(6), Ga-Br2 = 2.3259(6), Ga-Br3 = 2.3541(5), C2-C1-C4-C5 = 34.48, C2-C3-C6-C7 = 34.85.

The complex **2**, in which the metal center has a formal negative charge, was treated with AgSbF<sub>6</sub> to abstract a halide anion on the metal. This anion abstraction efficiently occurred only in the case of **2c** (Eq. 1), and the cases using **2a** or **2b** resulted in complicated mixture. Recrystallization of **3c** from CHCl<sub>3</sub> provided a single crystal to be suitable for X-ray diffraction analysis (Figure 2). Two phenyl groups at 2- and 4-positions in **3c** rotated in the same manner as **2c**. The gallium atom was coordinated with two bromide atoms and THF, and its geometry has a distorted tetrahedral. The coordination of THF to gallium center shows that **3c** works as a Lewis acid.





**Figure 2.** The X-ray crystallographic structure of the metalated pyrylium **3c** with the thermal ellipsoids shown at 50% probability. Selected bond lengths (Å) and dihedral angles (deg): C2-Ga = 2.004(5), Ga-O = 1.949(5), Ga-Br1 = 2.3155(9), Ga-Br2 = 2.3055(9), C2-C1-C4-C5 = 34.65, C2-C3-C6-C7 = 37.20.

Electronic absorption and fluorescence spectra of the organometallic pyryliums are shown in Figure 3. The metalated triphenylpyryliums **2a**, **2c** and **3c** exhibited broad absorption bands at the almost same wavelengths (389, 383, 398 nm, respectively). On the other hand, two sharp absorption bands (370, 417 nm) were observed for no-metalated triphenylpyrylium **4**. Thus, the longest absorption wavelength of organomettalic pyryliums became a little bit shorter than that of **4**. Fluorescence wavelengths are the almost same whether the triphenylpyryliums have a metal moiety or not. To compare the characteristics between metalated and no-metalated triphenylpyryliums, density functional theory (DFT) calculation was performed for **2c** and **4**. Time-dependent (TD)-DFT calculation showed that the lowest-energy transition for **2c** and **4** is assigned to the HOMO-6- LUMO and HOMO-LUMO transition, respectively (See Table S1 and S2 in Experimental Section for TD-DFT calculation). The energy level of HOMO-6 for **2c** is close to HOMO of **4**, but **2c** has the higher LUMO level than **6** to induce blue shift of the longest absorption wavelength (See Figure S2 in Experimental Section for energy level diagrams and molecular orbitals). The higher LUMO level for **2c** would be ascribed to less  $\pi$ -conjugation between phenyl groups and the pyrylium framework caused by the rotation of two phenyl groups at the 2- and 4-positions (Figure 1 and 2).



Figure 3. Electronic absorption (solid lines) and normalized fluorescence spectra (dotted lines) for 2a, 2c, 3c and 4 in CH<sub>2</sub>Cl<sub>2</sub>.

Reduction potential of the triphenylpyryliums including a carbon-metal bond is determined by cyclic voltammetry (Figure 4). The organometallic triphenylpyryliums **2a**, **2c** and **3c** have the almost same reduction peak at -1.02, 1.04, and -1.03 V, respectively. On the other hand, reduction potential of the no-metalated pyrylium **4** is -0.79 V. Therefore, the metalated triphenylpyryliums show more negative reduction potential than **4**, which agrees with the higher LUMO level of **2c**.



**Figure 4.** Cyclic voltammogram of **2a**, **2c**, **3c** and **4** (V vs. Fc/Fc+, in 0.1M <sup>n</sup>Bu<sub>4</sub>NClO<sub>4</sub>/CH<sub>2</sub>Cl<sub>2</sub>, scan rate =100 mVs<sup>-1</sup> for **4**, 800 mVs<sup>-1</sup> for **2a**, **2c** and **3c**, room temperature).

Reduction potential of excited state of the organometallic pyryliums **2a**, **2c** and **3c** is estimated using fluorescence wavelength and reduction potential of their groud state (Figure 5).<sup>2b</sup> The organometallic triphenylpyryliums show highly positive reduction potential although the potentials are less positive than the reference compound **4**. Therefore, the excited triphenylpyryliums including a carbon-metal bond work as a strong oxidant.



**Figure 5.** Reductive potential of excited state of the organometallic triphenylpyryliums **2a**, **2c**, **3c** and the triphenylpyrylium tetrafluoroborate **4**.

The organometallic triphenylpyryliums was applied as a photocatalyst including a Lewis acid moiety. Mattay reported that the triphenylpyrylium **4** worked as a photocatalyst in dimerization of 1,1-diphenylethyrene **5**, in which one electron oxidation of **5** followed by addition of the radical cation to another **5** led to the homo-coupling products.<sup>9</sup> In the case that the Lewis acid moiety of the triphenylpyrylium catches a radical acceptor, the radical cation generated from **5** reacts with the caught acceptor in prior to **5** to accomplish the desired cross-coupling. Thus, a conjugate addition of 1,1-diphenylethyrene **5** with  $\alpha,\beta$ -unsaturated ketone **6**<sup>10</sup> was carried out under blue LED light irradiation in the presence of synthesized triphenylpyryliums (Table 2). Gratifyingly, the triphenylpyrylium **3c** including a Lewis acidic gallium moiety gave target product **7** in high yield (entry 1). Irradiation conditions accelerated the cujugated addition because the yield was decreased in dark conditions (entry 2). On the other hand, the triphenylpyryliums bearing carbon-silver or carbon-gold bonds (**2d** and **2e**) were less effective for the cross-coupling than **3c** (entries 3 and 4). The zwitterionic organogallium pyrylium **2c** and the triphenylpyrylium **4**, which have no Lewis acidic moiety, afforded **7** in low yields (entries 5 and 6). Interestingly, the combination of **4** with GaBr<sub>3</sub> showed a lower yield than **3c** (entry 7). This result suggested that **3c** which has a Lewis acidic moiety promoted the conjugate addition.

Ph		Photocatalyst (5 mo	1%)			Ph	Ph
	o ∖ ∥	additive (5 mol%)		Ph	0	s,	
× +		CH <sub>2</sub> Cl <sub>2</sub> , 30 min			× + (		
<b>5</b> (3 equiv)	6	Blue LED		7		Ph Ph	Ph Ph
	_					8	9
	Entry	Photocatalyst	additive	-	Yield (%) <sup>[b]</sup>	0	
		- Dh		7	8	9	
		SbF <sub>6</sub>					
	1	$\int \int 0$	none	88	5	23	
	1	Ph Ph	none	00	5	23	
		- Dh					
		SbF <sub>6</sub>					
	<b>?</b> [c]	$\int_{-\infty}^{\infty} \int_{-\infty}^{\infty}$	none	51	0	0	
	2	Ph Ph	none	51	0	0	
		Dh					
		SbF <sub>6</sub>					
	3		none	42	7	12	
	5	Ph Y Ph	none	12	,	12	
		Ay 2d					
		24					
		SbF <sub>6</sub> +					
	4		none	31	6	14	
	4	Ph	none	51	0	14	
		Au 2e					
		Ph 人 +					
	5		none	37	4	7	
	5	Ph Ph	none	52	7	,	
		GaBr <sub>3</sub> 2c					
		Ph					
		BF <sub>4</sub>					
	6		none	30	8	17	
		Ph Y Ph <b>4</b>					
		- Ph					
	7		GoP.	16	Л	0	
	/	Ph	GaD13	40	4	7	
		4					
	[a] Reac	tion conditions: 5(3)	$e_{\text{auiv}}$ 6(1)	ouiv) nh	otocatalyst (	5  mol	

Table 2. Effect of photocatalysts on direct conjugate addition of the alkene 5 to the ketone 6.<sup>[a]</sup>

[a] Reaction conditions: **5** (3 equiv), **6** (1 equiv), photocatalyst (5 mol%), additive (5 mol%), CH<sub>2</sub>Cl<sub>2</sub>, RT, 30 min, 40W Blue LED. [b] NMR yield. [c] In dark.

Proposed mechanism of the conjugate addition is described in Scheme 1. Firstly, the enone 6 coordinates to the gallium moiety of the organometallic triphenylpyrylium 3c. Triphenylpyrylium 10 coordinated by 6 absorbs visible light and exicited pyrylium  $10^*$  oxidizes diphenylethyrene 5 to generate radical cation 11 (See Figure S3 in Experimental Section for Stern-Volmer fluorescence quenching studies of 3c). The conjugate addition of radical species 11 with 6 coordinating to the gallium proceeds smoothly because these reactants are adjacent to each other in the case that a photocatalyst possesses a Lewis acid moiety. Intramolecular single electron transfer (SET) from the pyrylium moiety of 12 to the  $\alpha$ -carbonyl radical occurs to give gallium enolate 13. Finally, proton transfer and ligand exchange afford the target product 7 and regenerate the photocatalyst 10.



Scheme 1. Proposed mechanism of the cross-coupling between 1,1-diphenylethyrene 5 and methyl vinyl ketone 6.

#### 4-3. Conclusion

Intramolecualr oxymetalation of the carbonyl-ene-yne **1** proceeded efficiently by using  $\pi$ electrophilic Lewis acids such as indium, gallium, silver and gold salts to construct the triphenylpyryliums framework. X-ray crystallographic analysis revealed that formation of the triphenylpyryliums bearing a carbon-metal bond at the 3-positon and the steric hindrance of metal moiety induced the rotation of phenyl groups in the 2-and 4-positions. Anion abstraction enhanced the Lewis acidity of the zwitterionic gallium complex **2c**. Evaluation of optical and electrochemical properties of the organometallic triphenylpyryliums suggested the increase of their LUMO level compared with the no-metalated triphenylpyrylium **4**, which is supported by the DFT calculations. However, exicited states of the metalated triphenylpyryliums have high reduction potential enough to oxidize simple alkenes. The triphenylpyrylium **3c** possessing a Lewis acidic gallium moiety worked well in the photo-induced direct conjugate addition of alkenes to enones. The pyrylium moiety generates the active radical species and the intramolecular gallium moiety not only activates the enone but also positions the radical acceptor close to the alkyl radical to accelerate the cross-coupling. Therefore, the triphenylpyryliums including a Lewis acidic moiety acts as a bifunctional catalyst in the reaction system.

#### 4-4. Experimental Section

#### General

New compounds were characterized by <sup>1</sup>H and <sup>13</sup>C NMR, <sup>13</sup>C off-resonance techniques, COSY, HMQC, HMBC, IR, MS, HRMS. <sup>1</sup>H (400 MHz) and <sup>13</sup>C NMR (100 MHz) spectra were obtained with TMS as internal standard. High-resolution mass spectra were obtained by magnetic sector type mass spectrometer. UV-vis-NIR spectra were recorded in in a quartz cell (optical path length: 1 cm) using a JASCO V-630 spectrophotometer. Fluorescence spectra were recorded using a HITACHI F-4500 fluorescence spectrophotometer. Blue LED light irradiation was performed using a Kessil A160WE Tuna Blue. Cyclic voltammetric measurements were performed with an ALS-600C electrochemical analyzer using a glassy carbon working electrode, a Pt counter electrode, and an Ag/AgNO<sub>3</sub> reference electrode at room temperature in CH<sub>2</sub>Cl<sub>2</sub> containing 0.1 M Bu<sub>4</sub>NClO<sub>4</sub> as the supporting electrolyte. All calculations were performed with Gaussian09 rev.E.01. The RB3LYP density functional and a mixed basis set of DGDZVP for Ga and Br and 6-31+G(d,p) for other atoms were used to optimize all molecular geometries. Excitation energies were calculated at the TD-B3LYP density functional and the same basis set as above. Column chromatography was performed on silica gel (MERK C60 or Fuji Silysia FL100DX). All reactions were carried out under nitrogen. Yields were determined by <sup>1</sup>H NMR using internal standards (1,1,2,2-tetrachloroethane).

#### Materials

Dehydrated toluene, THF were purchased and used without further purification. Anhydrous  $CH_2Cl_2$  was purchased and degassed by a repeated freeze-pump-thaw method (3 times). The carbonyl-ene-yne **1** is new compounds, and synthetic method and spectral data of **1** are shown below. AuSbF<sub>6</sub> was prepared from AgSbF<sub>6</sub> and AuCl and used without purification.<sup>11</sup> The other metal salts, 1,1-diphenylethyrene **5**, methyl vinyl ketone **6** and triphenylpyrylium tetrafluoroborate **4** are commercially available.

(starting material for synthesis of 1) (Z)-3-iodo-1,3-diphenylprop-2-en-1-one<sup>12</sup>



To a three necked flask, 1,3-diphenylprop-2-yn-1-one (8.25 g, 40 mmol), acetic acid (30 mL) and NaI (19.2 g, 128 mmol) was added. The mixture was stirred at 110 °C for 6 h and cooled to rt, The solution was poured into CH<sub>2</sub>Cl<sub>2</sub> (50 mL) and H<sub>2</sub>O (50 mL) and the resultant mixture was extracted with CH<sub>2</sub>Cl<sub>2</sub> (50 mL x 2). The collected organic layer was washed with 5% Na<sub>2</sub>CO<sub>3</sub> aq. and sat. Na<sub>2</sub>S<sub>2</sub>O<sub>3</sub> aq., and the organic layer was dried (MgSO<sub>4</sub>). The solvent was evaporated and the residue was purified by column chromatography (hexane/ethyl acetate = 90:10, column length 24 cm, diameter 26 mm, spherical silica gel) to give the product (7.0 g, 52%). <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 8.00 (d, *J* = 7.2 Hz, 2H, 9-H), 7.63-7.55 (m, 3H, 5-H and 11-H), 7.53 (s, 1H, 2-H), 7.48 (t, *J* = 7.2 Hz, 2H, 10-H), 7.41-7.35 (m, 3H, 6-H and 7-H); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 190.3 (s, C-1), 143.1 (s, C-4), 136.9 (s, C-8), 133.4 (d), 131.4 (d, C-2), 129.9 (d), 128.7 (d), 128.6 (d), 128.4 (d), 113.2 (s, C-3); MS: (EI, 70 eV) *m/z* 334 (M<sup>+</sup>, 28), 207 (29), 105 (100), 77 (57); HRMS: (CI, 70 eV) Calculated (C<sub>15</sub>H<sub>12</sub>IO) 334.9933 (M + 1) Found 334.9931.

(1) (Z)-1,3,5-triphenylpent-2-en-4-yn-1-one<sup>13</sup>



To a three necked flask, (*Z*)-3-iodo-1,3-diphenylprop-2-en-1-one (3.33 g, 9.96 mmol), NEt<sub>3</sub> (40 mL), PdCl<sub>2</sub>(PPh<sub>3</sub>)<sub>2</sub> (0.072 g, 0.103 mmol) and CuI (0.024 g, 0.126 mmol) was added. After cooling the mixture to 0 °C, ethynylbenzene (1.40 g, 13.7 mmol) was added slowly. The mixture was stirred at rt for 11 h. The resultant solution was filtered through Celite pad and the solvent was evaporated. The residue was purified by column chromatography (hexane/ethyl acetate = 90:10, column length 24 cm, diameter 26 mm, spherical silica gel) to give the product (1.90 g, 62%). <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 8.05 (d, J = 7.2 Hz, 2H, 7-H), 7.92-7.85 (m, 2H, 11-H), 7.61-7.56 (m, 1H), 7.56-7.42 (m, 8H), 7.41-7.30 (m, 3H); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 190.1 (s, C-1), 138.6 (s), 137.5 (s), 135.1 (s), 132.7 (d), 132.2 (d), 130.0 (d), 129.2 (d), 128.7 (d), 128.5 (d), 128.3 (d), 127.6 (d), 127.3 (d), 122.5 (s), 101.7 (s), 88.0 (s); HRMS: (CI, 70 eV) Calculated (C<sub>23</sub>H<sub>17</sub>O) 309.1279 (M + 1) Found 309.1274.

#### Experimental procedure in oxymetalation of 1 (Table 1).

In a glove box filled for nitrogen, to a sealed vial, (*Z*)-1,3,5-triphenylpent-2-en-4-yn-1-one **1** (0.3 mmol), toluene (0.6 mL) and metal salt (0.3 mmol) were added. The solution was stirred at RT for 24 h. The reaction mixture was washed by hexane (3 mL x2) and CHCl<sub>3</sub> (3 mL x 6). The residue was dried under vacuum to give the product as a solid.

#### Experimental procedure in the direct conjugate addition (Table 2).

In a glove box filled for nitrogen, to a sealed vial, 1,1-diphenylethyrene 5 (0.3 mmol),  $CH_2Cl_2$  (1.5 mL), methyl vinyl ketone 6 (0.3 mmol) and photocatalyst (0.015 mmol) were added. The solution was stirred cooling with a fan at RT under 40W blue LED light irradiation. The reaction vessel was positioned at a distance of 7 cm from the blue LED. After stirring for 30 min, the reaction mixture was diluted with  $CH_2Cl_2$  (10 mL). The solvent was evaporated and the yield of products 7, 8 and 9 was determined by <sup>1</sup>H-NMR. The products 7, 8 and 9 are known compounds and the spctra agreed wirh the literature.<sup>9,10a</sup>

#### X-ray structure of the pyrylium without a carbon-metal bond

After oxymetalation of carbonyl-ene-yne 1 with AlBr<sub>3</sub> (Table 1, entry 7), the product was recrystallized from CH<sub>3</sub>CN to give a single crystal. X-ray diffraction analysis showed that the crystal is not the target organometallic triphenylpyrylium 2g but a protonated triphenylpyrylium as shown below.



**Figure S1.** The X-ray crystallographic structure of the triphenylpyrylium without a carbon-metal bond with the thermal ellipsoids shown at 50% probability. Selected dihedral angles (deg): C2-C1-C4-C5 = 11.44, C2-C3-C6-C7 = 19.44.

# TD-DFT calculation of 2c and 4

Table S1. Excitation energies of 2c	:.
-------------------------------------	----

Excited state	Excitation energy /eV (wavelength /nm)	Excitation amplitudes	Oscillator strength	<\$2>
7	3.16 (392.9)	0.694 (HOMO-6 - LUMO)	0.2612	0.000
8	3.23 (383.4)	0.612 (HOMO-7 - LUMO) 0.165 (HOMO-9 - LUMO) -0.295 (HOMO-8 - LUMO)	0.0198	0.000
11	3.39 (365.9)	0.465 (HOMO-9 - LUMO) -0.320 (HOMO-7 - LUMO) -0.407 (HOMO-8 - LUMO)	0.2401	0.000

## Table S2. Excitation energies of 4.

Excited state	Excitation energy /eV (wavelength /nm)	Excitation amplitudes	Oscillator strength	<\$2>
1	2.85 (435.6)	0.677 (HOMO - LUMO) -0.199 (HOMO-1 - LUMO)	0.3042	0.000
5	3.41 (363.7)	0.687 (HOMO-4 - LUMO) 0.157 (HOMO - LUMO+1)	0.2663	0.000
9	3.94 (315.1)	0.640 (HOMO-7 - LUMO) 0.277 (HOMO - LUMO+1)	0.3765	0.000



#### Energy level diagrams and molecular orbitals of 2c and 4

Figure S2. (A) HOMO-6, HOMO-7, LUMO and LUMO+1 of 2c. (B) HOMO, HOMO-1, LUMO and LUMO+1 of 4.

#### Stern-Volmer fluorescence quenching studies

Fluorescence quenching studies were performed using a HITACHI F-4500 fluorescence spectrophotometer. In each experiment, the photocatalyst and various concentrations of the quencher were combined in CH<sub>2</sub>Cl<sub>2</sub> in screw-top 1.0 cm quartz cell. The emission quenching of the triphenylpyrylium including a carbon-gallium bond **3c** was achieved using the photocatalyst concentration of 2.0 x 10<sup>-6</sup> M under excitation at 420 nm. The emission intensity was observed at 465 nm. Plots were constructed according to the Stern–Volmer equation  $I_0/I = 1 + k_q \tau_0[Q]$ .<sup>14</sup> The exicited



state of 3c was quenched by not the methyl vinyl ketone 6 but the 1,1-diphenylpyrylium 5.

Figure S3. Stern-Volmer fluorescence quenching studies for the oraganogallium triphenylpyrylium 3c.

#### Product

(2a) 2,4,6-triphenylpyrylium including InI<sub>3</sub> moiety at 3-position



To a sealed vial,  $InI_3$  (0.152 g, 0.307 mmol), toluene (0.6 mL) and (*Z*)-1,3,5-triphenylpent-2-en-4-yn-1one (0.0924 g, 0.30 mmol) were added. The mixture was stirred at RT for 24 h. The reaction mixture was washed by hexane (3 mL x 2) and CHCl<sub>3</sub> (3 mL x 6). The residue was dried under vacuum to give the product as a yellow solid (0.20 g, 83% yield). This compound was recrystallized from CH<sub>3</sub>CN to give a single crystal. The structure was determined by X-ray crystallographic analysis. Characterization by NMR study was also carried out and the spectra is shown below. <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 8.28-8.14 (m, 5H), 8.04 (d, *J* = 7.2 Hz, 2H), 7.85-7.69 (m, 9H); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 178.6 (s), 177.2 (s), 168.8 (s, C-3), 139.8 (s), 135.4 (d), 134.6 (d), 134.1 (d), 133.9 (s), 131.4 (d), 130.44 (d), 130.36 (d), 130.08 (d), 128.6 (s), 128.2 (s), 127.7 (d), 116.4 (d). (2b) 2,4,6-triphenylpyrylium including GaI<sub>3</sub> moiety at 3-position



To a sealed vial, GaI<sub>3</sub> (0.139 g, 0.309 mmol), toluene (0.6 mL) and (*Z*)-1,3,5-triphenylpent-2-en-4-yn-1-one (0.093 g, 0.302 mmol) were added. The mixture was stirred at RT for 24 h. The reaction mixture was washed by hexane (3 mL x 2) and CHCl<sub>3</sub> (3 mL x 6). The residue was dried under vacuum to give the product as a yellow solid (0.15 g, 66% yield). This compound was recrystallized from CH<sub>3</sub>CN to give a single crystal. The structure was determined by X-ray crystallographic analysis. Characterization by NMR study was also carried out and the spectra is shown below. <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 8.36 (d, J = 7.5 Hz, 2H), 8.18 (d, J = 8.0 Hz, 2H, 11-H), 8.16 (s, 1H, 2-H), 8.15 (d, J = 8.2 Hz, 2H), 7.84-7.77 (m, 2H), 7.77-7.66 (m, 7H); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 178.6 (s), 177.8 (s), 168.2 (s, C-3), 139.5 (s), 135.4 (d), 134.6 (d), 133.99 (s), 133.94 (d), 132.6 (s, C-4), 131.8 (d), 130.8 (d), 130.3 (d), 129.8 (d), 129.6 (d), 128.5 (s), 127.7 (d, C-11), 117.1 (d, C-2).

(2c) 2,4,6-triphenylpyrylium including GaBr<sub>3</sub> moiety at 3-position



To a sealed vial, GaBr<sub>3</sub> (0.094 g, 0.304 mmol), toluene (0.6 mL) and (*Z*)-1,3,5-triphenylpent-2-en-4-yn-1-one (0.0928 g, 0.301 mmol) were added. The mixture was stirred at RT for 24 h. The reaction mixture was washed by hexane (3 mL x 2) and CHCl<sub>3</sub> (3 mL x 6). The residue was dried under vacuum to give the product as an orange solid (0.10 g, 54% yield). This compound was recrystallized from CH<sub>3</sub>CN to give a single crystal. The structure was determined by X-ray crystallographic analysis. Characterization by NMR study was also carried out and the spectra is shown below. <sup>1</sup>H NMR: (400 MHz, CDCl<sub>3</sub>) 8.29 (d, J = 7.3 Hz, 2H), 8.19 (s, 1H, 2-H), 8.18 (d, J = 7.3 Hz, 2H, 11-H), 8.06 (d, J = 7.1 Hz, 2H), 7.82-7.77 (m, 2H), 7.77-7.65 (m, 7H); <sup>13</sup>C NMR: (100 MHz, CDCl<sub>3</sub>) 178.8 (s), 178.2 (s), 168.5 (s, C-3), 139.5 (s), 137.7 (s, C-4), 135.4 (d), 134.4 (d), 133.73 (d), 133.65 (s), 131.5 (d), 130.4 (d), 129.7 (d),

#### 4-5. References

- (1) J.-M. Kong, L.-S. Chia, N.-K. Goh, T.-F. Chia, R. Brouillard, Phytochemistry 2003, 64, 923.
- (2) (a) M. A. Miranda, H. García, *Chem. Rev.* 1994, 94, 1063; (b) N. A. Romero, D. A. Nicewicz, *Chem. Rev.* 2016, *116*, 10075.
- (3) (a) A. T. Balaban, Adv. Heterocycl. Chem. 2010, 99, 61; (b) P. W. Antoni, M. M. Hansmann, J. Am. Chem. Soc. 2018, 140, 14823.
- (4) For selected examples: (a) N. Iwasawa, M. Shido, H. Kusama, J. Am. Chem. Soc. 2001, 123, 5814; (b) N. Asao, T. Nogami, K. Takahashi, Y. Yamamoto, J. Am. Chem. Soc. 2002, 124, 764; (c) R. Yanada, K. Hashimoto, R. Tokizane, Y. Miwa, H. Minami, K. Yanada, M. Ishikura, T. Takemoto, J. Org. Chem. 2008, 73, 5135; (d) S. Zhu, Z. Zhang, X. Huang, H. Jiang, Z. Guo, Chem. Eur. J. 2013, 19, 4695; (e) G. Mariaule, G. Newsome, R. Y. Toullec, P. Belmont, V. Michelet, Org. Lett. 2014, 16, 4570; (f) H. Wang, Y. Kuang, J. Wu, Asian J. Org. Chem. 2012, 1, 302; (g) E. Tomás-Mendivil, F. C. Heinrich, J.-C. Ortuno, J. Starck, V. Michelet, ACS Catal. 2017, 7, 380.
- (5) (a) P. Legzdins, W. S. McNell, E. G. Vessey, *Organometallics* 1992, *11*, 2718; (b) N. Iwasawa, M. Shido, K. Maeyama, H. Kusama, *J. Am. Chem. Soc.* 2000, *122*, 10226; (c) K. Ohe, K. Miki, T. Yokoi, F. Nishino, S. Uemura, *Organometallics* 2000, *19*, 5525; (d) M. J. Shaw, J. Martz, *Organometallics* 2002, *21*, 3434; (e) M. J. Shaw, S. J. Afridi, S. L. Light, J. N. Mertz, S. E. Ripperda, *Organometallics* 2004, *23*, 2778.
- (6) W. Ding, L.-Q. Lu, Q.-Q. Zhou, Y. Wei, J.-R. Chen, W.-J. Xiao, J. Am. Chem. Soc. 2017, 139, 63.
- (7) Y. Kita, T. Yata, Y. Nishimoto, K. Chiba, M. Yasuda, Chem. Sci. 2018, 9, 6041.
- (8) (a) G. Fang, X. Bi, Chem. Soc. Rev. 2015, 44, 8124. (b) R. Dorel, A. M. Echavarren, Chem. Rev. 2015, 115, 9028.
- (9) J. Mattay, M. Vondenhof, R. Denig, Chem. Ber. 1989, 122, 951.
- (10) (a) K. Okamoto, E. Tamura, K. Ohe, *Angew. Chem. Int. Ed.* 2013, *52*, 10639. (b) J. Lv, X. Zhong, S. Luo, *Chem. Eur. J.* 2014, *20*, 8293.
- (11) D. Weber, T. D. Jones, L. L. Adduci, M. R. Gagné, Angew. Chem. Int. Ed. 2012, 51, 2452.
- (12) E. Piers, T. Wong, P. D. Coish, C. Rogers, Can. J. Chem. 1994, 72, 1816.
- (13) L. C. Wilkins, B. A. R. Günther, M. Walther, J. R. Lawson, T. Wirth and R. L. Melen, Angew. Chem. Int. Ed. 2016, 55, 11292.
- (14) C. K. Prier, D. W. C. MacMillan, Chem. Sci. 2014, 5, 4173.

## Conclusion

This thesis reported the direct functionalization of alkenes and regioselective formation of oxacycles from alkynes utilizing heavier group 13 metal salts which have moderate Lewis acidity and high  $\pi$ -electron affinity.

Chapter 1 revealed that various enol derivatives such as vinyl ethers, vinyl esters, vinyl silyl ethers and vinyl carbamates were suitable for the GaBr<sub>3</sub>-catalyzed cross coupling of enol derivatives with silyl ketene acetals. Mechanistic studies revealed that this coupling reaction proceeds through additionelimination mechanism. The turnover limiting step in the case of a vinyl ether is the elimination step while the addition step determines the reaction rate in the reaction using a vinyl ester.

In Chapter 2, I developed the formal hydrofunctionalization that installed a wide range of functional groups into the styrene derivatives. Various organosilicon nucleophiles were applicable in this hydrofunctionalization. This reaction is composed of two steps: Brønsted acid-catalyzed formation of cyclic ether followed by InI<sub>3</sub>-catalyzed C-O bond cleavage.

In Chapter 3, I achieved the facile synthesis of 4-substituted isocoumarins by 6-endo selective oxymetalation of 2-ethynylbenzoate. This cyclization proceeds in *anti*-Markovnikov addition manner. Full investigation of the mechanism by experimental data and theoretical calculations suggested that large ionic radius of indium had a great influence on the 6-endo selective heterocyclization. The synthesis of 3-nonsubstituted isocoumarins is difficult in spite of an important class of bioactive compounds. However, the isocoumarins bearing a carbon-indium bond enable easy access to 3-nonsubstituted isocoumarins.

Chapter 4 mentioned the synthesis of new types of organometallic pyryliums. I determined the structures by X-ray crystallographic analysis and evaluated their optical and electrochemical properties. The triphenylpyrylium including a gallium moiety worked as not only photocatalyst but also Lewis acid catalyst. This bifunctional catalyst showed high activity for the direct conjugate addition of alkenes to enones.

Chapters 1 and 2 give a knowledge that gallium and indium cleavage not only carbon-carbon multiple bonds but also carbon-oxygen bonds. It is difficult for  $\pi$ -electrophilic Lewis acidic transition metals such as silver and gold to cleavage carbon-oxygen bonds due to the low oxophilicity. On the other hand, heavier group 13 metals have moderate oxophilicity to accomplish such advanced transformations. A knowledge obtained from Chapters 3 and 4 is that gallium and indium provide stable organometallic ate complex because of their capacity of delocalizing anion charge.

The obtained knowledge provides an unprecedent strategy to functionalize carbon-carbon multiple bonds using main group metals and an opportunity to isolate unknown organometallic species that seems to be not isolated due to the unstability.